UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-K/A

Amendment No. 1

ANNUAL REPORT UNDER SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the fiscal year ended December 31, 2008

TRANSITION REPORT UNDER SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from _____ to _____

Commission file number: 0-12627

GLOBAL CLEAN ENERGY HOLDINGS, INC.

(Exact name of Small Business Issuer as specified in its charter)

Utah (State or other jurisdiction of

incorporation or organization)

6033 W. Century Blvd, Suite 895, Los Angeles, California 90045 (Address of principal executive offices)

(310) 641-4234

Issuer's telephone number:

Securities registered under Section 12(b) of the Act: None.

Securities registered under Section 12(g) of the Act: Common Stock, no par value.

Indicate by check mark if the registrant is a well known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes 🗆 No 🗹

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No 🗹

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and, (2) has been subject to such filing requirements for the past 90 days.

Yes 🗹 No 🗖

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§ 229.405 of this chapter) is not contained herein and, will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer	Accelerated filer	Non-accelerated filer \Box	Smaller reporting company 🗹
-		(Do not check if a smaller reporting company)	

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act).

Yes 🗆 No 🗹

The aggregate market value of the common stock held by non-affiliates of the registrant as of June 30, 2008 (the last business day of the registrant's most recently completed second fiscal quarter) was approximately \$8,865,000.

The outstanding number of shares of common stock as of April 8, 2009 was 229,381,338, which includes 4,567,519 shares of common stock currently held in escrow.

Documents incorporated by reference: None

87-0407858 (I.R.S. Employer Identification Number)

EXPLANATORY NOTE

We are filing this amendment (this "Amendment") to our Annual Report on Form 10-K for the year ended December 31, 2008 (our "Annual Report") to reflect changes made in response to comments received by us from the Staff of the Securities and Exchange Commission (the "SEC") in connection with the Staff's review of our Annual Report. We are only filing the items of our Annual Report that have been revised in response to the Staff's comments and all other information in our Annual Report remains unchanged. Accordingly, the Amendment should be read in conjunction with our Annual Report. Unless otherwise provided, all information contained in this Amendment is as of April 15, 2009, the original filing date of our Annual Report. This Amendment does not reflect events that have occurred after the filing of the Annual Report and does not modify or update the disclosure therein in any way other than as required to reflect the matters set forth herein.

The only changes to our Annual Report are in Item 9A(T) "Controls and Procedures" and in the financial statements and exhibits filed as part of Item 15 "Exhibits and Financial Statement Schedules." In Item 9A(T) we have revised our disclosure to more clearly present the conclusions of our principal executive, principal financial and principal accounting officers regarding the effectiveness of our disclosure controls and procedures in our Mexico subsidiary.

We have amended the exhibits filed as part of the Annual Report to include the following two additional agreements: (i) Limited Liability Company Agreement of GCE Mexico I, LLC, and (ii) Service Agreement between this company and Corporativo LODEMO S.A DE CV. The financial statements included in Item 15 have been modified (i) to mark as unaudited the information included in the Consolidated Statements of Operations and Cash Flows which is designated as "From Inception of the Development Stage on November 20, 1991 through December 31, 2008," (ii) to mark as unaudited the information included in the Consolidated Statements of Changes in Stockholders' Deficit for the Period from November 20, 1991 (Date of Inception of the Development Stage) through December 31, 2006, (iii) to remove all language from the Report of Independent Registered Public Accounting Firm referring to the periods prior to the year ended December 31, 2007 and to the reports of other auditors, and (iv) to clarify certain disclosures in the Notes to Consolidated Financial Statements regarding our Principles of Consolidation as they pertain to Asideros Globales Corporativo and GCE Mexico I, LLC.

Pursuant to the rules of the SEC, currently dated certifications from our principal executive and principal financial officer, as required by Sections 302 and 906 of the Sarbanes-Oxley Act of 2002, are filed or furnished herewith, as applicable.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA.

Financial Statements are referred to in Item 15, listed in the Index to Financial Statements and filed and included elsewhere herein as a part of this Annual Report on Form 10-K/A.

ITEM 9A(T). CONTROLS AND PROCEDURES.

Evaluation of Disclosure Controls and Procedures.

We maintain disclosure controls and procedures which are designed to ensure that the information required to be disclosed in the reports it files or submits under the Securities Exchange Act of 1934 (as amended, the "Act") is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including the Chief Executive Officer and the Chief Financial Officer ("Certifying Officers"), to allow timely decisions regarding required financial disclosures.

In connection with the preparation of this Annual Report, our Certifying Officers evaluated the effectiveness of management's disclosure controls and procedures, as of December 31, 2008, in accordance with Rules 13a-15(b) and 15d-15(b) of the Exchange Act. Based on that evaluation, the Certifying Officers concluded that management's disclosure controls and procedures were not effective as of December 31, 2008.

Material Weakness in Internal Control Over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting, as defined in Rule 15d-15(f) under the Exchange Act, and for assessing the effectiveness of internal control over financial reporting.

Internal control over financial reporting is intended to provide reasonable assurance regarding the reliability of our financial reporting and the preparation of financial statements for external purposes in accordance with accounting principles generally accepted in the United States. Internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of our assets, (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with accounting principles generally accepted in the United States and that our receipts and expenditures are being made only in accordance with authorizations of our management and directors, and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisitions, use, or disposition of our assets that could have a material effect on our financial statements.

Management, with the participation of our principal executive and financial officers, conducted an evaluation of the effectiveness of our internal control over financial reporting, as of December 31, 2008, based on criteria established in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). Based on that evaluation, management concluded that, as of December 31, 2008, our internal control over financial reporting was not effective.

Based on our evaluation of our internal control over financial reporting in our Mexico subsidiary, we have determined that we currently have inadequate controls over the accounting functions in Mexico and over cash management in Mexico. However, management does not believe that this material weakness resulted in any material misstatements in our financial condition for the current reporting period. Management is attempting to implement new controls to improve both of these deficiencies. The deficiencies consist of controls over the disbursement of cash from our accounts in Mexico and the proper categorization of such expenses for accounting purposes. The Company has begun to take appropriate steps to remediate these weaknesses as follows: The Company recently established new bank accounts in Mexico that require dual control of two persons for most disbursements. The Company has required that the Company be promptly notified of these disbursements for control and categorization purposes. Furthermore, we are commencing the implementation of procedures for remote real time access by the Company U.S. executives to bank accounts in Mexico. Accordingly, each disbursement will be able to be monitored in the U.S. to ensure proper use and to properly record such disbursements. The Company english of real time monitoring in the second half of 2009, assuming financial resources are available. The effectiveness of our internal controls following our remediation efforts will not be known until we test those controls in connection with management's tests of internal control.

Our Board of Directors believes that, with the exception of the issues identified relating to our operations in Mexico, our system of internal controls, disclosure controls and procedures are adequate to provide reasonable assurance that the information required to be disclosed in the our interim and annual reports is recorded, processed, summarized, and accurately reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our Board of Directors, the Audit Committee, management, including our certifying officers, as appropriate, to allow for timely decisions regarding required disclosure based closely on the definition of "disclosure controls and procedures" in Rule 13a-15(e). The Audit Committee cannot be certain that its remediation efforts will sufficiently cure management's identified material financial reporting weaknesses. Furthermore, the Audit Committee has not tested the operating effectiveness of the remediated controls, since the process is not yet complete. However, because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues, if any, within our company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of simple errors or mistakes.

This annual report does not include an attestation report of our independent registered public accounting firm regarding internal control over financial reporting. Management's report was not subject to attestation by our independent registered public accounting firm pursuant to temporary rules of the SEC that permit us to provide only management's report in this annual report.

Changes in Internal Control Over Financial Reporting

Except as reported above in this Item 9, there was no change in our internal control over financial reporting during the most recent fiscal quarter that materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART III

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES. The Company's financial statements and related notes thereto are listed and included in this Amendment No. 1 to Annual Report beginning on page F-1. The following documents are furnished as exhibits to this Form 10-K/A. Exhibits marked with an asterisk are filed herewith. The remainder of the exhibits previously have been filed with the Commission and are incorporated herein by reference.

Number	Exhibit
3.1	Amended and Restated Articles of Incorporation of the Company (filed as Exhibit 3.1 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1994, and incorporated herein by reference).
3.2	Amended Bylaws of the Company (filed as Exhibit 3.2 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1994, and incorporated herein by reference).
4.1	Certificate of Designations of Preferences and Rights of Series A Convertible Preferred Stock of Medical Discoveries, Inc. (filed as Exhibit 4.1 to Registration Statement No. 333-121635 filed on Form SB-2 on December 23, 2004, and incorporated herein by reference).
4.4	Amendment to Certificate of Designations of Preferences and Rights of Series A Convertible Preferred Stock of Medical Discoveries, Inc. (filed as Exhibit 4.2 to Registration Statement No. 333-121635 filed on Form SB-2 on December 23, 2004, and incorporated herein by reference).
4.5	Certificate Of Designation of Preferences and Rights Series B Convertible Preferred Stock of Medical Discoveries, Inc. (filed as Exhibit 4.1 to the Company's Current Report on Form 8-K filed November 13, 2007, and incorporated herein by reference)
10.1	2002 Stock Incentive Plan adopted by the Board of Directors as of July 11, 2002 (filed as Exhibit 10.5 to the Company's Quarterly Report on Form 10-QSB for the quarter ended June 30, 2002, and incorporated herein by reference).
10.2	Sale and Purchase Agreement between Attorney Hinnerk-Joachim Müller as liquidator of Savetherapeutics AG i.L. and Medical Discoveries, Inc. regarding the purchase of the essential assets of Savetherapeutics AG i.L. (filed as Exhibit 2.1 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2004, and incorporated herein by reference).
10.3	Share Exchange Agreement dated September 7, 2007 among Medical Discoveries, Inc., Richard Palmer, and Mobius Risk Group, LLC (filed as Exhibit 2.2 to the Company's Current Report on Form 8-K filed September 17, 2007, and incorporated herein by reference)
10.4	Definitive Master Agreement dated as of July 29, 2006, by and between MDI Oncology, Inc. and Eucodis Forschungs und Entwicklungs GmbH (filed as Exhibit 10.1 to the Company's Current Report on Form 8-K filed August 3, 2006, and incorporated herein by reference)
10.5	Loan and Security Agreement, dated September 7, 2007, between Medical Discoveries, Inc. and Mercator Momentum Fund III, L.P. (filed as Exhibit 10.1 to the Company's Current Report on Form 8-K filed September 17, 2007, and incorporated herein by reference).
10.6	Note Amendment And Maturity Date Extension, dated January 12, 2009, between the Company and Mercator Momentum Fund III, L.P.**
10.7	Consulting Agreement dated September 7, 2007 between Medical Discoveries, Inc. and Mobius Risk Group, LLC (filed as Exhibit 10.2 to the Company's Current Report on Form 8-K filed September 17, 2007, and incorporated herein by reference)

10.8	Employment Agreement dated September 7, 2007 between Medical Discoveries, Inc. and Richard Palmer (filed as Exhibit 10.3 to the
	Company's Current Report on Form 8-K filed September 17, 2007, and incorporated herein by reference)
10.9	Release and Settlement Agreement dated August 31, 2007 between Medical Discoveries, Inc. and Richard Palmer (filed as Exhibit 10.4 to the
	Company's Current Report on Form 8-K filed September 17, 2007, and incorporated herein by reference)
10.10	Release and Settlement Agreement, dated as of October 19, 2007, by and among the Company, on the one hand, and Mercator Momentum Fund,
	LP, Monarch Pointe Fund, Ltd., and Mercator Momentum Fund III, LP, on the other hand. (filed as Exhibit 10.1 to the Company's Current
	Report on Form 8-K filed October 26, 2007, and incorporated herein by reference)
10.11	Form of Warrant (filed as Exhibit 10.2 to the Company's Current Report on Form 8-K filed October 26, 2007, and incorporated herein by
	reference)
10.12	Securities Purchase Agreement, dated as of November 6, 2007, by and among Medical Discoveries, Inc. and the Purchasers (as defined therein)
	(filed as Exhibit 10.1 to the Company's Current Report on Form 8-K filed November 13, 2007, and incorporated herein by reference)
10.13	Employment Agreement dated March 20, 2008 between Global Clean Energy Holdings, Inc. and Bruce K. Nelson (filed as Exhibit 10.1 to the
	Company's Current Report on Form 8-K filed April 7, 2008, and incorporated herein by reference)
10.14	Exchange Agreement, effective April 18, 2008, by and between Global Clean Energy Holdings, Inc., on the one hand, and Mercator Momentum
	Fund, L.P., Mercator Momentum Fund III, L.P., and Monarch Pointe Fund, Ltd. (filed as Exhibit 10.1 to the Company's Current Report on Form
	8-K filed April 24, 2008, and incorporated herein by reference)
10.15	Amendment to Loan and Security Agreement, dated September 7, 2007, between Medical Discoveries, Inc. and Mercator Momentum Fund III,
	L.P. (filed as Exhibit 10.1 to the Company's Quarterly Report on Form 10-QSB filed August 14, 2008, and incorporated herein by reference)
10.16	Stock Purchase Agreement, dated October 30, 2008, between the Global Clean Energy Holdings, Inc. and the four shareholders of Technology
	Alternatives Limited, a Belizean Company formed under the Laws of Belize (filed as Exhibit 10.1 to the Company's Quarterly Report on Form
	10-QSB filed November 14, 2008, and incorporated herein by reference)
10.17	Limited Liability Company Agreement of GCE Mexico I, LLC, a Delaware Limited Liability Company, dated April 23, 2008*
10.18	Service Agreement, dated October 15, 2007, between the Company and Corporativo LODEMO S.A DE CV, a Mexican corporation*
14.1	Medical Discoveries, Inc. Code of Conduct**
23	Consent of Hansen, Barnett & Maxwell. P.C.*
31	Rule 13a-14(a) Certification, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 *
32	Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.*

Filed herewith.
 Filed with our Annual Report on Form 10-K for the year ended December 31, 2008, as originally filed on April 15, 2009.

SIGNATURES

In accordance with Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

GLOBAL CLEAN ENERGY HOLDINGS, INC.

December 1, 2009

By: /s/ RICHARD PALMER

Richard Palmer

President and Chief Executive Officer

In accordance with the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant in the capacities and on the dates indicated.

Signature	Title	Date
/s/ RICHARD PALMER	Chief Executive Officer	December 1, 2009
Richard Palmer	(Principal Executive Officer) and Director	
/s/ BRUCE NELSON Bruce Nelson	Chief Financial Officer (Principal Accounting Officer)	December 1, 2009
/s/ DAVID WALKER David Walker	Chairman, the Board of Directors	December 1, 2009
/s/ MARK A. BERNSTEIN Mark A. Bernstein	Director	December 1, 2009

Page

Financial Statements:	
Report of Independent Registered Public Accounting Firm	F-2
Consolidated Balance Sheets as of December 31, 2008 and 2007	F-3
Consolidated Statements of Operations for each of the two fiscal years	
in the period ended December 31, 2008 and from inception	F-4
Consolidated Statements of Changes in Stockholders' Deficit from inception	
through December 31, 2008	F-5
Consolidated Statements of Cash Flows for each of the two years	
in the period ended December 31, 2008 and from inception	F-7
Notes to Consolidated Financial Statements	F-8

HANSEN, BARNETT & MAXWELL, P.C.

Registered with the Public Company Accounting Oversight Board

A Professional Corporation CERTIFIED PUBLIC ACCOUNTANTS AND BUSINESS CONSULTANTS 5 Triad Center, Suite 750 Salt Lake City, UT 84180-1128 Phone: (801) 532-2200 Fax: (801) 532-7944 www.hbmcpas.com

an independent member of
BAKER TILLY
INTERNATIONAL
A Member of the Forum of Firms

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Shareholders Global Clean Energy Holdings, Inc. Los Angeles, CA

We have audited the accompanying consolidated balance sheets of Global Clean Energy Holdings, Inc. and subsidiaries (a development stage company) as of December 31, 2008 and 2007, and the related consolidated statements of operations, stockholders' deficit, and cash flows for the years then ended. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audit included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Global Clean Energy Holdings, Inc. and subsidiaries as of December 31, 2008 and 2007, and the results of their operations and their cash flows for the years then ended, in conformity with U.S. generally accepted accounting principles.

The accompanying consolidated financial statements have been prepared assuming that the Company will continue as a going concern. The Company is a development stage enterprise previously engaged in developing bio-pharmaceutical research and currently developing bio-diesel fuels. As discussed in Note B to the financial statements, the stockholders' deficit and the operating losses since inception raise substantial doubt about the Company's ability to continue as a going concern. Management's plans concerning these matters are also described in Note B. The financial statements do not include any adjustments that might result from the outcome of these uncertainties.

<u>/s/ HANSEN, BARNETT & MAXWELL, P.C.</u> HANSEN, BARNETT & MAXWELL, P.C.

Salt Lake City, Utah April 14, 2009

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A Development Stage Company) CONSOLIDATED BALANCE SHEETS

	December 31, 2008		December 31, 2007	
ASSETS				
CURRENT ASSETS				
Cash and cash equivalents	\$	291,309	\$	805,338
Subscription receivable	φ		Ψ	75,000
Other current assets		131,715		51,073
Total Current Assets		423,024		931,411
PROPERTY AND EQUIPMENT				
Land		2,051,282		-
Plantation development costs		2,117,061		308,777
Plantation equipment		509,037		-
Office equipment		10,993		1,127
		4,688,373		309,904
Less accumulated depreciation		(22,296)		(563)
	_	4,666,077	_	309,341
OTHER ASSETS		2,691		-
TOTAL ASSETS	\$	5,091,792	\$	1,240,752
LIABILITIES AND STOCKHOLDERS' DEFICIT				
CURRENT LIABILITIES				
Accounts payable	\$	1,890,999	\$	1.656.292
Accrued payroll and payroll taxes		1,158,808		950,971
Accrued interest payable		522,097		300,651
Accrued return on minority interest		138,014		-
Secured promissory note		460,000		250,000
Notes payable to shareholders		56,000		56,000
Convertible notes payable		193,200		193,200
Research and development obligation		2,607,945		2,701,555
Financial instrument		-		2,166,514
Total Current Liabilities		7,027,063		8,275,183
MORTGAGE NOTE PAYABLE		2,051,282		-
MINORITY INTEREST		1,962,022		-
		, ,		
STOCKHOLDERS' DEFICIT				
Preferred stock - no par value; 50,000,000 shares authorized				
Series A, convertible; zero and 28,928 shares issued and outstanding, respectively (aggregate liquidation preference of \$0 and \$2,892,800, respectively)		-		514.612
Series B, convertible; 13,000 shares issued or subscribed (aggregate liquidation preference of \$1,300,000)		1,290,735		1,290,735
Common stock, no par value; 500,000,000 shares authorized; 224,813,819 and 174,838,967 shares issued and outstanding,				
respectively		17,634,474		16,526,570
Additional paid-in capital		3,672,724		1,472,598
Deficit accumulated prior to the development stage		(1,399,577)		(1,399,577
Deficit accumulated during the development stage		(27,146,931)		(25,439,369)
Total Stockholders' Deficit	_	(5,948,575)	_	(7,034,431)
TOTAL LIABILITIES AND STOCKHOLDERS' DEFICIT	\$	5,091,792	\$	1,240,752

See Notes to Consolidated Financial Statements

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A Development Stage Company) CONSOLIDATED STATEMENTS OF OPERATIONS

		For the Years Ended December 31,			From Inception of the Development Stage on November 20, 1991 through		
	_	2008	_	2007	De	ecember 31, 2008	
Operating Expenses						(Unaudited)	
General and administrative	\$	1,828,727	\$	2,949,885	\$	9,729,285	
Research and development	÷	-	φ	986,584	Ŷ	986,584	
Loss from Operations		(1,828,727)		(3,936,469)		(10,715,869)	
Other Income (Expenses)							
Unrealized gain (loss) on financial instrument		5,469		(147,636)		4,722,632	
Interest income		4,310		4,441		66,915	
Interest expense		(234,470)		(51,929)		(1,472,019)	
Interest expense from amortization of discount on secured promissory note		(36,369)		(250,000)		(286,369)	
Gain on debt restructuring		-		485,137		2,524,787	
Other income		-		-		906,485	
Total Other Income (Expenses)		(261,060)		40,013		6,462,431	
Loss from Continuing Operations Before Minority Interest in Net Loss		(2,089,787)		(3,896,456)		(4,253,438)	
Minority interest in net loss		315,115	_	<u> </u>		315,115	
Loss from Continuing Operations		(1,774,672)		(3,896,456)		(3,938,323)	
Income (Loss) from Discontinued Operations (net of gain on disposal of MDI-P of \$258,809 in 2007)		67,110		(518,428)	_	(22,516,409)	
Net Loss		(1,707,562)		(4,414,884)		(26,454,732)	
Preferred stock dividend from beneficial conversion feature						(692,199)	
Net Loss Applicable to Common Shareholders	\$	(1,707,562)	\$	(4,414,884)	\$	(27,146,931)	
Basic and Diluted Loss per Common Share:							
Loss from Continuing Operations	\$	(0.009)	\$	(0.029)			
Income (Loss) from Discontinued Operations	\$	0.001	\$	(0.004)			
Net loss	\$	(0.008)	\$	(0.033)			
Basic and Diluted Weighted-Average Common Shares Outstanding		207,895,116		134,707,205			

See Notes to Consolidated Financial Statements

-4

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A Development Stage Company) CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' DEFICIT Period From November 20, 1991 (Date of Inception of the Development Stage) through December 31, 2008

	Preferred Sto	ock – Series A	Preferred S	itock – Series B	Comm	on stock	Additional Paid in	Accumulated Deficit Prior to Development	Deficit Accumulated During the Development	Escrow/ Subscription	
	Shares	Amount	Shares	Amount	Shares	Amount	Capital	Stage	Stage	Receivables	Total
Balance at October 31, 1991	-	s -	-	s -	1,750,000	\$ 252,997	\$-	\$ (1,482,514)	s -	\$-	\$ (1,229,517)
Restatement for reverse acquisition of WPI Pharmaceutical, Inc. by Medical Discoveries, Inc.						(252,997)		252,997			-
Shares issued in merger of WPI Pharmaceutical, Inc.											
Medical Discoveries, Inc., \$0.01 per share	<u> </u>	<u> </u>	<u> </u>		10,000,000	135,000	<u> </u>	(170,060)	<u> </u>	<u> </u>	(35,060)
Balance at November 20, 1991 (Date of Inception of Development Stage)					11,750,000	135,000		(1,399,577)			(1,264,577)
Issuance of common stock for:											
Cash						100.000					100.000
1992 - \$0.50 per share 1992 - \$1.50 per share		-		-	200,000 40,000	100,000 60,000	-	-	-	-	100,000 60,000
1992 - \$1.50 per share 1993 - \$0.97 per share	-	-	-	-	542,917	528,500	-	-	-	-	528,500
1993 - \$0.97 per share	-				617,237	739,500				-	739,500
1995 - \$0.67 per share	-	-		-	424,732	283,200	-		-	-	283,200
1996 - \$0.66 per share	-	-	-	-	962,868	635,000	-	-		(60,000)	575,000
1997 - \$0.43 per share	-	-	-	-		135,000	-	-	-	60,000	195,000
1998 - \$0.29 per share	-	-	-	-	2,236,928	650,000	-	-	-	-	650,000
1999 - \$0.15 per share	-	-	-	-	13,334	2,000	-	-	-	-	2,000
2001 - \$0.15 per share	-	-	-	-	660,000	99,000	-	-	-	-	99,000
2003 - \$0.04 per share	-	-	-	-	20,162,500	790,300	-				790,300
2004 - \$0.09 per share	-	-	-	-	20,138,024	1,813,186	-	-	-	-	1,813,186
2005 - \$0.18 per share	-	-		-	1,922,222	281,926	-			-	281,926
Services and Interest 1992 - \$0.50 per share					500,000	250,000					250,000
1992 - \$0.50 per share	-	-		-	251,450	127,900	-			-	127,900
1993 - \$0.51 per share			-	-	800,000	400,000					400,000
1994 - \$1.00 per share	-	-	-	-	239,675	239,675	-	-		-	239,675
1995 - \$0.39 per share	-		-		4,333,547	1,683,846		-		(584,860)	1,098,986
1996 - \$0.65 per share	-	-	-	-	156,539	101,550	-	-	-	-	101,550
1997 - \$0.29 per share	-	-	-	-	12,500	3,625	-	-	-	-	3,625
1998 - \$0.16 per share	-	-	-	-	683,000	110,750	-	-	-	-	110,750
1999 - \$0.30 per share		-	-		100,000	30,000	-	-		-	30,000
2001 - \$0.14 per share	-	-	-	-	1,971,496	284,689	-	-		-	284,689
2002 - \$0.11 per share	-	-	-	-		332,236	-	-	-	-	332,236
2003 - \$0.04 per share	-	-	-	-	694,739	43,395	-	-	-	-	43,395
2004 - \$0.06 per share	-		-	-		66,501			-	-	66,501
2005 - \$0.18 per share		-		-	104,167 435,556	11,312 78,400	-	-			11,312 78,400
2006 - \$0.18 per share Conversion of Debt		-	-	-	435,550	78,400	-				/8,400
1996 - \$0.78 per share					239,458	186,958		-			186,958
1997 - \$0.25 per share	-		-	-	100,000	25,000	-	-		-	25,000
1998 - \$0.20 per share	-		-		283,400	56,680		-			56,680
2002 - \$0.03 per share	-	-	-	-	17,935,206	583,500	-	-	-	-	583,500
2004 - \$0.07 per share	-	-	-	-	9,875,951	650,468	-	-	-	-	650,468
Conversion of preferred stock to common											
stock, 2006	(7,580)	(8,722)	-	-	10,242,424	8,722	-		-	-	-
Other Issuances											
1993 -License - \$0.50 share					2,000,000	1,000,000					1,000,000
1997 - Settlement of contract	-	-	-	-	2,000,000 800,000	200,000			-	-	200,000
1997 - Settlement of contract 1998 - Issuance of common stock from				-	800,000	200,000				· ·	200,000
exercise of warrants, \$0.001 per share					200,000	200					200
2000 - Reversal of shares issued	-	-		-	(81,538)	- 200	-		-	-	200
Escrow and Subscription Receivables					(01,000)						
1996 - Common stock canceled -\$0.34											
per share	-	-	-	-	(1,400,000)	(472,360)	-	-	-	472,360	-
2000 - Issuance for escrow receivable -											
\$0.09 per share	-	-	-	-	5,500,000	500,000	-	-	-	(500,000)	-
2000 - Write-off of subscription											
receivable	-	-	-	-	-	-				112,500	112,500
2000 - Research and development costs	-	-	-	-	-	-	-	-	-	115,400	115,400
2001 - Research and development costs				-		-				132,300	132,300
2001 - Operating expenses	-	-		-	-	-	-		-	25,000	25,000
2004 - Termination of escrow agreement					(2,356,200)	(227,300)				227,300	
agreement	-	-	-		(2,330,200)	(227,500)	-			227,500	-

(Continued)

See Notes to Consolidated Financial Statements

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A Development Stage Company) CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' DEFICIT - (Continued) Period From November 20, 1991 (Date of Inception of the Development Stage) through December 31, 2008

Description and Warrants Image: constraint of the proof bit A port of the proof bit A			Stock Series A		Stock - Series B	Comme		Additional Paid in	Accumulated Deficit Prior to Development	Deficit Accumulated During the Development	Escrow/ Subscription	
Description and Woman - 57.556 21,99 - - 21,99 1990 Named spring price 30, 10, get data - 23,53,08 - - 23,55,00 1990 Named spring price 30, 10, get data - 23,53,08 - - 23,55,00 1990 - - 23,53,08 - - 23,55,00 1990 - - 23,53,08 - - 23,55,00 1990 - - 199,000 - 199,000 - 199,000 1990 - - - 199,000 - 199,000 - 199,000 - 199,000 - 199,000 - 199,000 - 199,000 - 199,000 199,000 - 199,000 - 199,000 - - 199,000 - - 199,000 - - 199,000 - - 199,000 - - 199,000 - - - 199,0		Snares	Amount	Snares	Amount	Shares	Amount	Capital	Stage	Stage	Receivables	1 otai
1997-5022 product oper SU: 14 produ	Balance carried forward	(7,580)	\$ (8,722)	-	\$-	117,749,868	\$ 12,528,359	s -	\$ (1,399,577)	s -	s -	\$ 11,120,060
1999 Named option price 30: 491 per consistence of the Services 10000 2,000 - - 2,000 1990 - - 2,000 - - 2,000 1990 - - 2,000 - - 2,000 1990 - - 2,000 - - 2,000 1990 - - 100,000 2,000 - - 100,000 2003 - - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - 100,000 - - - 100,000 - - - 100,000 - - - 100,000 - - - 100,000 - - - 100,000 - - -<												
min - 17000 24000 - - 24000 Via COpon Index for Series - 235.833 - - 235.833 199 - - 190.357 - - 199.437 2001 - 190.437 - - 199.437 2001 - 190.438 - 199.437 2003 - - 190.438 - 199.437 2004 - - 102.548 - 102.548 2005 - - 102.548 - 102.549 2006 - - 102.546 - 102.546 2007 - - 102.546 - 10.596 2008 - - 20.000 - - 10.596 2008<-Instance of proferent acks and ack and				-		87,836	21,959	-		-		21,959
Value of Optime Insued for Service 998 .						170.000	24.000					24.000
199 . . . 194,57 . . 194,57 2001 . . . 194,657 . . 194,657 2003 194,657 . . 194,657 2004 195,000 . . . 195,000 . . . 195,000 . . . 195,000 195,000 .			-	-	-	170,000	24,000	-	-	_	-	24,000
30h - - - 194.465 - 194.465 301 - - 144.958 - 194.558 304 - 144.958 - 194.558 304 - 162.500 - 167.500 206e - 162.954 - 162.954 1984 - 102.054 - 102.954 2006 - - 20.000 - - 102.954 2007 - - 20.000 - - 102.954 2008 Samara for cah 12.000 53.334 - 390.000 68.45 477.521 - - 107.000 2008 Samara for cah 30.000 - - 662.199 - - 107.920 692.199 (692.199) - - - 107.920 - 107.920 - 107.920 - - - - 107.920 - - - - <td></td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td></td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td></td>		-	-	-	-	-		-	-	-	-	
202 - - - - 124,683 - - 124,683 - - 124,683 - - 255,000 - 255,000 - 255,000 - 755,000 - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - 755,000 - - 755,000 - - 755,000 - 755,000 - 755,000 - - - 755,000 - 12,000,00 - - - - - 755,000 - - - - - 12,00,015 - - -<				-	-	-	196,587	-		-		
203 -		-	-	-	-	-	-		-	-	-	
		-	-	-	-	-			-	-		
Date .				-								
Other 1984 - Calc contributed - - 102.964 - - - 102.964 1984 - Calc contributed - - 20.000 - - 20.000 2004 - Smart spectral spectra spectral spectral spectral spectral spectral spectr		-	-	-	-	-	-		-	-	-	
193	Other											
In safely define standarding - - - 20,000 - - - 20,000 2004 - issuance of preferent sock and warrans for cash 12,000 23,334 - - 350,000 68,844 477,821 - - 10,000 2005 - issuance of preferent sock and warrans for cash 50,000 -		-		-	-	-	102,964		-	-		102,964
Balance of preferred stock and warmans for cosh 12,000 58,845 477,21 - - 100,000 2004 - Converbile preferred stock and stock							20.000					20.000
warms for cash 12.000 \$23,354 - - \$500.00 68,845 477,821 - - 1.070.000 2004 - Coversion divided - - - 602.199 - (602.199) -	to satisfy debt restructuring		-			-	20,000		-	-		20,000
2041 - Conventible prediction conversion divided .		12 000	523 334			350.000	68 845	477 821				1.070.000
beneficial conversion dividend - - - 602,199 - - - 2005 - Issues of preferred stock and warrants for cash 30,000 - <td< td=""><td></td><td>12,000</td><td>525,554</td><td>-</td><td></td><td>550,000</td><td>00,045</td><td>477,021</td><td></td><td>-</td><td></td><td>1,070,000</td></td<>		12,000	525,554	-		550,000	00,045	477,021		-		1,070,000
2005		-	-	-	-	-	-	692,199	-	(692,199)	-	-
205 -												
Interview .		30,000	-	-	-	-	-	-	-	-	-	-
Not loss from inception through December 31, 2006												
2006 .				-		-		(2,435,713)		-		(2,435,713)
Balance at December 31, 2006 (Unaudited) 34,420 514,612 - 118,357,704 15,299,017 1,056,020 (1,399,577) (21,024,485) - (5,554,413) Issuance of common stock for Ciobal Clean Energy Holdings, LLC -										(20, 222, 286)		(20,222,286)
(Unaudited) 34,420 514,612 - 118,357,704 15,299,017 1,056,020 (1,399,577) (21,024,485) - (5,554,413) Issuance of common stock for Global Clean Energy Holdings, LLC - - - - 96,554 Issuance of Series B preferred stock for cash, net of offering costs - 13,000 1,290,735 - - - 12,90,735 Conversion of preferred stock to common stock for costs in sunce of options - - 29,652 - - 29,652 Stare-based compensation from issuance of common stock, S0027 per share - - 29,652 - - 29,652 Stare-based compensation from issuance of common stock, Mol27 per share - - 4,357,298 117,647 - - 117,647 Notizition of Share-based compensation for common stock hall in serve - - 510,248 - - 117,647 Augusting in the serve - - - 510,248 - - - 2,012,488 Augustind in dinserupon satisfaction of setis for thy year endel Decembe	2000									(20,332,280)		(20,332,280)
(Unaudited) 34,420 514,612 - 118,357,704 15,299,017 1,056,020 (1,399,577) (21,024,485) - (5,554,413) Issuance of common stock for Global Clean Energy Holdings, LLC - - - - 96,554 Issuance of Series B preferred stock for cash, net of offering costs - 13,000 1,290,735 - - - 12,90,735 Conversion of preferred stock to common stock for costs in sunce of options - - 29,652 - - 29,652 Stare-based compensation from issuance of common stock, S0027 per share - - 29,652 - - 29,652 Stare-based compensation from issuance of common stock, Mol27 per share - - 4,357,298 117,647 - - 117,647 Notizition of Share-based compensation for common stock hall in serve - - 510,248 - - 117,647 Augusting in the serve - - - 510,248 - - - 2,012,488 Augustind in dinserupon satisfaction of setis for thy year endel Decembe	Balance at December 31, 2006											
Energy Holdings, LLC - - - 36,540,146 986,584 - - - 986,584 net of offering costs - - 10,983,521 - - - 1,220,735 Coversion of prefered stock to common stock (5,492) - - - 29,652 - - 29,652 Share-based compensation from issuance - - 4,357,298 117,647 - - - 10,983,521 Share-based compensation from issuance - - 4,357,298 117,647 - - 117,647 common stock koll in sectow - - - 510,248 - - 510,248 Release of scrowed sharse upon satisfaction of - - - - - - - 510,248 Release of scrowed sharse upon satisfaction of -		34,420	514,612	-	-	118,357,704	15,299,017	1,056,020	(1,399,577)	(21,024,485)	-	(5,554,413)
Energy Holdings, LLC - - - 36,540,146 986,584 - - - 986,584 net of offering costs - - 10,983,521 - - - 1,220,735 Coversion of prefered stock to common stock (5,492) - - - 29,652 - - 29,652 Share-based compensation from issuance - - 4,357,298 117,647 - - - 10,983,521 Share-based compensation from issuance - - 4,357,298 117,647 - - 117,647 common stock koll in sectow - - - 510,248 - - 510,248 Release of scrowed sharse upon satisfaction of - - - - - - - 510,248 Release of scrowed sharse upon satisfaction of -												
Issume of Series B preferred stock for cash, net of offering costs - 13,000 1,290,735 29,652 29,653 Share-based compensation fom issume of options issume of options issume of common stock, 80,027 per share nomization of share-based compensation for issume common stock 80,027 per share nomization of share-based compensation for issume subscriptions issume of common stock, 80,027 per share nomization of share-based compensation for issume subscriptions issume issu												
net of officing costs - 13,000 1,290,735 - - 1,290,735 Share-based compensation from issuance - 10,983,521 - - 29,652 Share-based compensation from issuance - 29,652 - - 29,652 Share-based compensation from issuance - 4,357,298 117,647 - - 117,647 nortization of share-based compensation for - - 510,248 - - 510,248 - - 510,248 - - - - 510,248 -		-	-	-	-	36,540,146	986,584	-	-	-	-	986,584
Conversion of prefered stock to common stock (5,492) - 10,983,521 - - 29,652 - 29,652 Share-based compensation from issuance of options - - 4,357,298 117,647 - - 117,647 Amorization of share-based compensation for common stock held in escrow - - - 510,248 - - 510,248 - - 510,248 Release of secrowed shares upon satisfaction of underlying milestons - - 4,567,518 123,322 - - - - - 510,248 - - 510,248 - - 510,248 - - 510,248 - <				13 000	1 290 735			_				1 290 735
Share-based compensation from issuance of options 29,652 . . 29,652 Share-based compensation from issuance of common stock, \$0,027 per share . <		(5.492)			1,270,755	10.983.521		-		-		1,270,755
Share-based compensation from issuance - - 4,357,298 117,647 - - 117,647 Amortization of share-based compensation for orstock, 50.027 per share - - 510,248 - - 510,248 Common stock, 600 (25,002) - - - - 510,248 - - 510,248 Release of escrowed shares topon suitafication of turn-topon suitafication of accounts -	Share-based compensation from issuance	(,,,,,										
of common stock, \$0.027 pr share - - 4.357,298 117,647 - - 117,647 Amorization of share-based compensation for common stock held in escrow - - - 510,248 - - 510,248 Release of secrowed share-based compensation for underlying milestones - - 4,567,518 123,322 (123,322) -<		-		-	-	-	-	29,652	-	-	-	29,652
Amortization of share-based compensation for common stock held in escrow .												
common stock held in escrow - - 510,248 - - 510,248 Release of escrowed shares upon satisfaction of underlying milestones - - 4,567,518 123,322 (123,322) - <t< td=""><td></td><td>-</td><td>-</td><td>-</td><td>-</td><td>4,357,298</td><td>117,647</td><td>-</td><td>-</td><td>-</td><td>-</td><td>117,647</td></t<>		-	-	-	-	4,357,298	117,647	-	-	-	-	117,647
Release of serrowed shares upon satisfaction of underlying milestones - - 4,567,518 123,322 (123,322) - <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>510 249</td> <td></td> <td></td> <td></td> <td>510 249</td>								510 249				510 249
underlying milestones - - 4,567,518 123,322 (123,322) - - - Adjustment of outstanding shares - - 32,780 - <				-				510,248	-			510,248
Adjustment of outstanding shares - - 32,780 -		-		-		4,567,518	123,322	(123,322)		-		-
Balance at December 31, 2007 28,928 514,612 13,000 1,290,735 174,838,967 16,526,570 1,472,598 (1,399,577) (25,439,369) - (7,034,431) Reclassification of financial instrument to equity - - - 2,161,045 - - 2,161,045 Exchange of Series A preferred stock for common stock (28,928) (514,612) - 28,927,000 514,612 - 100,000 - - - 160,934 - - 160,934 - - 160,934 - - 184,146	Adjustment of outstanding shares	-	-	-	-	32,780		-	-	-	-	-
Reclassification of financial instrument to	Net loss for the year ended December 31, 2007				-	-		-		(4,414,884)		(4,414,884)
Reclassification of financial instrument to												
equity - - - - 2,161,045 - - 2,161,045 Exchange of Series A preferred stock for common stock (28,928) (514,612) - - 28,927,000 514,612 - 160,934 - - - 160,934 - - - 187,293 <td></td> <td>28,928</td> <td>514,612</td> <td>13,000</td> <td>1,290,735</td> <td>174,838,967</td> <td>16,526,570</td> <td>1,472,598</td> <td>(1,399,577)</td> <td>(25,439,369)</td> <td>-</td> <td>(7,034,431)</td>		28,928	514,612	13,000	1,290,735	174,838,967	16,526,570	1,472,598	(1,399,577)	(25,439,369)	-	(7,034,431)
Exchange of Series A preferred stock for common stock (28,928) (514,612) - - 28,927,000 514,612 - 160,934 - - - 160,934 - - 160,934 - - 184,146 - - 184,146 - - 184,146 - - 184,146 - - 187,293 - - 187,293 - - 187,293 - </td <td></td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>2,161.045</td> <td>-</td> <td>-</td> <td>-</td> <td>2.161.045</td>		-	-	-	-	-	-	2,161.045	-	-	-	2.161.045
common stock (28,928) (514,612) - 28,927,000 514,612 - 160,934 - - 184,146 - - 184,146 - - 184,146 - - 184,146 - - 187,293 - - 187,293 - - 187,293 - -												
Issuance of common stock for eash at \$0.036 per share 2,777,778 100,000 100,000 Suance of variants in sutsification of accounts payable and amendment of note payable 160,934 160,934 160,934 Share-based compensation from issuance of options 184,146 184,146 184,146 Amortization of share-based compensation for common stock held in escrow 187,293 187,293 Release of servowed shares upon satisfaction of underlying milestones 18,270,074 493,292 (493,292) 187,293 Net loss for the year ended December 31, 2008 (1,707,562)		(28.928)	(514.612)	-		28,927,000	514.612	-	-	-		-
Issuance of warrants in suitStiction of accounts payable and amendment of note payable of options o	Issuance of common stock for cash at \$0.036	(, , , ,										
payable and amendment of note payable 160,934 160,934 Share-based compensation from issuance of options 184,146 184,146 Amortization of share-based compensation for common stock held in escrow 187,293 187,293 Release of servowd shares upon satisfaction of underlying milestones 18,270,074 493,292 (493,292)		-	-	-	-	2,777,778	100,000	-	-	-	-	100,000
Share-based compensation from issuance of options - - 184,146 - - 184,146 Amortization of share-based compensation for common stock held in escrow - - - 187,293 - - 187,293 Release of escrowed shares upon satisfication of underlying milestones - - 187,292 (493,292) -												
of options				-	-	-		160,934				160,934
Amortization of share-based compensation for common stock held in escrow								184 146				184 146
common stock held in escrow			-	-	-	-	-	104,140	-	-	-	104,140
Release of escrowed shares upon satisfaction of underlying milestones - - 18,270,074 493,292 (493,292) Net loss for the year ended December 31, 2008 - - - (1,707,562) - (1,707,562)			-	-	-	-	-	187,293	-	-	-	187,293
Net loss for the year ended December 31, 2008	Release of escrowed shares upon satisfaction of											
			-	-	-	18,270,074	493,292	(493,292)	-	-	-	-
Balance at December 31, 2008 - <u>\$ - 13,000</u> <u>\$ 1,290,735</u> <u>224,813,819</u> <u>\$ 17,634,474</u> <u>\$ 3,672,724</u> <u>\$ (1,399,577)</u> <u>\$ (27,146,931)</u> <u>\$ - </u> <u>\$ (5,948,575)</u>	Net loss for the year ended December 31, 2008	<u> </u>								(1,707,562)		(1,707,562)
Balance at December 31, 2008 - S - 15,000 S 1,290,755 Z24,815,819 S 17,654,474 S 5,672,724 S (1,599,577) S (27,146,931) S - S (5,948,575)	D 1 (D 1 01 0000		0	12.000	6 1 200 525	224 012 012	6 17 (34 /7)	6 2 (72 72)	0 (1 200 577)	6 (07 14/ C21)	6	6 (6 040 675)
	Balance at December 31, 2008		3 -	13,000	\$ 1,290,735	224,813,819	\$ 17,634,474	\$ 3,672,724	\$ (1,399,577)	\$ (27,146,931)	3 -	\$ (5,948,575)

See Notes to Consolidated Financial Statements

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A Development Stage Company) CONSOLIDATED STATEMENTS OF CASH FLOWS

	For the Years Ended December 31,				From Inception of the Development Stage on November 20, 1991 through			
		2008		2007		mber 31, 2008		
					ת	J naudited)		
Cash Flows From Operating Activities Net loss	\$ (1,707,562)	¢	(4,414,884)	¢	(26,454,732)		
Adjustments to reconcile net loss to net cash used in operating activities	ф (1,707,302)	ф	(4,414,884)	φ	(20,434,732)		
Foreign currency transaction loss (gain)		(107,369)		296,370		250,022		
Gain on debt restructuring		(107,207)		(485,137)		(2,524,787)		
Share-based compensation for services, expenses, litigation, and research and development		371,439		3,118,021		12,714,180		
Commitment for research and development obligation		-		-		2,378,445		
Depreciation		1,365		10,494		139,031		
Reduction of escrow receivable from research and development		-		-		272,700		
Unrealized loss (gain) on financial instrument		(5,469)		147,636		(4,722,632)		
Interest expense from amortization of discount on secured promissory note		36,369		250,000		286,369		
Minority interest in net loss		(315,115)		-		(315,115)		
Reduction of legal costs		-		-		(130,000)		
Write-off of subscriptions receivable		-		-		112,500		
Impairment loss on assets		-		-		9,709		
Gain on disposal of assets, net of losses		-		(258,809)		(228,445)		
Write-off of receivable		-		-		562,240		
Note payable issued for litigation		-		-		385,000		
Changes in operating assets and liabilities						,		
Accounts receivable		-		-		(7,529)		
Other current assets		(80,642)		(51,073)		(131,715)		
Accounts payable and accrued expenses		802,314		678,104		5,020,326		
Net Cash Used in Operating Activities	(1,004,670)		(709,278)		(12,384,433)		
Cash Flows From Investing Activities		1,004,070)	_	(70),270)	-	(12,504,455)		
Plantation development costs	(1,787,916)		(308,777)		(2,096,693)		
Purchase of property and equipment	((518,903)		(308,777)		(740,237)		
Proceeds from disposal of assets		(318,903)		310.000		310,000		
Change in deposits		(2,691)		510,000		(53,791)		
Issuance of note receivable		(2,091)		-		(313,170)		
Payments received on note receivable		-		-		130,000		
•		-		1 222				
Net Cash Provided by (Used in) Investing Activities	(2,309,510)	_	1,223		(2,763,891)		
Cash Flows From Financing Activities								
Proceeds from common stock, preferred stock, and warrants for cash		175,000		1,215,735		11,424,580		
Proceeds from issuance of preferred membership in GCE Mexico I, LLC		2,415,151		-		2,415,151		
Contributed equity		-		-		131,374		
Proceeds from notes payable and related warrants		260,000		350,000		1,946,613		
Payments on notes payable		(50,000)		(100,000)		(951,287)		
Proceeds from convertible notes payable		-		-		571,702		
Payments on convertible notes payable		-		-	_	(98,500)		
Net Cash Provided by Financing Activities		2,800,151		1,465,735		15,439,633		
Net Increase (Decrease) in Cash and Cash Equivalents		(514,029)		757,680		291,309		
Cash and Cash Equivalents at Beginning of Year		805,338		47,658		-		
Cash and Cash Equivalents at End of Year		291,309		805,338		291,309		
Supplemental Disclosures of Cash Flow Information:	0	10.007	¢	10.11-				
Cash paid for interest	\$	13,024	\$	12,146				
Noncash Investing and Financing Activities:			¢					
Reclassification of financial instrument to permanent equity			\$	-				
Acquisition of land in exchange for mortgage note payable		2,051,282		-				
Exchange of Series A preferred stock for common stock								
		514,612		-				
Release of common stock held in escrow		493,292		123,322				
Issuance of warrants in satisfaction of accounts payable		124,565		-				
Accrual of return on minority interest		138,014		-				
Equipment depreciation capitalized to plantation development costs		20,638		-				

See Notes to Consolidated Financial Statements

NOTE A - ORGANIZATION AND SIGNIFICANT ACCOUNTING POLICIES

<u>History</u>

Medical Discoveries, Inc. was incorporated under the laws of the State of Utah on November 20, 1991. Effective as of August 6, 1992, the Company merged with and into WPI Pharmaceutical, Inc., a Utah corporation ("WPI"), pursuant to which WPI was the surviving corporation. Pursuant to the MDI-WPI merger, the name of the surviving corporation was changed to Medical Discoveries, Inc. ("MDI"). MDI's initial purpose was the research and development of an anti-infection drug know as MDI-P.

On March 22, 2005, MDI formed MDI Oncology, Inc., a Delaware corporation, as a wholly-owned subsidiary to acquire and operate the assets and business associated with the Savetherapeutics transaction. With this transaction, MDI acquired the SaveCream technology and carried on the research and development of this drug candidate. As discussed in Note M, MDI made the decision in 2007 to discontinue further development of these two drug candidates and sell these technologies.

On September 7, 2007, MDI entered into a share exchange agreement pursuant to which it acquired all of the outstanding ownership interests in Global Clean Energy Holdings, LLC, discussed further in Note C. Global Clean Energy Holdings, LLC was an entity that had certain trade secrets, know-how, business plans, term sheets, business relationships, and other information relating to the start-up of a business related to the cultivation and production of seed oil from the seed of the Jatropha plant. With this transaction, MDI commenced the research and development of a business whose purpose will be providing feedstock oil intended for the production of bio-diesel.

On January 29, 2008, a meeting of shareholders was held and, among other things, the name Medical Discoveries, Inc. was changed to Global Clean Energy Holdings, Inc. (the "Company").

Effective April 23, 2008, the Company entered into a limited liability company agreement to form GCE Mexico I, LLC (GCE Mexico) along with six unaffiliated investors. The Company owns 50% of the common membership interest of GCE Mexico and five of the unaffiliated investors own the other 50% of the common membership interest. Additionally, a total of 1,000 preferred membership units were issued to two of the unaffiliated investors. GCE Mexico owns a 99% interest in Asideros Globales Corporativo, (Asideros) a corporation newly organized under the laws of Mexico, and the Company owns the remaining 1% directly. GCE Mexico was organized primarily to, among other things, acquire land in Mexico through Asideros for the cultivation of the Jatropha plant.

Principles of Consolidation

The consolidated financial statements include the accounts of Global Clean Energy Holdings, Inc., its subsidiaries, and the variable interest entities of GCE Mexico and Asideros. All significant intercompany transactions have been eliminated in consolidation.

Financial Accounting Standards Board Interpretation No. 46 (revised December 2003), *Consolidation of Variable Interest Entities*, (FIN 46(R)), requires that if an entity is the primary beneficiary of a variable interest entity (VIE), the entity should consolidate the assets, liabilities and results of operations of the VIE in its consolidated financial statements. Global Clean Energy Holdings, Inc. considers itself to be the primary beneficiary of GCE Mexico and Asideros, and accordingly, has consolidated these entities since April 2008, with the equity interests of the unaffiliated investors in GCE Mexico presented as Minority Interests in the accompanying consolidated financial statements.

Development Stage Company

The Company has not yet obtained substantial revenue from its planned principal operations and is, therefore, considered a development stage company as defined in Statement of Financial Accounting Standards (SFAS) No. 7, Accounting and Reporting by Development Stage Enterprises.

Cash and Cash Equivalents

For purposes of the statement of cash flows, the Company considers all highly liquid debt instruments maturing in three months or less to be cash equivalents.

Concentration of Credit Risk

The Company's financial instruments that are exposed to concentration of credit risk consist primarily of cash and cash equivalents on deposit in excess of federallyinsured limits in the aggregate amount of \$27,891 at December 31, 2008. The Company has maintained its cash balances at what management considers to be high credit-quality financial institutions.

Property and Equipment

As described in Note D, substantially all property and equipment relate to the development of a plantation to cultivate the *Jatropha Curcas* plant. Property and equipment are stated at cost. Depreciation of office equipment is computed using the straight-line method over estimated useful lives of 5 years. Plantation equipment is depreciated using the straight-line method over estimated useful lives of 5 to 15 years and is currently being capitalized as part of plantation development costs. Plantation development costs are being accumulated in the balance sheet during the development period and will be accounted for in accordance with Statement of Position 85-3, *Accounting by Agricultural Producers and Agricultural Cooperatives* (SOP 85-3). Plantation development costs are not currently being depreciated. Under the provisions of SOP 85-3, land developments and other improvements with indefinite lives are capitalized and not depreciated. Other developments that have a limited life and intermediate-life plants that have growth and production cycles of more than one year are depreciated over their respective lives once they are placed in service. Upon completion of the plantation development, the development costs having a limited life and the costs of cultivating the Jatropha plants will be depreciated over the useful lives of the related assets. Land, plantation development costs, and plantation equipment are located in Mexico.

Except for costs incurred during the development period of the plantation, normal maintenance and repair items are charged to costs and expensed as incurred. During the development period, maintenance, repairs, and depreciation of plantation equipment have been capitalized as part of the plantation development costs. The cost and accumulated depreciation of property and equipment sold or otherwise retired are removed from the accounts and gain or loss on disposition is reflected in results of operations.

In accordance with SFAS No. 144, Accounting for the Impairment or Disposal of Long-Lived Assets, the carrying value of intangible assets and other long-lived assets is reviewed on a regular basis for the existence of facts or circumstances that may suggest impairment. The Company recognizes impairment when the sum of the expected undiscounted future cash flows is less than the carrying amount of the asset. Impairment losses, if any, are measured as the excess of the carrying amount of the asset over its estimated fair value.

Income Taxes

The Company utilizes the liability method of accounting for income taxes. Under the liability method, deferred tax assets and liabilities are determined based on differences between financial reporting and tax bases of assets and liabilities and the carryforward of operating losses and tax credits, and are measured using the enacted tax rates and laws that will be in effect when the differences are expected to reverse. A valuation allowance against deferred tax assets is recorded when it is more likely than not that such tax benefits will not be realized. Research tax credits are recognized as utilized.



Revenue Recognition

Revenue is recognized in accordance with Securities and Exchange Commission Staff Accounting Bulletin No. 104, *Revenue Recognition in Financial Statements*. Revenue is recognized when all of the following criteria are met: persuasive evidence of an arrangement exists; delivery has occurred or services have been rendered; the seller's price to the buyer is fixed or determinable; collectibility is reasonably assured; and title and the risks and rewards of ownership have transferred to the buyer.

Research and Development

Prior to the discontinuation of its bio-pharmaceutical business as discussed in Note M, research and development had been the principal function of the Company. For fiscal years ended December 31, 2006 and earlier, research and development expense included certain costs which were directly associated with the Company's research and development of the Company's anti-infective pharmaceutical, MDI-P, as well as the purchase of the intellectual property assets of Savetherapeutics AG. For the year ended December 31, 2007, research and development costs related to the exchange of common stock for the trade secrets, know-how, etc. of Global Clean Energy Holdings, LLC (See Note C). Research and development costs totaled \$0 and \$986,584 for the years ended December 31, 2007, respectively. For years prior to the discontinuation of its bio-pharmaceutical business, research and development costs are included in loss from discontinued operations.

Foreign Currency Translation

The Company's functional and reporting currency is the United States dollar. Monetary assets and liabilities denominated in foreign currencies are translated using the exchange rate prevailing at the balance sheet date. Gains and losses arising on translation or settlement of foreign currency denominated transactions or balances are included in the determination of income or loss. Foreign currency transactions are primarily undertaken in Euros. Foreign currency balances denominated in Euros relate to the discontinued bio-pharmaceutical business. Consequently, foreign currency gains and losses have been included in loss from discontinued operations. The Company has not entered into derivative instruments to offset the impact of foreign currency fluctuations.

Fair Value of Financial Instruments

The Company estimates that the fair value of all financial instruments at December 31, 2008 do not differ materially from the aggregate carrying values of its financial instruments recorded in the accompanying balance sheet. The estimated fair value amounts have been determined by the Company using available market information and appropriate valuation methodologies. Considerable judgment is required in interpreting market data to develop the estimates of fair value, and accordingly, the estimates are not necessarily indicative of the amounts that the Company could realize in a current market exchange.

Estimates

Management uses estimates and assumptions in preparing financial statements. Those estimates and assumptions affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities, and reported revenues and expenses. Significant estimates used in preparing these financial statements include a) those assumed in determining the valuation of common stock, warrants, and stock options, b) estimated useful lives of plantation equipment, and c) undiscounted future cash flows for purpose of evaluating possible impairment of long-term assets. It is at least reasonably possible that the significant estimates used will change within the next year.



Basic and Diluted Loss per Share

Basic loss per share is computed on the basis of the weighted-average number of common shares outstanding during the year. Diluted loss per share is computed on the basis of the weighted-average number of common shares and all dilutive potentially issuable common shares outstanding during the year. Common stock issuable upon conversion of debt and preferred stock, common stock held in escrow, stock options and stock warrants have not been included in the loss per share for 2008 and 2007 as they are anti-dilutive. The potentially issuable common shares as of December 31, 2008 and 2007 are as follows:

	Decem	December 31,		
	2008	2007		
Convertible notes	128,671	128,671		
Convertible preferred stock - Series A	-	57,856,000		
Convertible preferred stock - Series B	11,818,181	11,818,181		
Warrants	29,742,552	31,033,379		
Compensation-based stock options and warrants	51,809,083	44,883,000		
Common stock held in escrow	4,567,519	22,837,593		
	98,066,006	168,556,824		

Stock Based Compensation

The Company recognizes compensation expense for stock-based awards expected to vest on a straight-line basis over the requisite service period of the award based on their grant date fair value. The Company estimates the fair value of stock options using a Black-Scholes option pricing model which requires management to make estimates for certain assumptions regarding risk-free interest rate, expected life of options, expected volatility of stock and expected dividend yield of stock.

Reclassifications

Certain amounts from the 2007 consolidated balance sheet have been reclassified in the current presentation to conform to the 2008 presentation of current liabilities. These reclassifications had no effect on the total amount of current liabilities or the amount of stockholders' deficit.

Recently Issued Accounting Statements

In September 2006, the Financial Accounting Standards Board (FASB) issued SFAS No. 157, *Fair Value Measurements* (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value in generally accepted accounting principles, and expands disclosures about fair value measurements. This statement is effective for the Company's fiscal year beginning January 1, 2008 for financial assets and liabilities and January 1, 2009 for non-financial assets and liabilities. The adoption of SFAS 157 for financial assets and liabilities on January 1, 2008 did not have a material impact on the Company's financial statements. Management is currently evaluating the impact of SFAS 157 for non-financial assets and liabilities, if any, on the reporting of its financial position and results of operations.

In December 2007, the FASB issued SFAS No. 141 (revised 2007), *Business Combinations* (SFAS 141(R)), which replaces SFAS 141, *Business Combinations*. SFAS 141(R) retains the underlying concepts of SFAS 141 in that all business combinations are still required to be accounted for at fair value under the acquisition method of accounting, but SFAS 141(R) changed the method of applying the acquisition method in a number of significant aspects. Acquisition costs will generally be expensed as incurred; noncontrolling interests will be valued at fair value at the acquisition date; in-process research and development will be recorded at fair value as an indefinite-lived intangible asset at the acquisition date; restructuring costs associated with a business combination will generally be expensed subsequent to the acquisition date; and changes in deferred tax asset valuation allowances and income tax uncertainties after the acquisition date generally will affect income tax expense. SFAS 141(R) is effective on a prospective basis for all business combinations for which the acquisition date is on or after the beginning of the first annual period subsequent to December 15, 2008, with the exception of the accounting for valuation allowances on deferred taxes and acquired tax contingencies. SFAS 141(R) amends SFAS 141(R) would also apply the provisions of SFAS 141(R). Early adoption is not permitted. Management is currently evaluating the effects, if any, that SFAS 141(R) may have on the Company's financial statements. Management does not expect that it will have any immediate effect on the Company's financial statements; however, the revised standard will govern the accounting for any future business combinations that the Company may enter into.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS 160). This statement is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008, with earlier adoption prohibited. This statement requires the recognition of a noncontrolling interest (minority interest) as equity in the consolidated financial statements and separate from the parent's equity. The amount of net income attributable to the noncontrolling interest will be included in consolidated net income on the face of the income statement. It also amends certain of ARB No. 51's consolidation procedures for consistency with the requirements of SFAS 141(R). This statement also includes expanded disclosure requirements, if SFAS 160 were effective, the minority interest in the consolidated balance sheet would be presented as noncontrolling interest in Owners' Equity (Deficit), the minority interests of the Company and of the noncontrolling interests.

In March 2008, the FASB issued SFAS No. 161, *Disclosures about Derivative Instruments and Hedging Activities, an amendment of FASB Statement No. 133* (SFAS 161). SFAS 161 requires enhanced disclosures about an entity's derivative and hedging activities. Entities will be required to provide enhanced disclosures about: (a) how and why an entity uses derivative instruments; (b) how derivative instruments and related hedge items are accounted for under SFAS No. 133 and its related interpretations; and (c) how derivative instruments and related hedge items affect an entity's financial position, financial performance and cash flows. The provisions of SFAS 161 are effective January 1, 2009. Management is currently evaluating the impact of SFAS 161 on the Company's financial statements.

In June 2008, the FASB ratified EITF Issue No. 07-5, *Determining Whether an Instrument (or an Embedded Feature) Is Indexed to an Entity's Own Stock*(EITF 07-5). EITF 07-5 provides that an entity should use a two step approach to evaluate whether an equity-linked financial instrument (or embedded feature) is indexed to its own stock, including evaluating the instrument's contingent exercise and settlement provisions. It also clarifies on the impact of foreign currency denominated strike prices and market-based employee stock option valuation instruments on the evaluation. EITF 07-5 is effective for fiscal years beginning after December 15, 2008. Management is currently evaluating the impact of SFAS 161 on the Company's financial statements.

NOTE B — BASIS OF PRESENTATION AND GOING CONCERN

The accompanying consolidated financial statements have been prepared assuming that the Company will continue as a going concern. As shown in the accompanying financial statements, the Company incurred a net loss applicable to common shareholders of \$1,707,562 during the year ended December 31, 2008, and has incurred losses applicable to common shareholders since inception of the development stage of \$27,146,931. The Company also used cash in operating activities of \$1,004,670 during the year ended December 31, 2008. At December 31, 2008, the Company has negative working capital of \$6,604,039 and a stockholders' deficit of \$5,948,575. Those factors raise substantial doubt about the Company's ability to continue as a going concern.

The Company discontinued its former bio-pharmaceutical business during the quarter ended March 31, 2007. Management plans to meet its cash needs through various means including selling assets related to its former bio-pharmaceutical business, securing financing, entering into joint ventures, and developing a new business model. In order to fund its new operations related to the cultivation of the Jatropha plant, the Company sold Series B preferred stock during the quarter ended December 31, 2007 in the amount of \$1,300,000 and issued a secured promissory note under which the Company has borrowings of \$460,000 as of December 31, 2008. The Company is developing a new business operation to participate in the rapidly growing bio-diesel industry. The Company continues to expect to be successful in this new venture, but there is no assurance that its business plan will be economically viable. The ability of the Company to continue as a going concern.

NOTE C — JATROPHA BUSINESS VENTURE

Having agreed to discontinue its bio-pharmaceutical operations and dispose of the related assets, the Company considered entering into a number of other businesses that would enable it to be able to provide the shareholders with future value. The Company's Board of Directors decided to develop a business to produce and sell seed oils, including seed oils harvested from the planting and cultivation of the *Jatropha curcas* plant, for the purpose of providing feedstock oil intended for the generation of methyl ester, otherwise known as bio-diesel (the "Jatropha Business"). The Company's Board concluded that there was a significant opportunity to participate in the rapidly growing biofuels industry, which previously was mainly driven by high priced, edible oil-based feedstock. In order to commence its new Jatropha Business, the Company entered into various transactions during September and October of 2007, including: (i) hired Richard Palmer, an energy consultant, and a member of Global Clean Energy Holdings LLC ("Global") to act as its new President, Chief Operating Officer and future Chief Executive Officer, (ii) engaged Mobius Risk Group, LLC, a Texas company engaged in providing energy risk advisory services, to provide it with consulting services related to the development of the Jatropha Business, (iii) acquired certain trade secrets, know-how, business plans, term sheets, business relationships, and other information relating to the cultivation and production of seed oil from the Jatropha plant for the production of bio-diesel from Global, and (iv) engaged Corporativo LODEMO S.A DE CV to assist with the development of the Jatropha Business in Mexico. Subsequent to entering into these transactions, the Company agreement to form GCE Mexico I, LLC (GCE Mexico). In August 2008 the Company terminated the agreement with Mobius Risk Group, LLC. Through Asideros Globales Corporativo (Asideros), a Mexican corporation of which GCE Mexico holds a 99% equity interest and Global Clean Energy Holdings, Inc. holds a 1% eq

Share Exchange Agreement

The Company entered into a share exchange agreement (the Global Agreement) pursuant to which the Company acquired all of the outstanding ownership interests in Global Clean Energy Holdings, LLC, a Delaware limited liability company (Global), on September 7, 2007 from Mobius Risk Group, LLC (Mobius) and from Richard Palmer (Mr. Palmer). Mr. Palmer owns a 13.33% equity interest in Mobius and, as described further in this Note, became the Company's new President and Chief Operating Officer in September 2007 and its Chief Executive Officer in December 2007. Mobius and Mr. Palmer are considered related parties to the Company. Global is an entity that had certain trade secrets, know-how, business plans, term sheets, business relationships, and other information relating to the start-up of a business related to the cultivation and production of seed oil from the seed of the Jatropha plant, for the purpose of providing feedstock oil intended for the production of bio-diesel. Under the Global Agreement, the Company issued 63,945,257 shares of its common stock for all of the issued and outstanding membership interests of Global. Of the 63,945,257 shares of common stock were, however, held in escrow by the Company, subject to forfeiture in the event that certain specified performance and market-related milestones were not achieved. Upon the satisfaction, from time to time, of the operational and market capitalization condition milestones, the restricted shares would be released by the Company from escrow and delivered to the buyers in accordance with the terms and conditions of the Global Agreement. In the event that all of the milestone conditions were not achieved, the restricted shares that had not been released from escrow would be cancelled by the Company and thereafter cease to be outstanding.

Prior to the exchange of common stock, Global had no tangible assets or operations, but rather had certain trade secrets, know-how, business plans, term sheets, business relationships, and other information relating to the start-up of a business related to the cultivation and production of seed oil from the seed of the Jatropha plant. Accordingly, Global was not considered a business in accordance with FASB Emerging Issues Task Force Issue 98-3, *Determining Whether a Nonmonetary Transaction Involves Receipt of Productive Assets or of a Business.* With the exchange of the 36,540,146 shares of common stock, the Company acquired the trade secrets, know-how, business plans, term sheets, business relationships, and other information relating to the start-up of this new business. Accordingly, the Company has recorded research and development expense of \$986,584, or \$0.027 per share, for the value of the shares issued. The closing price of the Company's common stock on September 7, 2007 was \$0.027 per share.

Of the restricted shares issued under the Global Agreement, 13,702,556 shares were to be released from escrow if and when i) certain land lease agreements suitable for the planting and cultivation of *Jatropha curcas* were executed and ii) certain operation management agreements with a third-party land and operations management company with respect to the management, planting and cultivation of *Jatropha curcas* were executed. These restricted shares were to be held in escrow subject to the satisfaction of these milestones, at which time such shares would be released from escrow and delivered to the sellers. The Company has accounted for these potentially issuable shares as share-based compensation under SFAS No. 123(R) for shares of common stock that contain a performance or service condition. The Company has determined the value of these shares to be %369,969, or \$0.027 per share, and amortized this compensation over four months, the period of time in which the satisfaction of the operational milestones was expected to be fulfilled that would result in the release of the 13,702,556 shares from escrow. For accounting purposes, shares held in escrow are not considered outstanding, but are deemed to be potential dilutive shares for loss per share, calculations. During the years ended December 31, 2008 and 2007, the Company amortized and recognized \$21,581 and \$348,388 of share-based compensation related to these shares, respectively. With the acquisition of the land for the Jatropha Farm in April 2008, the operational milestones were satisfied under the Global Agreement. Consequently, 13,702,556 shares of common stock being held in escrow have been released to the former owners of Global Clean Energy Holdings, LLC.

The remaining 13,702,555 restricted shares issued under the Global Agreement are to be released from escrow upon satisfaction of certain market capitalization levels (based on the number of outstanding shares at the average closing price of the previous sixty trading days) and average daily trading volume (for the previous sixty trading days). These potentially issuable shares are to be released as follows:

- a. 4,567,518 shares are to be released upon the achievement of \$6 million market capitalization and 75,000 shares of average daily trading volume,
- b. 4,567,518 shares are to be released upon the achievement of \$12 million market capitalization and 100,000 shares of average daily trading volume, and
- c. 4,567,519 shares are to be released upon the achievement of \$20 million market capitalization and 125,000 shares of average daily trading volume.

These restricted shares were placed in escrow subject to the satisfaction of these milestones, at which time such shares are to be released from escrow and delivered to the sellers. On November 30, 2007, the first of these milestones was met and 4,567,518 shares were released from escrow and delivered to the sellers. During May 2008, the second market-related milestones under the Global Agreement were satisfied, which resulted in the release of an additional 4,567,518 shares of common stock being held in escrow. There are 4,567,519 shares of common stock held in escrow at December 31, 2008, which will be released upon the satisfaction of the third market-related milestones. The Company is accounting for these potentially issuable shares as share-based compensation under SFAS No. 123(R), for shares of common stock that contain a market condition. The Company determined the value of these shares to be \$369,969, or \$0.027 per share, and is amortizing this compensation over the periods of time in which the satisfaction of each of the three market capitalization and trading volume milestones is expected to be fulfilled that will result in the release of the 13,702,555 shares from escrow. The Company originally estimated these time periods to be approximately three months for the first tranche of stock and two years for the second and third tranches. For accounting purposes, shares held in escrow are not considered outstanding, but are deemed to be potential dilutive shares for loss per share calculations. During the years ended December 31, 2008 and 2007, the Company amortized and recognized \$165,712 and \$161,860, respectively, of share-based compensation related to these shares.

Palmer Employment Agreement

Effective September 1, 2007, the Company entered into an employment agreement with Richard Palmer pursuant to which the Company hired Mr. Palmer to serve as its President and Chief Operating Officer. Mr. Palmer was also appointed to serve as a director on the Company's Board of Directors to serve until the next election of directors by the Company's shareholders. Upon the resignation of the former Chief Executive Officer on December 21, 2007, Mr. Palmer also became the Company's Chief Executive Officer. The Company hired Mr. Palmer to take advantage of his experience and expertise in the feedstock/bio-diesel industry, and in particular, in the Jatropha bio-diesel and feedstock business. The term of employment commenced September 1, 2007 and ends on September 30, 2010, unless terminated in accordance with the provisions of the agreement.

Mr. Palmer's compensation package includes an annual base salary of \$250,000, subject to annual increases based on changes in the Consumer Price Index, and a bonus payment based on Mr. Palmer's satisfaction of certain performance criteria established by the compensation committee of the Company's Board of Directors. The bonus amount in any fiscal year will not exceed 100% of Mr. Palmer's base salary. Mr. Palmer is eligible to participate in the Company's employee stock option plan and other welfare plans. The Company granted Mr. Palmer an incentive option to purchase up to 12,000,000 shares of its common stock at an exercise price of \$0.03 per share (the trading price on the date the agreement was signed). The options vest upon the Company's achievement of certain market capitalization goals. When the Company's market capitalization reaches \$75 million, the incentive option will vest with respect to 6,000,000 shares. When the Company's market capitalization reaches \$120 million, the incentive option will vest with respect to the remaining 6,000,000 shares. The option expires five years after grant.

If Mr. Palmer's employment is terminated by the Company without "cause" or by Mr. Palmer for "good reason", he will be entitled to severance payments including 100% of his then-current annual base salary, plus 50% of the target bonus for the fiscal year in which his employment is terminated, and the incentive option to purchase 12,000,000 shares of common stock shall vest following termination of Mr. Palmer's employment.

The Company has accounted for the options under Mr. Palmer's employment agreement as share-based compensation under SFAS No. 123(R), for options to purchase common stock that contain a market condition. The Company valued these options at \$264,000 using the Black-Scholes pricing model. The weighted average fair value of the stock options was \$0.022 per share. The weighted-average assumptions used for the calculation of fair value were risk-free rate of 4.21%, volatility of 116%, expected life of five years, and dividend yield of zero. The Company is amortizing this compensation over the period of time in which the satisfaction of each of the two market capitalization milestones is expected to be fulfilled that will result in the vesting of these stock options. The Company currently estimates these time periods to be three years. During the years ended December 31, 2008 and 2007, the Company amortized and recognized \$88,000 and \$29,652, respectively, of share-based compensation related to these options.

Mobius Consulting Agreement

Concurrent with the execution of the Global Agreement, the Company entered into a consulting agreement with Mobius pursuant to which Mobius agreed to provide consulting services to the Company in connection with the Company's new Jatropha bio-diesel feedstock business. The Company engaged Mobius as a consultant to obtain Mobius' experience and expertise in the feedstock/bio-diesel market to assist the Company and Mr. Palmer in developing this new line of operations for the Company. Mobius agreed to provide the following services to the Company: (i) manage and supervise a contemplated research and development program contracted by the Company and conducted by the University of Texas Pan American regarding the location, characterization, and optimal economic propagation of the Jatropha plant; and (ii) assist with the management and supervision of the planning, construction, and start-up of plant nurseries and seed production plantations in Mexico, the Caribbean or Central America.

The term of the agreement was twelve months and the scope of work under the agreement has been completed. Mobius supervised the hiring of certain staff to serve in management and operations roles of the Company, or hired such persons to provide similar services as independent contractors. Mobius' compensation for the services provided under the agreement was a monthly retainer of \$45,000. The Company also reimbursed Mobius for reasonable business expenses incurred in connection with the services provided. The agreement contained customary confidentiality provisions with respect to any confidential information disclosed to Mobius or which Mobius received while providing services under the agreement. Under this agreement, the Company has paid Mobius or accrued \$437,279 during the year ended December 31, 2008, of which \$42,155 was expensed as compensation to Mobius and \$395,124 was capitalized as plantation development costs pursuant to AICPA Statement of Position 85-3, *Accounting by Agricultural Producers and Agricultural Cooperatives*. During the year ended December 31, 2007, the Company paid Mobius or accrued \$191,547, of which \$40,797 was expensed as compensation to Mobius and \$150,750 was capitalized as plantation development costs. The Company owed Mobius \$322,897 and \$50,700 for accrued, but unpaid, compensation and costs as of December 31, 2008 and 2007, respectively.

LODEMO Agreement

On October 15, 2007, the Company entered into a service agreement with Corporativo LODEMO S.A DE CV, a Mexican corporation (the LODEMO Group). The Company had decided to initiate its Jatropha Business in Mexico, and had identified parcels of land in Mexico to plant and cultivate Jatropha. In order to obtain all of the logistical and other services needed to operate a large-scale farming and transportation business in Mexico, the Company entered into the service agreement with the LODEMO Group, a privately held Mexican company with substantial land holdings, significant experience in diesel distribution and sales, liquids transportation, logistics, land development and agriculture.

Under the supervision of the Company's management and Mobius, the LODEMO Group is responsible for the establishment, development, and day-to-day operations of the Jatropha Business in Mexico, including the extraction of the oil from the Jatropha seeds, the delivery of the Jatropha oil to buyers, the purchase or lease of land in Mexico, the establishment and operation of one or more Jatropha nurseries, the clearing, planting and cultivation of the Jatropha fields, the harvesting of the Jatropha seeds, the operation of the Company's oil extraction facilities, and the logistics associated with the foregoing. Although the LODEMO Group is responsible for identifying and acquiring the farmland, ownership of the farmland or any lease thereto will be held directly by the Company or by a Mexican subsidiary of the Company. The LODEMO Group will be responsible for hiring and managing all necessary employees. All direct and budgeted costs of the Jatropha Business in Mexico will be borne by the Company.

The LODEMO Group will provide the foregoing and other necessary services for a fee primarily based on the number of hectares of Jatropha under cultivation. The Company has agreed to pay the LODEMO Group a fixed fee per year of \$60 per hectare of land planted and maintained with minimum payments based on 10,000 hectares of developed land, to follow a planned planting schedule. The Agreement has a 20-year term but may be terminated earlier by the Company under certain circumstances. The LODEMO Group will also potentially receive incentive compensation for controlling costs below the annual budget established by the parties, production incentives for increased yield and a sales commission for biomass sales. Under this agreement, the Company has paid the LODEMO Group or accrued \$1,089,554 and \$158,028 during the years ended December 31, 2008 and 2007, respectively, all of which was capitalized as plantation development costs pursuant to AICPA Statement of Position 85-3, *Accounting by Agricultural Producers and Agricultural Cooperatives*. During the year ended December 31, 2008, the Company issued warrants to acquire 2,076,083 shares of common stock to the LODEMO Group and an affiliated entity in satisfaction of accounts payable in the amount of \$124,565. As of December 31, 2008, the Company had prepaid \$98,159 of plantation development costs to the LODEMO Group \$117,758 for accrued, but unpaid, compensation and costs.

GCE Mexico I, LLC

Effective April 23, 2008, the Company entered into a limited liability company agreement ("LLC Agreement") to form GCE Mexico I, LLC, a Delaware limited liability company (GCE Mexico), with six unaffiliated investors (collectively, the Investors). GCE Mexico was organized primarily to acquire approximately 5,000 acres of farm land (the Jatropha Farm) in the State of Yucatan in Mexico to be used primarily for the (i) cultivation of *Jatropha curcas*, (ii) the marketing and sale of the resulting fruit, seeds, or pre-processed crude Jatropha oil, whether as biodiesel feedstock, biomass or otherwise, and (iii) the sale of carbon value, green fuel value, or renewable energy credit value (and other similar environmental attributes) derived from activities at the Jatropha Farm.

Under the LLC Agreement, the Company owns 50% of the issued and outstanding common membership units of GCE Mexico. The remaining 50% of the common membership units was issued to five of the Investors. The Company and the other owners of the common membership interest were not required to make capital contributions to GCE Mexico.

In addition, two of the Investors agreed to invest approximately \$4.2 million in GCE Mexico through the purchase of preferred membership units and through the funding of the purchase of land in Mexico. An aggregate of 1,000 preferred membership units were issued to these two Investors who each agreed to make capital contributions to GCE Mexico of up to \$2,232,624, in installments and as required, to fund the development and operations of the Jatropha Farm. Shortly after entering into the LLC Agreement, the preferred members made an initial capital contribution of \$957,191 toward the development of the Jatropha Farm. Additional capital contributions of \$1,457,960 have been received by GCE Mexico from these Investors during the remainder of the year ended December 31, 2008. The agreement calls for additional contributions, as requised by management and as required by the operation in 2009 and the following years. Subsequent to December 31, 2008, these Investors have made additional capital contributions of \$1,071,278. These Investors are entitled to earn a preferential 12% per annum cumulative compounded return on the cumulative balance of their preferred membership interest.

These investors also directly funded the purchase of approximately 5,000 acres of land in the State of Yucatan in Mexico by the payment of \$2,051,282. The land was acquired in the name of Asideros and Asideros issued a mortgage in the amount of \$2,051,282 in favor of these two Investors. The mortgage bears interest at the rate of 12% per annum, payable quarterly. The Board has directed that this interest shall continue to accrue until such time as the Board determines that there is sufficient cash flow to pay all accrued interest. The entire mortgage, including any unpaid interest, is due April 23, 2018.

Since the acquisition of the land, approximately 2,500 acres have been improved so far, approximately 750 acres have been planted, and roads and other infrastructure have been developed on the farm. Furthermore, heavy equipment is now in place that will greatly facilitate rapid improvement and planting.

The net income or loss of Asideros is allocated to its shareholders based on their respective equity ownership, or 99% to GCE Mexico and 1% directly to the Company. GCE Mexico has no operations separate from its investment in Asideros. According to the LLC Agreement of GCE Mexico, the net loss of GCE Mexico (composed solely of its share of the operating results of Asideros) is allocated to its members according to their respective investment balances. Accordingly, since the common membership interest did not make a capital contribution, all of the losses have been allocated to the preferred membership interest. The Minority Interest presented in the accompanying consolidated balance sheet includes the carrying value of the preferred membership interests and of the common membership interests owned by the Investors, and excludes any common membership interest in GCE Mexico held by the Company. Accordingly, the Minority Interest is composed of the following elements at December 31, 2008:

Capital contribution from preferred membership interest	\$ 2,415,151
Allocation of net loss of GCE Mexico to the	
preferred membership interest	(315,115)
Accrual of preferential return for the preferred	
membership interest	(138,014)
Investment of common membership interest held by	
other Investors, excluding the Company	
Minority Interest	\$ 1,962,022



NOTE D – PROPERTY AND EQUIPMENT

Property and equipment as of December 31, 2008 and 2007 are as follows:

	2008	2007
Land	\$ 2,051,282	\$ -
Plantation development costs	2,117,061	308,777
Plantation equipment	509,037	-
Office equipment	10,993	 1,127
Total cost	4,688,373	309,904
Less accumulated depreciation	(22,296)	 (563)
Property and equipment, net	\$ 4,666,077	\$ 309,341

The Company has capitalized farming equipment and costs related to the development of land for farm use in accordance with AICPA Statement of Position 85-3, *Accounting by Agricultural Producers and Agricultural Cooperatives* Plantation equipment is depreciated using the straight-line method over estimated useful lives of 5 to 15 years and is currently being capitalized as part of plantation development costs. Plantation development costs are not currently being depreciated. Upon completion of the plantation development, development costs having a limited life and intermediate-life plants that have growth and production cycles of more than one year will be depreciated over the useful lives of the related assets.

Commencing in June 2008, Asideros purchased certain equipment for purposes of rapidly clearing the land, preparing the land for planting, and actually planting the Jatropha trees. The land, plantation development costs, and plantation equipment are located in Mexico.

NOTE E – ACCRUED PAYROLL AND PAYROLL TAXES

Accrued payroll and payroll taxes principally relate to unpaid compensation for officers and directors that are no longer affiliated with the Company. Accrued payroll taxes will become due upon payment of the related accrued compensation. Accrued payroll and payroll taxes are composed of the following:

December 31, 2008		December 31, 2007	
\$	570,949	\$	583,332
	311,200		311,200
	38,510		38,510
	238,149		17,929
\$	1,158,808	\$	950,971
		2008 \$ 570,949 311,200 38,510 238,149	\$ 570,949 \$ 311,200 38,510 238,149



On August 31, 2007, the Company entered into a Release and Settlement Agreement with Judy Robinett, the Company's then-current Chief Executive Officer. Under the agreement, Ms. Robinett agreed to, among other things, assist the Company in the sale of its legacy assets and complete the preparation and filing of the delinquent reports to the Securities and Exchange Commission. Under the agreement, Ms. Robinett agreed to (i) forgive her potential right to receive \$1,851,805 in accrued and unpaid compensation, unaccrued and pro-rata bonuses, and severance pay and (ii) the cancellation of stock options to purchase 14,000,000 shares of common stock at an exercise price of \$0.02 per share. In consideration for her services, the forgiveness of the foregoing cash payments, the cancellation of the stock options, and settlement of other issues, the Company agreed to, among other things, to pay Ms. Robinett \$500,000 upon the receipt of the cash payment under the agreement to sell the SaveCream Assets to Eucodis. Pursuant to this agreement, Ms. Robinett resigned on December 21, 2007.

NOTE F — DEBT

Secured Promissory Note

In order to fund ongoing operations pending closing of the sale of the SaveCream Assets, the Company entered into a loan agreement with, and issued a promissory note in favor of, Mercator Momentum Fund III, L.P. (Mercator) in September 2007. At that time, Mercator, along with two other affiliates, owned all of the issued and outstanding shares of the Company's Series A Convertible Preferred Stock, and is considered a related party to the Company. The loan is secured by a lien on all of the assets of the Company. Under the loan agreement, interest was originally payable on the loan at a rate of 12% per annum, payable monthly.

Pursuant to the loan agreement, Mercator made available to the Company a secured term credit facility in principal amount of \$1,000,000. The promissory note initially was due and payable on December 14, 2007. As of December 13, 2007, the Company owed Mercator \$250,000 under the loan. Mercator agreed to extend the maturity date of the \$250,000 to February 21, 2008. In March, 2008, the loan was paid down to \$200,000 and the maturity date was extended to June 21, 2008. In May 2008, the Company and Mercator entered into an amendment to the loan agreement, whereby, Mercator loaned the Company and additional \$250,000 increasing the outstanding balance to \$450,000. In connection with the amendment, the interest rate was reduced to 8.68% and the due date was extended to August 19, 2008. Additionally, as part of the amendment, the Company issued Mercator a two-year warrant to purchase 581,395 shares of common stock at \$0.129 per share. For the consideration of increasing the note by \$10,000, the maturity date was further extended to January 13, 2009. On December 9, 2008 these notes were assigned to the limited partners of Mercator. Subsequent to December 31, 2008, these notes were further extended from January 13, 2009 to July 31, 2009 for consideration of increasing the total principal balance of the notes by \$15,000 and increasing the interest rate to 10.68%.

In connection with the closing of the original loan, the Company agreed to (i) the cancellation of certain warrants to purchase 27,452,973 shares of common stock at \$0.1967 per share previously issued to the lender and certain of its affiliates and (ii) the issuance of new warrants to purchase 27,452,973 shares of common stock at \$0.01 per share. The new warrants permit the cash-less exercise of the warrants and expire on September 30, 2013. As more fully described in Note G, the warrants that were cancelled were being accounted for as a liability in the accompanying financial statements because the Company was unable to guarantee that there would be enough shares of common stock to settle other "freestanding instruments." The carrying value of the liability related to these warrants on the date of cancellation was \$62,205. The new warrants that were issued in connection with this loan agreement were also characterized as a liability in these financial statements. The fair value of the new warrants was determined to be \$691,815, or \$0.0252 per share, using the Black-Scholes pricing model. The weighted-average assumptions used for the calculation of fair value were risk-free interest rate of 4.10%, volatility of 123%, expected life of six years, and dividend yield of zero. On the date of issuance, the fair value of the new warrants has been recorded as (i) a charge of \$441,815 to "Unrealized Gain (Loss) on Financial Instrument" in the accompanying Consolidated Statement of Operations. The discount to the note was amortized over the original term of the loan agreement from September 7, 2007 to December 14, 2007, and recorded as "interest expense from amortization of discount on secured promissory note" in the amount of \$250,000.

The proceeds of \$250,000 resulting of the amendment of the loan agreement in May 2008 have been allocated between the promissory note and the warrant based on the relative fair value of each instrument. The fair value of the warrant was estimated on the date of issuance using the Black-Scholes option pricing model. The assumptions used for valuing the warrant were risk-free interest rate of 2.4%, volatility of 168%, expected life of 2.0 years, and dividend yield of zero. The allocation resulted in a \$36,369 discount to the promissory note, which has been amortized as additional interest over the period from May 19, 2008 through the original extended due date of August 19, 2008 under the amendment.

Notes Payable

The Company has notes payable to shareholders in the aggregate amount of \$56,000 at December 31, 2008 and 2007. The notes originated between 1997 and 1999, bear interest at 12%, are unsecured, and are currently in default. Accrued interest on the notes totaled \$78,821 and \$72,091 at December 31, 2008 and 2007, respectively.

Convertible Notes Payable

The Company has convertible notes payable to certain individuals in the aggregate amount of \$193,200 at December 31, 2008 and 2007. The notes originated in 1996, bear interest at 12%, are unsecured, and are currently in default. Each \$1,000 note is convertible into 667 shares of the Company's common stock. Accrued interest on the convertible notes totaled \$248,799 and \$225,552 at December 31, 2008 and 2007, respectively.

Long-Term Liability and Gain on Debt Restructuring

On June 10, 2006, the Company entered into an agreement with a former creditor to forgive certain amounts owed. The balance owed before the agreement was \$229,066. According to the agreement, \$3,975 was paid on the date of the agreement, another \$3,975 was paid on August 13, 2006, and \$131,116 was forgiven. The remaining balance of \$90,000 was to be due and payable immediately upon the Company's receipt of \$1 million in cumulative license revenue from the Company's drug MDI-P in any human indication. The remaining liability of \$90,000 was recorded as Long-Term Liability. As further described in Note M, this liability was extinguished as a result of the sale of MDI-P for less than \$1 million. Accordingly, this liability was no longer owed and was written off in 2007. Additionally, as further described in Note L, the Company entered into a settlement agreement with its former chief executive officer during 2007, which resulted in a gain of \$395,137 on the settlement of compensation owing to her. As a consequence of these two transactions, the Company recorded gain on debt restructuring in the amount of \$485,137 in the accompanying financial statements for the year ended December 31, 2007.

NOTE G - STOCKHOLDERS' EQUITY

Common Stock

As more fully described in Note C, the Company issued 63,945,257 shares of its common stock for all of the issued and outstanding membership interests of Global Clean Energy Holdings, LLC. Of the 63,945,257 shares issued under the Global Agreement, 36,540,146 shares were issued and delivered at the closing of the Global Agreement without any restrictions and have been recorded in the accompanying financial statements as issued and outstanding. The remaining 27,405,111 shares of common stock were held in escrow by the Company until the achievement of certain operational and market-related milestones. During the year ended December 31, 2007, 4,567,518 shares were released from escrow upon achieving the first market-related milestones. During the year ended December 31, 2008, an additional 18,270,074 shares were released from escrow upon the achievement of the operational milestones and the second market-related milestones. At December 31, 2008, there are 4,567,519 shares still held in escrow pending achievement of the third market-related milestones. Shares held in escrow are not reported in the accompanying financial statements as issued and outstanding.

On September 14, 2007, the Company entered into a one-year agreement with a consultant for investor relations services. Under the agreement, the Company agreed to pay total compensation of \$105,000 over the one-year term. As additional compensation, the Company issued 4,357,298 shares of common stock to the consultant and granted piggyback registration rights for the stock to be registered in connection with the Company's next registration of securities. The issuance of the common stock was expensed as share-based compensation in the amount of \$117,647, or \$0.027 per share on the date of the agreement.

On November 13, 2008, the Company entered into stock purchase agreements with certain individuals for the issuance of 2,777,778 shares of common stock for \$100,000, or \$0.036 per share.

Series A Convertible Preferred Stock, Warrants and Financial Instrument

During the year ended December 31, 2005, the Company issued an additional 30,000 shares of Series A Convertible Preferred Stock and warrants to purchase 22,877,478 shares of common stock for a total offering price of \$3.0 million. In connection with the offering, the Company issued to the placement agent warrants to purchase 1,220,132 shares. Each share of Preferred Stock entitled the holder to convert the share of Preferred Stock into the number of shares of common stock resulting from dividing \$100 by the conversion price.

The conversion feature of the Series A Convertible Preferred Stock had more of the attributes of an equity instrument than of a liability instrument, and thus was not considered a derivative. However, at the time of issuance, the Company was unable to guarantee that there would be enough shares of stock to settle other "freestanding instruments." Accordingly, all of the warrants attached to the convertible preferred stock were measured at their fair value and recorded as a liability in the financial statements. For these same reasons, all other warrants and options outstanding on March 11, 2005 or issued during the remainder of 2005 and through 2007 (except for stock options issued to employees) were measured at their fair value and recorded as additional liability in the financial statements.

At December 31, 2006, the fair value of the financial instrument was \$294,988 based on a Black-Scholes calculation with the weighted-average assumptions for volatility of 138%, risk-free interest rate of 5.0%, an expected life of one year, and a dividend yield of zero. During the year ended December 31, 2007, the Company remeasured the fair value of the outstanding warrants. At December 31, 2007, the fair value was determined to be \$2,166,514 based on a Black-Scholes pricing calculation with the weighted-average assumptions for volatility of 136%, a risk-free interest rate of 3.7%, an expected life of 7.3 years, and a dividend yield of zero. For the year ended December 31, 2007 the Company recorded an unrealized loss on financial instrument of \$147,636. For the period from December 31, 2007 through January 29, 2008, the fair value of this liability decreased by \$5,469 resulting in a balance of \$2,161,045. On January 29, 2008, the shareholders of the Company approved an increase in the number of authorized shares of common stock from 250 million to 500 million. Consequently, as the result of this amendment to the Company's Articles of Incorporation, the Company is now able to settle all 'freestanding instruments''. Accordingly, the Company reclassified the liability, characterized in the accompanying financial statements as "Financial Instrument", in the amount of \$2,161,045, to permanent equity in January 2008.



In September 2007, the preferred stockholders converted 5,492 shares of Series A Preferred Stock into 10,983,521 shares of common stock at a conversion price of \$0.05 per share. This preferred stock also did not have any assigned value.

Mercator Momentum Fund, LP; Monarch Pointe Fund, Ltd.; and Mercator Momentum Fund III, LP, each a private investment entity (collectively, the MAG Funds) were the preferred stockholders who purchased all of the shares of the Company's Series A Preferred Convertible Stock in 2004 and in 2005. In connection with the 2005 investment, the Company had agreed to eliminate the conversion price floor of the Series A Stock. The Company failed to file an amendment to the Series A Stock Certificate of Designations of Preferences and Rights for the Series A Stock that would have eliminated the conversion price floor. Accordingly, in connection with an intended conversion of some of their Series A Stock in September 2007, the MAG Funds were required to convert Series A Stock at a conversion price higher than the price that would have applied if the Amendment had been filed as agreed.

On October 22, 2007, the Company executed and entered into a Release and Settlement Agreement (the Release Agreement), with the MAG Funds to settle all losses and damages that MAG may have suffered, and may hereafter suffer, as result of the Company's failure to file the amendment to the Series A Stock Certificate of Designations of Preferences and Rights for the Series A Stock. Pursuant to the Release Agreement, the Company issued to the MAG Funds a ten-year warrant to acquire up to 17,000,000 shares of the Company's common stock at an exercise price of \$0.01 per share, expiring October 17, 2017. The initial warrant price is subject to adjustments in connection with (i) the Company's issuance of dividends in shares of Common Stock, or shares of Common Stock or other securities convertible into shares of Common Stock without consideration, (ii) any cash paid or payable to the holders of Common Stock other than as a regular cash dividend, and (ii) future stock splits, reverse stock splits, mergers or reorganizations, and similar changes affecting common stockholders. The issuance of the warrant has been accounted for as share-based compensation in the amount of \$1,181,890 based on a Black-Scholes pricing calculation with the assumptions for volatility of 141.5%, a risk-free interest rate of 4.57%, an expected life of 10 years, and a dividend yield of zero. The fair value of the warrant has been included in the liability for the financial instrument.

The warrant issued to the MAG Funds contain beneficial ownership limitations, which preclude the MAG Funds from exercising its warrant if, as a result of such conversion or exercise, the MAG Funds would own beneficially more than 9.99% of the Company's outstanding common stock then outstanding. Pursuant to the Release Agreement, the MAG Funds released the Company from any and all claims, past, present or future, relating to the losses or the Company's failure to file the amendment. In addition, MAG has agreed not to pursue litigation against the Company in connection with the losses or the Company's failure to file the amendment.

Effective April 18, 2008, the Company entered into an exchange agreement (the Exchange Agreement) with Mercator Momentum Fund, L.P., Mercator Momentum Fund III, L.P., and Monarch Pointe Fund, Ltd. (collectively, the MAG Funds), comprising all of the holders of the Company's Series A Convertible Preferred Stock (the Series A Stock). Pursuant to the Exchange Agreement, the MAG Funds agreed to exchange 28,928 shares of the Series A Stock, constituting all of the issued and outstanding shares of the Series A Stock, for an aggregate of 28,927,000 shares of the Company's common stock. The exchange ratio was determined by dividing the \$100 purchase price of the preferred shares by \$0.10 per share of common stock.

Prior to the Exchange Agreement, the Series A Stock had been convertible at a price equal to 75% of the "Market Price", as defined in the Certificate of Designations of Preferences and Rights of the Series A Stock. The conversion price could not exceed \$0.1967 and had a conversion price floor of \$0.05. On April 18, 2008, the closing price of the Company's common stock was \$0.10 and the "Market Price" would have been \$0.045 per share. In connection with the Exchange Agreement, the Company agreed to waive the limitation that the MAG Funds could not own more that 9.99% of the Company's outstanding common stock as a concession for the MAG Funds agreeing to a conversion price that was more favorable to the Company.

Series B Preferred Stock

In order to obtain additional working capital, on November 6, 2007, the Company entered into a Securities Purchase Agreement with two accredited investors, pursuant to which the Company sold a total of 13,000 shares of our newly authorized Series B Convertible Preferred Stock ("Series B Shares") for an aggregate purchase price of \$1,300,000, less offering costs of \$9,265. Each share of the Series B Shares has a stated value of \$100. The Company collected \$1,225,000 of the proceeds from the sales of the Series B Preferred Stock in 2007. The remaining proceeds of \$75,000 were collected in February 2008, and are reflected as a subscription receivable in the accompanying Balance Sheet at December 31, 2007.

The Series B Shares may, at the option of each holder, be converted at any time or from time to time into shares of our common stock at the conversion price then in effect. The number of shares into which one Series B Share shall be convertible is determined by dividing \$100 per share by the conversion price then in effect. The initial conversion price per share for the Series B Shares is \$0.11, which is subject to adjustment for certain events, including stock splits, stock dividends, combinations, or other recapitalizations affecting the Series B Shares.

Each holder of Series B Shares is entitled to the number of votes equal to the number of shares of our common stock into which the Series B Shares could be converted on the record date for such vote, and shall have voting rights and powers equal to the voting rights and powers of the holders of the Company's common stock. In the event of our dissolution or winding up, each share of the Series B Shares is entitled to be paid an amount equal to \$100 (plus any declared and unpaid dividends) out of the assets of the Company then available for distribution to shareholders.

No dividends are required to be paid to holders of the Series B shares. However, the Company may not declare, pay or set aside any dividends on shares of any class or series of our capital stock (other than dividends on shares of our common stock payable in shares of common stock) unless the holders of the Series B shares shall first receive, or simultaneously receive, an equal dividend on each outstanding share of Series B shares.

NOTE H — INCOME TAXES

Income taxes are provided for temporary differences between financial and tax bases of assets and liabilities. The following is a reconciliation of the amount of benefit that would result from applying the federal statutory rate to pretax loss with the benefit from income taxes for the years ended December 31, 2008 and 2007:

	 2008		2007
Federal income tax benefit at statutory rate of 34%	\$ 581,000	\$	1,501,000
State income tax, net of federal benefit	102,000		265,000
Unrealized gain (loss) on financial instrument	2,000		(59,000)
Foreign currency translation adjustment	43,000		(119,000)
Amortization of discount on notes payable	(15,000)		(100,000)
Share-based compensation, net	(147,000)		(764,000)
Expiration of operating loss and research credit carryforwards	(511,000)		(164,000)
Adjustment of operating loss carryforwards	-		1,627,000
Research and development	-		(395,000)
Other differences	(1,000)		(4,000)
Change in valuation allowance	(54,000)	_	(1,788,000)
	\$ -	\$	

The components of deferred tax assets and liabilities are as follows at December 31, 2008 and 2007, using a combined deferred income tax rate of 40%:

	2008	2007
Net operating loss carryforward	\$ 9,483,000	\$ 9,534,000
Research and development credits	-	80,000
Share-based compensation	716,000	714,000
Accrued compensation	511,000	408,000
Deferred revenue	-	(80,000)
Valuation allowance	(10,710,000)	(10,656,000)
Net deferred tax asset	\$ -	\$ -

Inasmuch as it is not possible to determine when or if the net operating losses will be utilized, a valuation allowance has been established to offset the benefit of the utilization of the net operating losses.

The Company has available net operating losses of approximately \$23,700,000 which can be utilized to offset future earnings of the Company. The utilization of the net operating losses are dependent upon the tax laws in effect at the time such losses can be utilized. The loss carryforwards expire between the years 2009 and 2028. Should the Company experience a significant change of ownership, the utilization of net operating losses could be reduced.

NOTE I – CONSULTING AGREEMENTS

In February 2007, the Company engaged the Emmes Group, a consulting firm, to assist it in resolving its financial issues, to obtain advice regarding any strategic alternatives that may be available to it, and to prevent the Company from losing all of its assets in bankruptcy. The Executive Vice President and Managing Director of the Emmes Group was appointed to be a director of the Company in August 2007. The Company explored a number of transactions that would (i) prevent the Company's shareholders from losing their entire investment in the Company and (ii) enable the Company to repay some of its currently outstanding debts and liabilities. The consulting agreement had a term of one year. As compensation for its services, the consultant received \$15,000 per month plus a warrant to purchase 5,000,000 shares of the Company's common stock. The warrant has an exercise price of \$0.03 per share, contains a cash-less exercise provision, and expires ten years from date of issue. The Company valued this warrant at \$146,000 using the Black-Scholes pricing model. The weighted average fair value of the stock options was \$0.0292 per share. The weighted-average assumptions used for the calculation of fair value were risk-free interest rate of 4.84%, volatility of 134%, expected life of ten years, and dividend yield of zero. The fair value of the warrant was included in the liability for the financial instrument.

In February 2007, the Company entered into another consulting agreement with an individual to assist it in the preparation of financial statements and reporting to the SEC. The consulting agreement had a term of one year. As compensation for its services, the consultant was to receive \$10,000 per month plus a warrant to purchase 5,000,000 shares of the Company's common stock. The warrant has an exercise price of \$0.03 per share, contains a cash-less exercise provision, and expires ten years from date of issue. The Company valued this warrant at \$146,000 using the Black-Scholes pricing model. The weighted average fair value of the stock options was \$0.0292 per share. The weighted-average assumptions used for the calculation of fair value were risk-free interest rate of 4.84%, volatility of 134%, expected life of ten years, and dividend yield of zero. The fair value of the warrant was expensed as share-based compensation on the date of issue. The fair value of the warrant was included in the liability for the financial instrument. This consulting agreement was terminated in May 2007. Since the consulting agreement was terminated prior to its expiration date, the Company's obligations under the consulting agreement, if any, for the period after the termination date are unclear. No demand for any additional compensation has been made against the Company under the consulting agreement.

NOTE J – EMPLOYMENT AGREEMENT

On March 20, 2008, the Company entered into an employment agreement with Bruce K. Nelson pursuant to which the Company hired Mr. Nelson to serve as its Executive Vice-President and Chief Financial Officer effective April 1, 2008. The initial term of employment commenced March 20, 2008 and continues through March 20, 2010. Thereafter, the term of employment shall automatically renew for successive one-year periods unless otherwise terminated in accordance with the employment agreement.

Mr. Nelson's compensation package includes a base salary of \$175,000, subject to annual increases based on the Consumer Price Index for the immediately preceding 12month period, and a bonus payment based on Mr. Nelson's satisfaction of certain performance criteria established by the compensation committee of the Company's Board of Directors. The bonus amount in any fiscal year will not exceed 100% of Mr. Nelson's base salary. Mr. Nelson is eligible to participate in the Company's employee stock option plan and other benefit plans.

The Company granted Mr. Nelson an option (the Initial Option) to acquire up to 2,000,000 shares of the Company's common stock at an exercise price of \$0.05. The Initial Option vests in tranches of 500,000 shares after 90 days, nine months, fifteen months, and two years of the employment term. The Initial Option expires after 10 years. The Company also granted Mr. Nelson an option (the Performance Option) to acquire up to 2,500,000 shares of the Company's common stock at an exercise price of \$0.05, subject to the Company's achievement of certain market capitalization goals. The Performance Option expires after five years.

The Company was permitted to terminate Mr. Nelson's employment on the first anniversary of the employment term, provided that the Company pay Mr. Nelson three (3) months salary if such termination was without "cause." If Mr. Nelson's employment is terminated by the Company without "cause" or by Mr. Nelson for "good reason" after the first anniversary of the employment term, Mr. Nelson will be entitled to receive severance payments including (i) an amount equal to his unpaid salary through the end of the second year of the employment agreement, and (ii) 100% of Initial Option shall vest, to the extent not already vested.

The Company has accounted for the options under Mr. Nelson's employment agreement as share-based compensation under SFAS No. 123(R). The Company valued these options at \$189,500 using the Black-Scholes pricing model. The weighted average fair value of the stock options was \$0.042 per share. The weighted-average assumptions used for the calculation of fair value were risk-free rate of 2.38%, volatility of 127%, expected life of 5.2 years, and dividend yield of zero. The Company is amortizing this compensation over the vesting period for the Initial Option and over the period of time in which the satisfaction of market capitalization milestones for the Performance Option are expected to be fulfilled that will result in the vesting of these stock options. The Company currently estimates these time periods to be three years for the Performance Options. During the year ended December 31, 2008, the Company amortized and recognized \$91,346 of share-based compensation related to these options.

NOTE K – STOCK OPTIONS AND WARRANTS

Stock Options and Compensation-Based Warrants

The Company has two incentive stock option plans wherein 24,000,000 shares of the Company's common stock are reserved for issuance thereunder. As of December 31, 2008, 300,000 shares remain available under these plans. As more fully described in Notes C, G, I, and J, the Company has issued stock options and compensation-based warrants during the years ended December 31, 2008 and 2007 to acquire 4,500,000 and 39,000,000 million shares, respectively, of the Company's common stock. Additionally, during the year ended December 31, 2008, the Company issued warrants to acquire 2,076,083 shares of common stock to Lodemo and an affiliated entity in satisfaction of accounts payable in the amount of \$124,565. The Company also granted options to acquire 700,000 shares of common stock to independent contractors during the year ended December 31, 2007, as more fully described in Note L, the Company canceled an option to acquire 14,000,000 shares of common stock pursuant to a settlement agreement with the Company's former chief executive officer. No income tax benefit has been recognized for share-based compensation arrangements and no compensation cost has been capitalized in the balance sheet.

A summary of the status of options and compensation-based warrants at December 31, 2008 and 2007, and changes during the years then ended is presented in the following table:

	Shares Under Option	Weighted Average Exercise Price	Weighted Average Remaining Contractual Life	Aggregate Intrinsic Value
Outstanding at December 31, 2006	19,883,000	\$ 0.05		
Granted	39,000,000	0.02		
Expired	-	-		
Cancelled	(14,000,000)	0.02		
Outstanding at December 31, 2007	44,883,000	0.03		
Granted	7,276,083	\$ 0.04		
Expired		-		
Outstanding at December 31, 2008	52,159,083	\$ 0.03	6.4 years	\$ 316,141
Exercisable at December 31, 2008	36,134,083	\$ 0.03	7.4 years	\$ 316,141

At December 31, 2008, 80,000 of the options outstanding have no stated contractual life. Except for warrants issued in satisfaction of accounts payable, the fair value of each stock option grant and compensation-based warrant is estimated on the date of grant or issuance using the Black-Scholes option pricing model. In the case of the warrants issued in satisfaction of accounts payable, the warrants were valued at the amount of the accounts payable satisfied. The weighted-average fair value of stock options and compensation-based warrants issued during the year ended December 31, 2008 was \$0.039. The weighted-average assumptions used for options granted and compensation-based warrants issued during the year ended December 31, 2008 were risk-free interest rate of 2.2%, volatility of 132%, expected life of 4.9 years, and dividend yield of zero. The weighted-average fair value of stock options and compensation-based warrants issued during the year ended December 31, 2007 were risk-free interest rate of 4.5%, volatility of 132%, expected life of 4.9 years, and dividend yield of zero. The weighted-average fair value of stock options granted and compensation-based warrants issued during the year ended December 31, 2007 were risk-free interest rate of 4.5%, volatility of 132%, expected life of 8.5 years, and dividend yield of zero. The assumptions employed in the Black-Scholes option pricing model include the following. The expected life of stock options represents the period of time that the stock options granted are expected to be outstanding prior to exercise. The expected volatility is based on the historical price volatility of the Company's common stock. The risk-free interest rate represents the U.S. Treasury constant maturities rate for the expected life of the related stock options. The dividend yield represents anticipated cash dividends to be paid over the expected life of the stock options.

Share-based compensation from all sources recorded during the years ended December 31, 2008 and 2007 was \$371,439 and \$3,118,021, respectively. Share-based compensation has been included in the accompanying Consolidated Statements of Operations as follows:

Period Reported	Adr	eneral and ninistrative Expense	Research and Development Expense	Loss from Discontinued Operations	 Total
Year ended December 31, 2008	\$	371,439	\$ -	\$ -	\$ 371,439
Year ended December 31, 2007		2,014,637	986,584	116,800	\$ 3,118,021
	F-28				

As of December 31, 2008, there is approximately \$295,000 of unrecognized compensation cost related to stock-based payments that will be recognized over a weighted average period of approximately 1.6 years.

Stock Warrants

A summary of the status of the warrants granted at December 31, 2008 and 2007, and changes during the years then ended is presented in the following table:

*** * * * *

	Shares Under Warrant	Weighted Average Exercise Price
Outstanding at December 31, 2006	38,973,861	\$ 0.19
Issued	29,161,157	0.01
Cancelled	(29,161,157)	0.20
Expired	(7,940,482)	0.15
Outstanding at December 31, 2007	31,033,379	0.02
Issued	581,395	0.13
Expired	(1,872,222)	0.18
Outstanding at December 31, 2008	29,742,552	\$ 0.01

NOTE L – RELEASE AND SETTLEMENT AGREEMENT WITH CHIEF EXECUTIVE OFFICER

On August 31, 2007, the Company entered into a Release and Settlement Agreement with Judy Robinett, the Company's then-current Chief Executive Officer, pursuant to which Ms. Robinett agreed to continue to act as the Company's transitional Chief Executive Officer. Under the agreement, Ms. Robinett agreed to, among other things, assist the Company in the sale of its legacy assets, complete the preparation and filing of the delinquent reports to the Securities and Exchange Commission (the SEC) that related to the periods prior to the appointment of Mr. Palmer, and provide certain shareholder and creditor related services. Upon the completion of the foregoing matters, in particular the filing of the delinquent reports to the SEC, Ms. Robinett was to resign, and Mr. Palmer was to thereafter assume the office of Chief Executive Officer. Under the agreement, Ms. Robinett agreed to (i) forgive her potential right to receive \$1,851,805 in accrued and unpaid compensation, un-accrued and pro-rata bonuses, and severance pay and (ii) the cancellation of stock options to purchase 14,000,000 shares of common stock at an exercise price of \$0.02 per share. In consideration for her services, the forgiveness of the foregoing cash payment, the cancellation of the foregoing stock options, and settlement of other issues, the Company agreed to (a) pay Ms. Robinett \$500,000 upon the receipt of the Eucodis cash payment under the agreement to sell the SaveCream Assets, (b) pay Ms. Robinett a commission of fifteen percent of the gross proceeds received by the Company from the sale of the MDI-P asset, (c) pay Ms. Robinett \$20,833 in monthly salary for serving as transitional Chief Executive Officer of the Company during the period from April 1, 2007 until the effective date of her resignation, and (d) permit Ms. Robinett to retain some of her previously granted incentive stock options in such an amount allowing her to purchase up to two million shares of common stock, which options shall continue to have the same terms and conditions as c

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A DEVELOPMENT STAGE COMPANY) NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

NOTE M – DISCONTINUED OPERATIONS

Prior to 2007, the Company was a developmental-stage bio-pharmaceutical company engaged in the research, validation, development and ultimate commercialization of two drugs known as SaveCream and MDI-P. SaveCream is a drug candidate that the Company was developing to reduce breast cancer tumors. MDI-P was a drug candidate being developed as an anti-infective treatment for bacterial infections, viral infections and fungal infections. During the three months ended March 31, 2007, the Board of Directors determined that it could no longer fund the development of these drug candidates and could not obtain additional funding for these drug candidates. The Board evaluated the value of its developmental stage drug candidates and in March 2007, the Board determined that the best course of action was to discontinue further development of these drug candidates and sell these technologies.

Plan to Sell SaveCream Assets

On March 8, 2007, the Company entered into a binding letter of intent with Eucodis Pharmaceuticals Forschungs und Entwicklungs GmbH, an Austrian company (Eucodis), regarding their intent to proceed with the evaluation, negotiation, and execution of a sale and purchase agreement related to certain assets of the Company. On July 6, 2007, the Company entered into a sale and purchase agreement (the Asset Sale Agreement) with Eucodis, pursuant to which Eucodis agreed to acquire certain assets of the Company in consideration for a cash payment and the assumption by Eucodis of certain indebtedness of the Company. The assets to be acquired by Eucodis pursuant to the Asset Sale Agreement included all of the Company's right, title and interest in all patents, patent applications, United States and foreign regulatory files and data, pre-clinical study data and anecdotal clinical trial data concerning SaveCream. In addition, at the closing of the sale, the Company was to assign to Eucodis all of its right, title and interest in a co-development agreement with Haucodis, dated as of July 29, 2006, related to the co-development and licensing of SaveCream (including the intellectual property rights acquired in connection with that development) and their rights under certain other contracts relating to SaveCream. The sale to Eucodis was scheduled to close at the end of January 2008 after the Company's shareholders approved the sale. On January 29, 2008, the shareholders of the Company approved the transaction. Shortly before the scheduled closing, Eucodis informed the Company that it was unable to complete the transaction as agreed because it had insufficient funds and needed to obtain additional financing.

The Company thereafter commenced discussions with Eucodis regarding the possibility of obtaining financing and possibly deferring the closing of the sale. However, as of February 27, 2008, Eucodis still had not obtained sufficient financing to complete its purchase of the SaveCream technology. Accordingly, on February 27, 2008, the Company delivered to Eucodis a letter formally notifying Eucodis that the Asset Agreement had been terminated. On February 29, 2008, Eucodis informed the Company that (i) it was completing an agreement for financing, which financing would provide Eucodis with sufficient funds to purchase the SaveCream assets for the purchase price, and substantially on the terms set forth in the Asset Sale Agreement, and (ii) that it still desired to complete the transaction contemplated by the Asset Sale Agreement. On February 29, 2008, the Company prepared a letter agreement again agreeing to sell the SaveCream assets to Eucodis on substantially the terms set forth in the Asset Sale Agreement (as amended). Under the letter agreement, the sale to Eucodis was scheduled to occur at such time as Eucodis completed its financing, but in no event later than April 30, 2008. As of April 30, 2008, Eucodis had not completed its financing, therefore, the Asset Sale Agreement, as amended by the Letter Agreement, terminated on its own terms. The Company continued discussions with Eucodis and explored other potential purchasers of SaveCream. All discussions and agreements with Eucodis were terminated in July 2008 due to their inability to obtain their own pending financing. As a result of the failed funding of Eucodis, they were forced to cease their operations. However, the principal of Eucodis has agreed to continue to work with the Company in connection with the sale of the Company's legacy assets.

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A DEVELOPMENT STAGE COMPANY) NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The Company has engaged investment banking firms to expedite the sale of the SaveCream asset. The Company continues to seek interested parties that may purchase the asset. However, the recent contraction of the capital markets has negatively impacted the abilities for several potential purchasers to consummate a purchase. Although, management is continuing to taking steps to market and sell the SaveCream assets to potential buyers, no assurance can be given that this sale will actually be completed in the near future, or ever. Due to the inability of the engaged investment bankers to facilitate a sales transaction of the asset, the Company has terminated the engagement of the investment banking firms.

Agreement to Sell MDI-P

The Company also entertained various offers to purchase the Company's rights to the assets related to the MDI-P compound. On August 9, 2007, the Company sold the MDI-P related assets for \$310,000 in cash realizing a gain of \$258,809. The sale included the patents, name, and other intellectual property, research results and test data, production units and equipment, and other assets related to this technology. No liabilities were assumed by the purchaser in this transaction. A liability in the amount of \$90,000 was extinguished due to the sale. This extinguished liability was only payable when the Company received \$1 million in cumulative license revenue from the MDI-P compound in any human indication. Due to the sale of MDI-P for less than \$1 million, this liability was no longer owed and was written off.

Accounting for Discontinued Operations

Pursuant to accounting rules for discontinued operations, the Company has classified all revenue and expense related to the operations, assets, and liabilities of its biopharmaceutical business as discontinued operations. For all periods prior to March 2007, the Company has reclassified all revenue and operating expenses to discontinued operations, except for estimated general corporate overhead, because all of its operations related to the discontinued technologies. For the year ended December 31, 2007, revenues of \$200,000 are included in the Loss from Discontinued Operations and the Company has recorded a gain from the sale of MDI-P of \$258,809. For the year ended December 31, 2008, the Income from Discontinued Operations consists of the foreign currency transaction gains in the amount of \$107,369 related to current liabilities associated with the discontinued operations that are denominated in euros, less \$40,259 of expenses related to the SaveCream asset. The assets that were under contract to be sold to Eucodis have no carrying value in the accompanying balance sheet, while the liabilities that were to be assumed in the planned sale were formerly segregated in the balance sheets and were previously characterized as Current Liabilities Associated with Assets Held for Sale. As a consequence of the termination of the Asset Sale Agreement in 2008, these current liabilities have been reclassified into the captions Research and Development Obligation and Accounts Payable, as appropriate. The Company has not recorded any gain or loss through December 31, 2008 associated with the planned sale of the SaveCream assets.

Transactions Related to the SaveCream Asset Purchase

On March 16, 2005, the Company completed the purchase of the intellectual property assets (the "Assets") of Savetherapeutics AG, a German corporation in liquidation in Hamburg, Germany ("SaveT"). The Assets consisted primarily of patents, patent applications, pre-clinical study data and clinical trial data concerning SaveCream, a developmental-stage topical aromatase inhibitor treatment for breast cancer. The purchase price of the Assets was $\epsilon_{2,350,000}$, payable as follows: $\epsilon_{500,000}$ at closing, $\epsilon_{500,000}$ upon conclusion of certain pending transfers of patent and patent application rights from the inventors to the Company, and the remaining $\epsilon_{1,350,000}$ upon successful commercialization of the Assets.

F-31

GLOBAL CLEAN ENERGY HOLDINGS, INC. AND SUBSIDIARIES FORMERLY KNOWN AS MEDICAL DISCOVERIES, INC. (A DEVELOPMENT STAGE COMPANY) NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The pending transfers of patent and patent application rights have not occurred. The Company has deemed the transfers are reasonably likely to occur due to existing contractual commitments of the inventors and the reasonably likely success of the Company's action in German court proceeding to affect these transfers. Accordingly, the Company has recorded the second \notin 500,000 payment as Research and Development Obligation in these financial statements. In July 2006 the Company entered into a co-development and license agreement with Eucodis, which provided for up-front licensing fees and milestone payments in excess of the \notin 1,350,000 threshold for successful commercialization of the Assets. Accordingly, in the year ended December 31, 2006 the Company recorded the final \notin 1,350,000 purchase price payment as Research and Development Obligation of \notin 1,850,000 is included in current liabilities in the amount of \$2,607,945 and \$2,701,555, based on exchange rates in effect at December 31, 2008 and 2007, respectively.

On March 27, 2009, the Company was informed by German counsel that pending action in German courts appears to be settled in favor of the Company.

NOTE N – SUBSEQUENT EVENTS

Acquisition of Jatropha Farm in Belize

On October 29, 2008, the Company entered into a Stock Purchase Agreement with the four shareholders of Technology Alternatives Limited (TAL), a company formed under the Laws of Belize. TAL owns and operates a 400 acre farm in subtropical Belize, Central America, that currently is producing Jatropha. TAL has also been performing plant science research and has been providing technical advisory services for propagation of Jatropha for a number of years.

The shareholders of TAL are unaffiliated persons residing in the United Kingdom. Pursuant to the Stock Purchase Agreement, the Company will acquire 100% of the issued and outstanding shares of TAL for common stock in the Company, thereby making TAL a wholly-owned subsidiary of the Company. It is anticipated that the Company will issue 8,952,756 common shares in exchange for all of the outstanding shares of TAL. In addition to receiving the Company's shares, the sellers will be repaid the promissory notes previously issued to them by TAL. However, as of March 27, 2009 all conditions precedent required for the exchange of consideration had not been satisfied, and shares have not been issued. Consequently, this transaction is not reflected in the Company's financial statements dated as of December 31, 2008.

Furthermore, the seller had an obligation to maintain the asset in accordance with the Stock Purchase Agreement and failed to do so. Therefore, the sellers have agreed to decrease the acquisition price and to decrease the principal amounts of the promissory notes.

The selling shareholders had previously made loans to TAL to fund the operations of TAL. As of October 29, 2008 the transaction contemplated by the Stock Purchase Agreement, the remaining outstanding balance of these loans, in the aggregate, was determined to be \$453,611. To reflect the current value of TAL, these notes will be reduced to \$303,611 at closing. At the closing, the promissory notes evidencing these loans will be replaced by new promissory notes issued by TAL to the selling shareholders. The new notes have the following terms: (i) Interest free for 90 days; (ii) Interest accrues at an annual rate of 8% per annum commencing on the 91st day after the issuance of the notes; (iii) Interest accrues until maturity; (iv) The entire remaining unpaid balance of the notes is due and payable on August 31, 2009; (v) TAL and/or the Company may prepay the notes at any time without penalty, and the Company is required to prepay the notes if and when it receives future funding in an amount that, in the Company's reasonable discretion, is sufficient to permit the prepayment of the notes without adversely affecting the Company's operations or financial condition. The new notes are secured by the deed of legal mortgage on the 400 acre farm owned by TAL. Accordingly, in the event that TAL defaults under the notes, the selling shareholders will have the right to foreclose on the 400 acre Jatropha farm.

The acquisition will be accounted for under the purchase method of accounting and the results of operations of TAL will be consolidated with the results of operations of the Company from the date of acquisition.

LIMITED LIABILITY COMPANY AGREEMENT

OF

GCE MEXICO I, LLC

A DELAWARE LIMITED LIABILITY COMPANY

THE SECURITIES REPRESENTED BY THIS AGREEMENT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 NOR REGISTERED OR QUALIFIED UNDER ANY STATE SECURITIES LAWS. SUCH SECURITIES MAY NOT BE OFFERED FOR SALE, SOLD, DELIVERED AFTER SALE, TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS QUALIFIED AND REGISTERED UNDER APPLICABLE STATE AND FEDERAL SECURITIES LAWS OR UNLESS, IN THE OPINION OF COUNSEL SATISFACTORY TO THE COMPANY, SUCH QUALIFICATION AND REGISTRATION IS NOT REQUIRED. ANY TRANSFER OF THE SECURITIES REPRESENTED BY THIS AGREEMENT IS FURTHER SUBJECT TO OTHER RESTRICTIONS, THE TERMS AND CONDITIONS WHICH ARE SET FORTH IN THIS AGREEMENT.

LIMITED LIABILITY COMPANY AGREEMENT

OF

GCE MEXICO I, LLC

A DELAWARE LIMITED LIABILITY COMPANY

This Limited Liability Company Agreement is made as of April 23, 2008, by and among Stewart Resnick and Lynda Resnick, as trustees of the Stewart and Lynda Resnick Revocable Trust dated December 27, 1998, as amended, Selim Zilkha, as trustee of the Selim K. Zilkha Trust, Michael Zilkha, as trustee of the DMZ 2000 Trust, Michael Zilkha, as trustee of the LLZ 2000 Trust, Nadia Z. Wellisz, as trustee of the JW 2000 Trust, Nadia Z. Wellisz, as trustee of the DW 2000 Trust and Global Clean Energy Holdings, Inc., a Utah corporation, with reference to the following facts:

A. The parties desire to form GCE Mexico I, LLC as a limited liability company under the laws of the State of Delaware and, to that end, have filed a Certificate of Formation for the Company with the Delaware Secretary of State.

B. The parties now desire to adopt a limited liability company agreement to govern their respective rights and obligations as members and as the manager of the Company.

NOW, THEREFORE, in consideration of the mutual covenants contained herein and for other good and valuable consideration, the receipt of which is acknowledged, the parties agree that the following shall be the Limited Liability Company Agreement of the Company.

ARTICLE I

DEFINITIONS

When used in this Agreement, the following terms have the following meanings:

1.1 "Act" means the Limited Liability Company Act of the State of Delaware.

1.2 "Adjusted Capital Account" of a Member means the Capital Account of that Member increased by the Member's share of Company Minimum Gain and Member Minimum Gain.

1.3 "Adjusted Capital Contribution" of a Member means the excess of (a) that Member's Capital Contribution to the Company, over (b) the Distributions to the Member under Section 6.10(b) and the Distributions under Section 10.5(a) that shall constitute return of capital Distributions. Distributions to a Member under Section 10.5(a) first shall constitute return of capital Distributions to the extent that these Distributions reduce the Member's Adjusted Capital Contribution to an amount not less than zero, and thereafter Distributions under Section 10.5(a) shall constitute Distributions of Unpaid Preferred Return to the extent that these Distributions reduce the Member's Unpaid Preferred Return to an amount not less than zero.

1.4 "<u>Affiliate</u>" of a Member or Manager means (a) a Person directly or indirectly (through one or more intermediaries) controlling, controlled by or under common control with that Member or Manager; (b) an officer, director, trustee, partner, member or immediate family member of that Member or Manager; or (c) a member of the immediate family of an officer, director, trustee, partner or member of that Member or Manager; provided, however, that (i) neither the Company nor any of its Subsidiaries will be deemed an Affiliate of a Member or Manager and (ii) neither a Member nor a Manager nor any of their respective Affiliates will be deemed an Affiliate of the Company or any of the Company or any of the Company's Subsidiaries. For these purposes "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

1.5 "AGC" means ASIDEROS GLOBALES CORPORATIVO, a wholly-owned Subsidiary of the Company formed under the laws of Mexico.

1.6 "Agreement". means this Limited Liability Company Agreement of GCE Mexico I, LLC, as originally executed and as amended from time to time.

1.7 "Applicable Purchase Price" has the meaning specified in Section 5.3(d).

1.8 "Bankruptcy" of a Person means the institution of any proceedings under any federal or state law for the relief of debtors, including the filing by or against that Person of a voluntary or involuntary case under the federal bankruptcy law, which proceedings, if involuntary, are not dismissed within sixty (60) days after their filing; an assignment of the property of that Person for the benefit of creditors; the appointment of a receiver, trustee or conservator of any substantial portion of the assets of that Person, which appointment, if obtained ex parte, is not dismissed within sixty (60) days thereafter; the seizure by a sheriff, receiver, trustee or conservator of any substantial portion of the assets of that Person; the failure by that Person generally to pay its debts as they become due within the meaning of Section 303(h)(1) of the United States Bankruptcy Code, as determined by the Bankruptcy Court; or that Person's admission in writing of its inability to pay its debts as they become due.

1.9 "Bona Fide Offer" means an offer in writing to a Member offering (subject to no financing contingencies) to purchase all or any part of that Member's Membership Interest or any interest therein and setting forth all of the material terms and conditions of the proposed purchase from an offeror who is ready, willing and able to consummate the purchase and who is neither the Company nor an Affiliate of that Member.

1.10 "Board" means the Board of Directors of the Company established pursuant to Section 5.1(a) and consisting of all of the Board Members.

1.11 "Board Member" means each member of the Board appointed by the Members pursuant to Section 5.1(a).

1.12 "Budget" means the budget for the operations of the Company and its Subsidiaries for a Fiscal Year, including without limitation, clearing, planting and farm management and all activities relating thereto (but excluding the cost of acquiring any real estate), setting forth month by month information for such Fiscal Year. The initial Budget approved by the Members is attached hereto as Exhibit B.

1.13 "Business Day" means any day of the year in which banks are not required or authorized to close in Los Angeles, California.

1.14 "<u>Capital Account</u>" of a Member means the capital account of that Member determined from the inception of the Company strictly in accordance with the rules set forth in Section 1.704-1(b)(2)(iv) of the Treasury Regulations. In the event that assets of the Company other than cash are distributed to a Member in kind, Capital Accounts shall be adjusted for the hypothetical "book" gain or loss that would have been realized by the Company if the distributed assets had been sold for their Fair Market Values in a cash sale (in order to reflect unrealized gain or loss). In the event of the liquidation of the Company, Capital Accounts shall be adjusted for the hypothetical "book" gain or loss that would have been realized by the Company if all Company assets had been sold for their Fair Market Values in a cash sale (in order to reflect unrealized gain or loss). In the event of the liquidation of the hypothetical "book" gain or loss. In the event of the hypothetical "book" gain or loss. In the event of the liquidation of the Company, Capital Accounts shall be adjusted for the hypothetical "book" gain or loss. In the company assets had been sold for their Fair Market Values in a cash sale (in order to reflect unrealized gain or loss). There shall be only one Capital Account for each Member, regardless of whether the Member owns more than one class of interest in the Company.

1.15 "Capital Contribution" of a Member, at any particular time, means the amount of money or property which that Member has theretofore contributed to the capital of the Company.

1.16 "Certificate of Formation" means the Certificate of Formation of the Company as filed under the Act with the Delaware Secretary of State.

1.17 "Closing" has the meaning specified in Section 7.8.

1.18 "Code" means the Internal Revenue Code of 1986.

1.19 "Common Member" means a Member who holds Common Units.

1.20 "Common Unit" means a unit of Membership Interest having the rights and obligations specified with respect to Common Units in this Agreement.

1.21 "Company" means GCE Mexico I, LLC, a Delaware limited liability company.

1.22 "<u>Company Minimum Gain</u>" with respect to any taxable year of the Company means the "partnership minimum gain" of the Company as determined for "book" purposes and computed strictly in accordance with the principles of Section 1.704-2(d) of the Treasury Regulations.

1.23 "Distributable Cash" at any time means that portion of the cash then on hand or in bank accounts of the Company which the Board deems available for distribution to the Members, taking into account (a) the amount of cash required for the payment of all current expenses, liabilities and obligations of the Company (whether for expense items, capital expenditures, improvements, retirement of indebtedness or otherwise), including, without limitation, all accrued interest and any other amounts payable under the Land Acquisition Loans, and (b) the amount of cash necessary to establish prudent reserves for the payment of future capital expenditures, improvements, retirements of indebtedness, amounts that will become payable under the Land Acquisition Loans, operations and contingencies, known or unknown, liquidated or unliquidated, including, but not limited to, liabilities which may be incurred in litigation and liabilities undertaken pursuant to the indemnification provisions of this Agreement. Notwithstanding anything in this Agreement to the contrary, Distributable Cash shall be computed so that, after the distribution of Distributable Cash under Sections 6.10(a) and (b), one dollar of cash or distributable for distribution to Members under Section 6.10(c); this sentence shall apply until all principal, accrued interest and other amounts payable under the Land Acquisition Loans have been paid.

1.24 "Distribution" means the transfer of money or property by the Company to one or more Members without separate consideration.

1.25 "Economic Interest" means a share, expressed as a percentage, of one or more of the Company's Net Profits, Net Losses, Tax Credits, Distributable Cash or other Distributions, but does not include any other rights of a Member, including, without limitation, the right to vote or participate in the management of the Company or the right to information concerning the business and affairs of the Company.

1.26 "Economic Risk of Loss" means the economic risk of loss within the meaning of Section 1.752-2 of the Treasury Regulations.

1.27 "Exculpatory Liability" means a liability that is treated as an "exculpatory liability" pursuant to Part V.B of Treasury Decision 8385, 56 Federal Register 66978-66995 (December 27, 1991).

1.28 "Fair Market Value" means, with respect to an asset or group of assets, the price at which that asset or group of assets would be sold for cash payable at closing between a willing buyer and a willing seller, each having reasonable knowledge of all relevant facts concerning the asset or group of assets and neither acting under any compulsion to buy or sell.

1.29 "Fiscal Year" means the Company's fiscal year, which shall be the calendar year.

1.30 "Former Member" has the meaning specified in Section 8.2.

1.31 "Former Member's Interest" has the meaning specified in Section 8.2.

1.32 "GCE" means Global Clean Energy Holdings, Inc. or any permitted successor-in-interest to some or all of its Membership Interest.

1.33 "GCE ROFO Notice" has the meaning specified in Section 7.6(b).

1.34 "GCE ROFR Notice" has the meaning specified in Section 7.6(b).

1.35 "Land Acquisition Loan" has the meaning specified in Section 3.2(a).

1.36 "Lender" has the meaning specified in Section 3.2(b).

1.37 "Manager" shall mean GCE or any successor manager approved by the Preferred Members pursuant to Section 5.2(e) or unanimously approved by all of the Board Members.

1.38 "<u>Member</u>" means each Person who (a) is an initial signatory to this Agreement, has been admitted to the Company as a Member in accordance with the Certificate of Formation or this Agreement or is a transferee of a Member who has become a Member in accordance with ARTICLE VII, and (b) has not suffered a Membership Termination Event or Transferred its entire Membership Interest in accordance with the provisions of ARTICLE VII.

1.39 "Member IP" has the meaning specified in Section 5.8(b).

1.40 "Member Minimum Gain" has the meaning given to the term "partner nonrecourse debt minimum gain" in Section 1.704-2(i) of the Treasury Regulations.

1.41 "<u>Member Nonrecourse Debt</u>" means any "partner nonrecourse liability" or "partner nonrecourse debt" under Section 1.704-2(b)(4) of the Treasury Regulations. Subject to the foregoing, it means any Company liability to the extent the liability is nonrecourse for purposes of Section 1.1001-2 of the Treasury Regulations, and a Member (or related Person within the meaning of Section 1.752-4(b) of the Treasury Regulations) bears the Economic Risk of Loss under Section 1.752-2 of the Treasury Regulations because, for example, the Member or related Person is the creditor or a guarantor.

1.42 "<u>Member Nonrecourse Deductions</u>" means the Company deductions, losses and Code Section 705(a)(2)(B) expenditures, as the case may be (as computed for "book" purposes), that are treated as deductions, losses and expenditures attributable to Member Nonrecourse Debt under Section 1.704-2(i)(2) of the Treasury Regulations.

1.43 "<u>Membership Interest</u>" means a Member's total interest as a member of the Company, including that Member's share of the Company's Net Profits, Net Losses, Distributable Cash or other Distributions, its right to inspect the books and records of the Company and its right, to the extent specifically provided in this Agreement, to participate in the business, affairs and management of the Company and to vote or grant consent with respect to matters coming before the Company.

1.44 "<u>Membership Termination Event</u>" with respect to any Member means one or more of the following: the insanity, permanent disability, withdrawal, resignation, Bankruptcy, dissolution or an attempted Transfer of a Member's Membership Interest or Economic Interest which is not made in accordance with the provisions of ARTICLE VII.

1.45 "<u>Net Profits</u>" and "<u>Net Losses</u>" mean, for each Fiscal Year of the Company, the net income or net loss, respectively, of the Company. For this purpose, "income" shall refer to all items (other than Capital Contributions) that increase capital accounts under Treasury Regulations Section 1.704-1(b)(2)(iv) (such as "book" gain and income), and "loss" shall refer to all items (other than Distributions) that decrease capital accounts under Treasury Regulations Section 1.704-1(b)(2)(iv) (such as "book" deduction and loss). Notwithstanding the foregoing, all items specially allocated under Sections 6.1 through 6.7 shall be excluded from the computation of Net Profits and Net Losses.

1.46 "Nonrecourse Deductions" in any fiscal period means the amount of Company "book" deductions that are characterized as "nonrecourse deductions" under Treasury Regulations Section 1.704-2(b) of the Treasury Regulations.

1.47 "Non-Transferring Member" has the meaning specified in Section 7.6(a).

1.48 "Notice" has the meaning specified in Section 7.7(a).

1.49 "Offering Member(s)" has the meaning specified in Section 7.7(a).

1.50 "Other Member(s)" has the meaning specified in Section 7.7(a).

1.51 "Percentage Interest" means the percentage interest of a Member with respect to Common Units set forth opposite the name of that Member in <u>Exhibit A</u>, as such percentage may be adjusted from time to time pursuant to the provisions of this Agreement.

1.52 "Person" means any entity, corporation, company, association, joint venture, joint stock company, partnership (whether general, limited relimited liability), trust, limited liability company, real estate investment trust, organization, individual (including any personal representative, executor or heir of a deceased individual), nation, state, government (including any agency, department, bureau, board, division or instrumentality thereof), trustee, receiver or liquidator.

1.53 "Preferred Member" means a Member who holds Preferred Units.

1.54 "Preferred Return" of a Preferred Member means a cumulative preferential rate of return in an amount equal to twelve percent (12%) per annum (compounded annually on January 1 of each year), prorated for fractional periods, on the amount of that Member's Adjusted Capital Contribution. The Preferred Return shall be computed on the basis of a computational year of 360 days comprised of equal months of 30 days each. Contributions shall not bear the Preferred Return for the day on which funds are contributed to the Company; however, funds distributed to Members that reduce a Member's Adjusted Capital Contribution shall bear the Preferred Return for the day on which those funds are distributed to the Member.

1.55 "Preferred Unit" means a unit of Membership Interest having the rights and obligations specified with respect to Preferred Members and Preferred Units in this Agreement.

1.56 "Project" means the acquisition by the Company, whether directly or indirectly through one or more Subsidiaries, of up to 2,100 hectares of land located in the State of Yucatan, in Mexico to be used primarily for planting, growing and harvesting Jatropha curcas, and the marketing, distribution and sale of the resulting fruit, seeds, or pre-processed Crude Jatropha Oil, as biodiesel feedstock, biomass or otherwise, and the sale or utilization of environmental attributes relating thereto, including without limitation, carbon value, green fuel value, or renewable energy credit value.

1.57 "Purchase Price" has the meaning specified in Section 7.7(a)

1.58 "Resnick Trust" means the Stewart and Lynda Resnick Revocable Trust dated December 27, 1998, as amended, or any permitted successor-in-interest to some or all of its Membership Interest.

1.59 "Resnick/Zilkha Members" has the meaning specified in Section 7.6(b).

1.60 "ROFO Notice" has the meaning specified in Section 7.6(a).

1.61 "ROFR Notice" has the meaning specified in Section 7.6(a).

1.62 "Subsidiary" means with respect to any Person, any corporation, association, joint venture, partnership, limited liability company or other business entity (whether now existing or hereafter organized) of which at least a majority of the voting stock or other ownership interests having ordinary voting power for the election of directors (or the equivalent) is, at the time as of which any determination is being made, owned or controlled by such Person or one or more subsidiaries of such Person or by such Person and one or more subsidiaries of such Person.

1.63 "Tax Credits" means all credits against income or franchise taxes and credits allowable to Members under state, federal or other tax statutes.

1.64 "<u>Tax Matters Partner</u>" means the Member appointed pursuant to the provisions of Section 9.3 to serve as the "tax matters partner" of the Company for purposes of Sections 6221-6233 of the Code. Initially, the Tax Matters Partner shall be GCE.

1.65 "<u>Transfer</u>" means, with respect to a Membership Interest or any interest therein, the sale, assignment, transfer, disposition, pledge, hypothecation or encumbrance thereof, whether direct or indirect, voluntary, involuntary or by operation of law, and whether or not for value, of (a) all or any part of that Membership Interest or interest therein or (b) in the case of GCE, a controlling interest in any Person which directly or indirectly through one or more intermediaries holds GCE's Membership Interest or interest therein.

- 1.66 "Transferring Member" has the meaning specified in Section 7.6(a).
- 1.67 "Treasury Regulations" means the regulations of the United States Treasury Department pertaining to the income tax.
- 1.68 "Units" means either Common Units and/or Preferred Units.

1.69 "<u>Unpaid Preferred Return</u>" of a Member means the excess of (i) the Preferred Return of the Member accrued to date over (ii) the sum of all Distributions to the Member under Section 6.10(a) and the Distributions to the Member under Section 10.5(a) that shall constitute return of capital Distributions. Distributions to a Member under Section 10.5(a) first shall constitute return of capital Distributions to the extent that these Distributions reduce the Member's Adjusted Capital Contribution to an amount not less than zero, and thereafter Distributions under Section 10.5(a) shall constitute Distributions of Unpaid Preferred Return to the extent that these Distributions reduce the Member's Unpaid Preferred Return to an amount not less than zero.

1.70 "Zilkha Members" means the Zilkha Trust, DMZ 2000 Trust, LLZ 2000 Trust, JW 2000 Trust and DW 2000 Trust, or any permitted successor-in-interest to some or all of any of their Membership Interests.

- 1.71 "Zilkha Transferor" has the meaning specified in Section 7.5(a)
- 1.72 "Zilkha Trust" means the Selim K. Zilkha Trust or any permitted successor-in-interest to some or all of its Membership Interest.

References in this Agreement to "Articles," "Sections," "Exhibits" and "Schedules," shall be to the Articles, Sections, Exhibits and Schedules of this Agreement, unless otherwise specifically provided; all Exhibits and Schedules to this Agreement are incorporated herein by reference; any of the terms used in this Agreement may, unless the context otherwise requires, be used in the singular or the plural and in any gender depending on the reference; the words "herein", "hereof" and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement; and except as otherwise specified in this Agreement, all references in this Agreement (a) to any Person shall be deemed to include such Person's permitted heirs, personal representatives, successors and assigns; and (b) to any agreement, any document or any other written instrument shall be a reference to such agreement, document or instrument together with all exhibits, schedules, attachments and appendices thereto, and in each case as amended, restated, supplemented or otherwise modified from time to time in accordance with the terms thereof; and (c) to any law, statute or regulation shall be deemed references to such law, statute or regulation shall be deemed der of regulation as the same may be supplemented, amended, consolidated, superseded or modified from time to time.

ARTICLE II

ORGANIZATIONAL MATTERS

2.1 Name. The name of the Company shall be "GCE MEXICO I, LLC." The business of the Company may be conducted under that name or, upon compliance with applicable law, under any other name that the Board deems appropriate or advisable.

2.2 Term. The term of the Company's existence commenced upon the filing of its Certificate of Formation with the Delaware Secretary of State on February 27, 2008 and shall continue until such time as it is terminated pursuant to ARTICLE X.

2.3 Office and Agent. The principal office of the Company shall be at 6033 W. Century Blvd., Suite 1090, Los Angeles, California 90045 or at such other place as the Board may determine from time to time. The Company may also have such offices within and without the State of California as the Board may from time to time determine. The name and business address of the Company's agent for service of process in the State of Delaware is Corporation Trust Center, 1209 Orange Street, in the city of Wilmington, County of New Castle, or as may otherwise be determined by the Board from time to time.

2.4 <u>Purpose of Company</u>. The Company may engage in any lawful activity for which a limited liability company may be organized under the Act; however, its primary purpose shall be to engage in the Project to establish a commercial product focused on the growing of the Jatropha curcas to supply crude Jatropha oil suitable for use as a biodiesel feedstock and to take all actions relating thereto. Notwithstanding the foregoing, the Company shall not engage in any business unrelated to the Project or in furtherance of the purposes of the Company, unless the Board consents thereto.

2.5 Intent. It is the intent of the Members that the Company shall always be operated in a manner consistent with its treatment as a "partnership" for Federal and state income tax purposes. It also is the intent of the Members that the Company not be operated or treated as a "partnership" for purposes of Section 303 of the United States Bankruptcy Code. No Member shall take any action inconsistent with that express intent.

2.6 Formation Expenses. GCE shall be responsible for and shall pay all fees and expenses incurred by it in connection with the formation of the Company, including, without limitation, all legal and accounting fees and expenses incurred by it in connection with the negotiation, preparation, execution and delivery of this Agreement. The Company shall pay, or reimburse the Resnick Trust and the Zilkha Trust for the payment of, fees and expenses incurred by the Resnick Trust and the Zilkha Trust in connection with the formation of the Company, including, without limitation, all legal and accounting fees and expenses incurred by the min connection with the negotiation, preparation, execution and delivery of this Agreement. The Company shall pay all filing fees, minimum franchise or other similar taxes and other governmental charges incident to its formation and qualification to do business.

ARTICLE III

CAPITAL CONTRIBUTIONS

3.1 Units.

(a) <u>Authorized Units</u>. The authorized Units which the Company has authority to issue consists of 1,000 authorized Common Units, and 1,000 authorized Preferred Units. The ownership by a Member of Units shall entitle such Member to allocations of Net Profit and Net Loss and other items and Distributions as set forth in ARTICLE VI hereof. On the date hereof, the Company has issued to each of the Members the number of Common Units and Preferred Units set forth opposite the Member's name on Exhibit A attached hereto.

(b) Capital Contributions.

(i) Common Units. No Capital Contribution is required with respect to the issuance of Common Units.

(ii) <u>Preferred Units</u>. With respect to the Preferred Units, each Preferred Member shall make Capital Contributions, up to the aggregate amount set forth on <u>Exhibit A</u> attached hereto, as described in this paragraph. At least ten (10) Business Days prior to the commencement of any calendar quarter, the Manager shall provide written notice to each Preferred Member of the funds necessary for the Company's operations, as reflected in the then current Budget approved by the Board, for such quarter. At least thre (3) Business Days prior to the commencement of such quarter, each Preferred Member shall make a Capital Contribution in immediately available funds to the Company in an amount equal to fifty percent (50%) of the necessary funds set forth in such notice; provided, however, that, notwithstanding anything to the contrary contained herein, (A) the aggregate Capital Contribution obligation for each Preferred Member under this subparagraph shall not exceed \$1,116,312 and (B) in the event of a conflict between the amount of funds requested in the amount of funds requested in the then current Budget approved by the Board, the Capital Contribution obligations for a Preferred Member with respect to such request shall be limited to fifty percent (50%) of the lesser of the two amounts.

(iii) <u>Additional Capital Contributions</u>. No Member shall be required to make any Capital Contributions other than as described in this Section 3.1. To the extent approved by the Board, from time to time, the Members may be permitted to make additional Capital Contributions if and to the extent they so desire, and if the Members determine that such additional Capital Contributions are necessary or appropriate for the conduct of the Company's business. In that event, the Members shall have the opportunity, but not the obligation, to participate in such additional Capital Contributions on a pro rata basis in accordance with their Percentage Interests, and such Percentage Interests and Sections 6.8, 6.9 and 6.10 shall be modified in such manner as agreed to by the Board and the Members.

(iv) Each Member shall receive a credit to its Capital Account in the amount of any capital which it contributes to the Company. The Board shall update Exhibit A from time to time to credit each Member with the amount of any additional Capital Contributions hereafter made by such Member and additional Units issued to such Members.

(c) <u>Unit Rights</u>. Each class of Units, and each Unit in a class, shall have the rights and obligations described in this Agreement, including without limitation, with respect to Capital Account balances, allocations of Net Profit and Net Loss, Distributions, approval rights, Transfer restrictions and purchase and sale rights and obligations.

3.2 Loans.

(a) Loans for Land Acquisition. Each of the Resnick Trust and the Zilkha Trust shall lend up to \$1,026,000 to AGC to be used solely for the purpose of purchasing land necessary for the Project (each, a "Land Acquisition Loan") upon the satisfaction of all of the following terms and conditions: (a) the Board has authorized AGC to execute and deliver an agreement for the acquisition of real estate for the Project, (b) the Manager has given each of the Resnick Trust and the Zilkha Trust at least ten (10) Business Days prior written notice of the expected closing date for such acquisition and the amount of funds necessary for AGC to satisfy its obligations under such acquisition agreement, (c) AGC has executed and delivered to the Resnick Trust and the Zilkha Trust the promissory note and mortgage in the form of Exhibit C, and D, respectively, attached hereto; provided, however, that (i) the interest payable under such Land Acquisition Loans shall be twelve percent (12%) per annum, compounded annually on January 1 of each year, prorated for fractional periods, computed on the basis of a computational year of 360 days comprised of equal months of 30 days each, with no interest accruing for the day on which the loan is made, but with interest accruing for the day on which the loan is made, but with interest accruing for the day on which the loan is paid, (ii) the motivity date shall be ten (10) years after the date of such Land Acquisition Loans. Provided that such terms and conditions have been satisfied, each of the Resnick Trust and the Zilkha Trust shall, on the expected closing date, make a Land Acquisition Loan to AGC in immediately available funds in an amount equal to fifty percent (50%) of the necessary funds set forth in such notice; provided, however, that, notwithstanding anything to the contrary contained herein, the obligation for each of the Resnick Trust and the Zilkha Trust to make a Land Acquisition Loan under this Section shall not exceed \$1,026,000.

(b) <u>Treatment of Loans</u>. To the fullest extent permitted by law, all principal, interest, costs and expenses due and payable by the Company to the Members or Affiliates thereof in repayment of loans shall be treated in the same manner as liabilities payable to unaffiliated creditors of the Company and shall be paid and taken into account, as such, before any Distributions of Distributable Cash are made to the Members. Without limiting the foregoing, the Members acknowledge that any Member or Affiliate of a Member ("Lender") who loans money to the Company shall have rights, the exercise of which will be in conflict with the Company's best interests. In that regard, the Members hereby authorize, agree and consent to the Lender's exercise of any of Lender's rights under any promissory note, deed of trust, security agreement or other loan document, even though the Lender's exercise of those rights may be detrimental to the Company or the Company's business. Further, the Members agree that any Lender's proper exercise of the rights shall not be deemed a breach of that Lender's fiduciary duties (if any) to the Company.

3.3 Capital Accounts. The Company shall establish and maintain an individual Capital Account for each Member.

3.4 <u>No Priorities of Members: No Withdrawals of Capital</u>. Except as otherwise specified in ARTICLE VI, ARTICLE X and in the Act, no Member shall have a priority over any other Member as to any Distribution, whether by way of return of capital or by way of profits, or as to any allocation of Net Profits or Net Losses. No Member shall have the right to withdraw or reduce its Capital Contributions in the Company except as a result of the dissolution of the Company or as otherwise provided in the Act, and, except as provided in Section 10.3, no Member shall have the right to demand or receive property other than cash in return for its Capital Contributions or Membership Interest.

3.5 No Interest on Capital Contributions. No Member shall be entitled to receive any interest on its Capital Contributions; it being acknowledged that the Preferred Return does not constitute interest. This Section shall not restrict the right of any member to receive interest on loans made to the Company by such Member.

ARTICLE IV

MEMBERS

4.1 Limited Liability. Except as required under the Act or as expressly set forth in this Agreement, no Member shall be personally liable for any debt, obligation or liability of the Company, whether that liability or obligation arises in contract, tort or otherwise.

4.2 Admission of Additional Members. Subject to compliance with applicable law and the approval of the Board, additional Members may be admitted to the Company from time to time upon such terms and conditions as the Board may determine, and any such additional Members shall be granted Membership Interests and may participate in the management, Distributable Cash, Net Profits, Net Losses, Tax Credits and other Distributions of the Company on such terms as the Board may fix; provided, however, that if such terms may require amendment to this Agreement, all of the Members must also consent to such amendment.

4.3 Withdrawal. No Member may withdraw or resign from the Company except with the prior written consent of the other Members which consent may be given or withheld, conditioned or delayed in the other Members' sole discretion. Any such permitted withdrawal or resignation of a Member shall constitute a Membership Termination Event, and upon the occurrence thereof, that Member's Membership Interest may, at the election of the holders of a majority of the Units held by all other Members, either be converted to a bare Economic Interest or purchased as provided in Section 8.2. In addition, such Member will be liable to the Company and the other Members for all damages suffered by the Company and the other Members as a result of such withdrawal.

4.4 <u>Members Are Not Agents</u> No Member, acting solely in its capacity as a Member, may be an agent of the Company, nor may any Member, in that capacity, bind or execute any instrument on behalf of the Company without the prior written consent of the Manager.

4.5 Meetings of Members; Written Consent. Meetings of the Members shall be held at such times and places within or without the State of California as the Members may fix from time to time, but, in any event, any Member may call a special meeting of the Members upon fourteen (14) days prior written notice to the other Members. No annual, regular or special meetings of Members are required, but if such meetings are held, they shall be conducted pursuant to the Act. Members may participate in any meeting through the use of conference telephones or similar communications equipment as long as all Members participating can hear one another. A Member so participating is deemed to be present in person at the meeting. Any action which may be taken by the Members at a meeting may also be taken without a meeting, if a consent in writing setting forth the action so taken is signed by all the Members. A consent transmitted by electronic transmission by a Member or other Person authorized to act for that Member shall be deemed to be written and signed by that Member for these purposes, and the term "electronic transmission" means any form of communication not directly involving the physical transmission of paper that creates a record that may be retained, retrieved and reviewed by a recipient thereof and that may be directly reproduced in paper form by such a recipient through an automated process. Except where any of the Preferred Members have been given the right to act alone, the affirmative unanimous vote of all the Members shall be required for the Members to approve any action. The Zilkha Member shall vote as a block.

4.6 <u>Member Approvals</u>. Except where this Agreement requires the consent or approval of all of the Members, the consent or approval of the Members shall occur at such time as Members holding at least a majority of the Units in each class have given their consent or approval. Except where this Agreement requires the consent or approval of all of the Members of a class of Units, the consent or approval of Members holding a class of Units shall occur at such time as Members holding at least a majority of the Units in such class have given their consent or approval of the Units in such class have given their consent or approval of Members holding a class of Units shall occur at such time as Members holding at least a majority of the Units in such class have given their consent or approval

ARTICLE V

MANAGEMENT AND CONTROL OF THE COMPANY

5.1 Board of Directors.

(a) Election and Term of Board. The business and affairs of the Company shall be managed under the direction of the Board which shall be constituted in accordance with this Section 5.1. The expression of any power or authority of the Board in this Agreement shall not in any way limit or exclude any other power or authority of the Board which is not specifically or expressly set forth in this Agreement. The Members shall designate the members of the Board (the "<u>Board Members</u>") as follows: two (2) Board Members shall be designated by GCE and two (2) Board Members shall be designated by the Preferred Members, with each of the Resnick Trust and the Zilkha Trust having the right to designate one (1) Board Member; provided, however, that (a) if GCE shall be removed as the Manager pursuant to Section 5.2(e), the Board Members designated by GCE shall thereupon be deemed removed as Board Members and GCE shall have no further right to designate Board Members, or (b) if a Member's Membership Interest is converted to an Economic Interest, whether as a result of a Membership Termination Event, a Transfer to a Person who is not then admitted as a substituted Member or otherwise, any Board Member designated by such Member shall be deemed removed and such Member or its successor or transferee shall have no further right to designate any Board Member and FCE and the Preferred Members shall have the right to change the identity of the Board Member appointed by it at any time and for any reason, by written notice to the other Member, and each Board Member so appointed shall serve in that capacity until he or she resigns or is removed by the Member which appointed him or her, in its absolute discretion. Initially, Richard Palmer and Bruce K. Nelson shall be the Board Members appointed by GCE, and Stewart A. Resnick and Selim Zilkha shall be the Board Members appointed by the Preferred Members.

(b) Meetings of the Board. Meetings of the Board may be called at any time, upon five (5) Business Days' prior notice, by any Board Member or by the Manager; provided, however, that in case of an emergency such meeting may be called on one (1) Business Day's notice. Unless the Board agrees otherwise, such meetings shall be held at the principal offices of the Company. A notice of a meeting need not specify the purpose of that meeting, and notice of the meeting need not be given to any Board Member who signs a waiver of notice, a consent to holding the meeting or an approval of the minutes thereof, whether before or after the meeting, or who attends the meeting without protesting the lack of notice prior to the commencement of the meeting. Any Board Member may grant any other Board Member a proxy to act in his or her place at any meeting of the Board. Board Members may participate in any meeting of the Board by means of conference telephones or similar communications equipment so long as all Board Members in office shall be required for the Board to approve any action.

(c) <u>Written Consent</u>. Any action which may be taken by the Board Members at a meeting may also be taken without a meeting, if a consent in writing setting forth the action so taken is signed by all the Board Members. A consent transmitted by electronic transmission by a Board Member or other Person authorized to act for that Board Member shall be deemed to be written and signed by that Board Member for these purposes, and the term "electronic transmission" means any form of communication not directly involving the physical transmission of paper that creates a record that may be retained, retrieved and reviewed by a recipient thereof and that may be directly reproduced in paper form by such a recipient through an automated process.

(d) <u>Standard of Care</u>. Every Board Member shall discharge his or her duties in good faith, with the care an ordinary prudent person in a like position would exercise under similar circumstances, and in a manner he or she reasonably believes to be in the best interests of the Company and its Members; provided, however, that Board Members may, in their sole discretion, take actions which are in the best interests of the Member who appointed them regardless of whether such action is in the best interest of the Company or the other Members.

(e) <u>Limitation of Liability</u>. Except as otherwise prohibited by the Act, a Board Member shall not be liable, responsible or accountable for damages or otherwise to the Company for any action taken or failure to act on behalf of the Company within the scope of the authority conferred on the Board Members by this Agreement, by law or by the Members, unless such action or omission is performed or omitted fraudulently or constitutes willful misconduct or gross negligence. In no event shall the Board Members be liable to the Company, the Members or any of their respective Affiliates or constituent owners for any consequential, indirect, incidental or special damages arising from their acts or omissions.

(f) <u>Reliance</u>. In performing their duties, the Board Members shall be entitled to rely on information, opinions, reports or statements, including financial statements and other financial data, of any attorney, independent accountant or other Person as to matters which the Board Members believe to be within such Person's professional or expert competence unless the Board Members have actual knowledge concerning the matter in question that would cause such reliance to be unwarranted.

(g) Devotion of Time. The Board Members shall not be obligated to devote all of their time or business efforts to the affairs of the Company.

(h) <u>Actions Requiring Board Approval</u>. Without limiting the matters or actions that must be approved by the Board, the approval of the Board shall be required for any of the following actions:

(i) other than the declaration and payment of the Preferred Return, the declaration or payment of any Distributions on any Membership Interests;

(ii) the sale, exchange or other disposition of all, or substantially all, of the assets of the Company or any of its Subsidiaries occurring as part of a single transaction or plan, or in a series of transactions, except in the orderly liquidation and winding up of the business of the Company and its Subsidiaries upon its duly authorized dissolution;

(iii) the merger of the Company or any of its Subsidiaries with another limited liability company or a corporation, general partnership, limited partnership or other Person;

(iv) except to secure a Land Acquisition Loan, the encumbrance of any significant asset of the Company or any of its Subsidiaries, including without limitation, any real property owned by the Company or any of its Subsidiaries;

(v) any issuance of Membership Interests or any increase or decrease in the number of authorized Membership Interests or any redemption or repurchase of Membership Interests;

(vi) any change to the rights, preferences, and privileges of any class of Membership Interests or issuance of any Membership Interests;

(vii) any increase or decrease in the size of the Board;

(viii) the approval of the Budget and any changes in any line item of the Budget (except for changes that move line items between categories in the Budget as long as such change does not increase the Budget or alter the overall Budget); provided, however, that if the Board does not approve a Budget for a new Fiscal Year, the Budget for such year shall be the same as the then current Budget as adjusted to reflect increases for increased government assessments, cost increases under existing contracts and other increases consistent with the increase in consumer prices over the prior Fiscal Year which occurred for the area in which the Company operates as such increases in consumer prices are determined by a governmental agency or other Person approved by the Board;

- (ix) the admission of another Person as a Member of the Company;
- (x) any alteration of the primary purpose of the Company as set forth in Section 2.4;
- (xi) any decision to place the Company or any of its Subsidiaries into Bankruptcy; or
- (xii) any amendment to the Certificate of Formation or this Agreement.

5.2 Manager.

(a) <u>Management by Manager</u>. Subject to the other provisions of this Agreement, the day-to-day operations of the Company shall be managed by one (1) Manager within the parameters set forth in the Budget and by the Board. The initial Manager is GCE, who hereby accepts such appointment. The Manager may at any time be replaced by a unanimous vote of the Board Members. Additionally, the Preferred Members may remove and replace the Manager in accordance with Section 5.2(e). The Manager is hereby designated as a "manager" pursuant to Section 18-402 of the Delaware Act. Notwithstanding the foregoing, the Manager shall owe the Company the duties of loyalty and due care of the type owed by the officers of a corporation to such corporation and its stockholders under the Laws of the State of Delaware.

(b) <u>Duties of the Manager</u>. The Manager shall undertake all actions approved by the Board and shall cause the Preferred Return to be paid annually to the Preferred Members to the extent that Distributable Cash exists and payment of the Preferred Return is not then prohibited under this Agreement or law or any third party agreement approved by the Board.

Except for situations in which this Agreement provides for the Manager to act in a certain manner or the approval of the Members or the Board is specifically required by the Act, the Certificate of Formation or this Agreement, the Manager shall have full, complete and exclusive authority, power and discretion to manage and control the business, property and affairs of the Company, to make all decisions regarding those matters, to supervise, direct and control the actions of the officers, if any, of the Company and to perform any and all other actions customary or incident to the management of the Company's business, property and affairs. Subject to the foregoing, the Manager shall control and direct the administration of the business and affairs of the Company in accordance this Agreement and with sound business practice, taking such steps as are necessary or appropriate in its reasonable judgment to conserve and enhance the value and profitability of the Company's business, property and affairs. The Manager shall be serving in a fiduciary role for all of the Members and shall seek to obtain the best possible terms and conditions for the Project in connection with the sale of all off-take from the Project, including, without limitation, controlling and managing the monetization of all potential revenue sources from the Project or the Company's operations, including but not limited to fruit, seeds, pre-processed crude Jatropha oil, biomass, and by-product, press cake and hulls revenue and carbon credits.

Without limiting any of the other duties of the Manager described in this Agreement, the duties of the Manager shall include, among other things, responsibility for identifying the land, aggregating, negotiating the purchase, managing the Project development process and the Company on a day-to-day basis, including, but not limited to the following: (i) providing Jatropha seedlings/rootstock from existing GCE plantings, nurseries, or seed stocks; (ii) sourcing additional seeds from suppliers; (iii) overseeing any additional nursery construction and operations; (iv) managing land clearing, planting, harvesting and maintenance of the plantation (all or parts of which may be accomplished through GCE employees or through sub contractors subject to the payment restrictions set forth herein); (v) leasing or building storage facilities for seed and finished product; (vi) coordinating and managing all logistics relating to the Company's operations; (vii) managing all sales efforts; (viii) negotiating all off-take agreements and (ix) preparing and submitting the Budget as described in Section 5.2(c).

(c) <u>Budget</u>. At least two (2) months prior to the commencement of each Fiscal Year (other than the Company's initial Fiscal Year), the Manager shall prepare and deliver to each Board Member a proposed Budget for such Fiscal Year. Such proposed Budget shall be in the form of <u>Exhibit B</u>, provided, however, that if the Preferred Members request that the proposed Budget be prepared in a different form, the Manager shall prepare the proposed Budget in such form. If the Board does not approve a proposed Budget, the Manager shall promptly revise the proposed Budget to reflect comments received from Board Members and submit such revised Budget to the Board Members for approval.

(d) <u>Liability of Manager</u>. The Manager shall not be liable to the Company or to any Member for any losses or damages suffered by them, except as the result of the Manager's fraud, deceit, gross negligence, reckless or intentional misconduct, embezzlement, breach of fiduciary duty, knowing violation of law or breach of this Agreement or any other obligations of the Manager hereunder (whether committed knowingly, negligently or otherwise) or as a result of an act from which the Manager derives an improper personal benefit.

(e) <u>Removal of Manager</u>. The Manager may be removed or replaced by holders of all of the Preferred Units at any time or from time to time, without liability to the Company or any of its Members, for fraud, deceit, gross negligence, reckless or intentional misconduct, embezzlement, breach of a fiduciary duty, a material violation of law or a breach of this Agreement, any act from which the Manager derives an improper personal benefit, a breach of the Manager's obligations hereunder (which shall include, without limitation, taking any action not authorized under this Agreement), the Manager's failure to operate the Company within the budgetary guidelines established by the Board and such failure has a material adverse effect on the Company, the inability of Richard Palmer or his assignee to render services on behalf of the Manager and the failure of GCE to obtain a replacement acceptable to the holders of the Preferred Units within thirty (30) days, or a dissolution, merger, liquidation or bankruptcy of the Manager. If the Manager is so removed or replaced, the Preferred Members shall have the right under Section 5.3 to purchase the Common Units and all of the Membership Interest owned by the Manager or to require the Manager to purchase all of the Units and Membership Interests held by the Resnick Trust and the Zilkha Members.

5.3 Buy-Sell Rights Upon Removal of GCE as Manager.

(a) <u>Buy-Sell Right</u>. If GCE is removed as the Manager pursuant to Section 5.2(e), then the Preferred Members may, at any time within ninety (90) calendar days after such removal, elect to either purchase the Membership Interest of GCE or to sell the Membership Interests of the Resnick Trust and the Zilkha Members to GCE upon the terms and conditions set forth herein; provided that such election by the Preferred Members shall not give rise to any purchase right afforded to GCE pursuant to Section 7. To exercise such right, the Preferred Members shall deliver written notice to GCE of their intention to exercise such right; provided, however, that the Preferred Members shall not be required to state in such notice whether they have elected to sell the Membership Interests of the Resnick Trust and the Zilkha Members to GCE or to purchase the Membership Interest of GCE.

(b) <u>Suspension of Exercise of Buy-Sell Right</u>. If GCE disputes the basis for its removal as the Manager pursuant to Section 5.2(e), GCE shall initiate an arbitration proceeding pursuant to Section 13.3 no later than 110 days following such removal. The failure to so initiate such proceeding during such period, or the termination of such proceeding by GCE, shall constitute an irrevocable waiver and release by GCE of, and an absolute bar against the exercise by GCE of, any right, remedy, or claim regarding its removal as the Manager. If an arbitration proceeding is so initiated within such period, then the exercise of the rights under this Section 5.3 shall be suspended until the earlier of the termination or dismissal of the proceeding or the rendering of a decision by the arbitrator. If the arbitrator finds that the removal of the Manager was not in accordance with Section 5.2(e), then any exercise of the rights under Section 5.3 with respect to such removal shall be considered void and of no force or effect.

(c) Determination of Fair Market Value of the Company. If the Members are unable to agree upon a purchase price for the respective Membership Interests within thirty (30) calendar days following the delivery of the notice of exercise (or if an arbitration proceeding as to the removal is initiated by GCE pursuant to Section 5.3(b), within thirty (30) calendar days following the termination or dismissal of such proceeding or the rendering of a decision by the arbitrator), they shall within the next thirty (30) calendar days agree upon the selection of an appraiser to value the Fair Market Value of the Company as of the date of removal (or other repurchase event). If no agreement can be reached as to the selection of an appraiser, the Preferred Members shall promptly choose one appraiser by notice to the Preferred Members; provided, however, that all appraisers selected by the parties shall be reasonably experienced in valuing interests in businesses similar to the business then conducted by the Company. Each appraiser shall, within twenty (20) calendar days after his or her appointment, prepare an appraisal of the Fair Market Value of the Company as of the date of removal (or other repurchase event). In making the appraisal, the appraisers shall be obligated to take into account the value of comparable companies if known, and the present value of future cash flows to be derived from the existing assets of the Company and its Subsidiaries. The arithmetic average of the two appraisals shall be the Fair Market Value of the Company.

If either GCE or the Preferred Members fail to appoint an appraiser within thirty (30) calendar days after the lapse of the initial thirty (30) calendar day period referred to above, then the appraiser appointed by the party which does appoint an appraiser shall alone determine the Fair Market Value of the Company as of the date of removal (or other repurchase event), and its appraisal shall govern. Each party shall compensate the appraiser appointed by that party.

(d) <u>Applicable Purchase Price</u>. Within thirty (30) calendar days of such determination of the Fair Market Value of the Company, the Company's accountants shall prepare a schedule setting forth the amounts that would be distributed to each of the Members on the assumption that the Company completes a sale of its assets for, and receives cash equal to, such Fair Market Value as of the date of the notice, pays all outstanding obligations and customary closing and transaction costs that would have been likely to have been incurred if the Company was sold for such Fair Market Value, dissolves and then distributes the remaining balance to the Members in accordance with the terms of this Agreement. The amount that would be so distributed to each such Member shall be the purchase price (the "<u>Applicable Purchase Price</u>") for such Member's Membership Interest under this Section.

(e) Election. Within ten (10) calendar days of the determination of such Applicable Purchase Price, the Preferred Members shall notify GCE in writing as to whether they elect to purchase GCE's Membership Interest, or elect to require GCE to purchase the Membership Interests of the Resnick Trust and the Zilkha Members, for the Applicable Purchase Price; provided, however, that, notwithstanding anything to the contrary set forth in this Agreement: (i) the Applicable Purchase Price payable by any Member hereunder shall be reduced by an amount equal to all indebtedness then owed by the selling Member to that purchaser (without regard to whether or not such indebtedness is then due and payable in whole or in part); (ii) if GCE's Membership Interest is being purchased as a result of a breach of this Agreement by GCE, the Applicable Purchase Price payable by any Preferred Member hereunder shall be reduced by an amount equal to the damages suffered by that purchaser as a result of the breach; (iii) if at least five (5) Business Days prior to the Closing, GCE initiates an arbitration proceeding pursuant to Section 13.3 with respect to the amount of damages, then instead of withholding the amount of the damages from the payment of the purchase price, the Preferred Members shall at the Closing deposit the amount of the reduction for damages in an escrow which shall provide for the release of the funds to the Preferred Members if the arbitration is dismissed or terminated or otherwise in accordance with the findings of the arbitrator, and for all interest and earnings on such funds to be paid to the Preferred Members, and (iv) if the Preferred Members elect to require GCE to purchase the Membership Interests of the Resnick Trust and the Zilkha Members, then GCE must also cause AGC to pay to the Resnick Trust and the Zilkha Trust at the Closing all principal, accrued interest and other amounts payable under the Land Acquisition Loans; provided, however, if within ten (10) Business Days of the delivery of the notice from the Preferred Members, GCE provides written notice to the Preferred Members that the board of directors of GCE has made a good faith determination that GCE is unable to pay, or finance the payment of, the Applicable Purchase Price and cause AGC to pay to the Resnick Trust and the Zilkha Trust at the Closing all principal, accrued interest and other amounts payable under the Land Acquisition Loans, then the Preferred Members may elect to (A) dissolve the Company, (B) instead purchase the Membership Interests of GCE for the Applicable Purchase Price, or (C) cancel the exercise of its buy-sell rights under this Section 5.3, and any election by the Preferred Members shall be without prejudice to any right or remedy of the Preferred Members with respect to GCE.

5.4 <u>Transactions between the Company and the Manager or its Affiliates</u>. Subject to the terms and provisions of this Agreement, the Manager may provide, or cause the Company to engage one or more of its Affiliates to provide, any or all goods and/or services required by the Company or its Subsidiaries in the conduct of its business, provided that the Manager has first notified each of the Board Members in writing that the Manager or any of its Affiliates may be providing such goods and/or services to the Company or its Subsidiaries and the Board has approved the terms and conditions of any such engagement or the Budget expressly approved by the Board contains a line-item reflecting payments to the Manager or its Affiliates for such good or services.

5.5 Officers of the Company.

(a) Appointment of Officers. The Manager may, at its discretion, appoint officers of the Company at any time to conduct, or to assist the Manager in the conduct of, the day-to-day business and affairs of the Company. The officers of the Company may include a Chairperson, a President or Chief Executive Officer, one or more Senior Vice Presidents, one or more Vice Presidents, a Secretary, one or more Assistant Secretaries, a Chief Financial Officer, a Treasurer, one or more Assistant Treasurers and a Comptroller. The officers shall serve at the pleasure of the Manager, subject to all rights, if any, of an officer under any contract of employment; provided, however, that the Manager shall not enter into any employment agreement with an officer without the prior approval of the Board. Any individual may hold any number of offices. If a Manager is not an individual, that Manager's officers may serve as officers of the Company if appointed by the Manager. The officers shall exercise such powers and perform such duties as are typically exercised by similarly titled officers in a corporation and as shall be determined from time to time by the Manager, but subject in all instances to the supervision and control of the Manager.

(b) Signing Authority of Officers. The officers, if any, shall have such authority to sign checks, instruments and other documents on behalf of the Company as may be delegated to them by the Manager.

5.6 Limitations on Power of Manager and Officers. Notwithstanding any other provision of this Agreement, however, neither the Manager nor any officer of the Company or any of its Subsidiaries shall have any power or authority to approve or cause the Company or any of its Subsidiaries to engage in any of the following, without first obtaining the unanimous vote or written consent of all Board Members:

(a) entering into any contract with a supplier, customer, or partner that imposes material restrictions or limitations on the conduct of the Company or a Subsidiary such as, but not limited to, exclusivity provisions and non-compete provisions;

(b) entering into any contract that may restrict the payment of the Preferred Return;

(c) any borrowing of money (it being acknowledged that incurring customary trade payables in the ordinary course of business which when incurred are expected to be paid when due shall not be considered "the borrowing of money"), other than pursuant to Section 3.2, which, after giving effect to the borrowing, causes the Company or a Subsidiary to have more than \$25,000 in principal amount of borrowings outstanding;

(d) any loan by the Company or a Subsidiary to any Person, any guaranty by the Company or a Subsidiary of any other Person's obligations or any investment by the Company or a Subsidiary in the business of any other Person;

(e) any transaction between the Company or a Subsidiary and a Member or Manager or any Affiliate of a Member or Manager, or any transaction between the Company or a Subsidiary and any Person in which a Member or Manager or any Affiliate of a Member or Manager has a material financial interest;

(f) any act which would make it impossible to carry on the ordinary business of the Company or a Subsidiary;

(g) the formation by the Company or a Subsidiary of any Subsidiary;

(h) the formation of any joint venture or any investment by the Company or a Subsidiary in another Person; or

(i) any other act for which the approval of the Board is required under this Agreement.

5.7 <u>Competing Activities</u>. No Manager, Board Member or Member shall be obligated to present any prospective project, business venture, investment opportunity or economic advantage to the Company or any other Members, even if the opportunity is one of the character that, if presented to the Company or the other Members, could be taken by the Company or the other Members, and each Member shall have the right to hold any such prospective project, business venture, investment opportunity or economic advantage for its own account or to recommend the same to Persons other than the Company or the other Members. The Manager, Board Member, Members and their respective officers, directors, shareholders, partners, members, managers, agents, employees and Affiliates may engage or invest in, independently or with others, any business activity of any type or description, including without limitation those that might be the same as or similar to the Company's business and that might be in direct or indirect competition with the Company. Neither the Company nor the other Members shall have the right in or to such other ventures or activities or to the income or proceeds derived therefrom.

5.8 Intellectual Property.

(a) <u>Company Intellectual Property</u>. The Company shall, directly or indirectly through Subsidiaries, own all of the rights and assets of the Project, including, but not limited to land, equipment, seed procurement contracts, Jatropha oil off-take contracts, biomass sales and/or carbon credit sales agreements, as well as all intellectual property, whether patentable or not, created, developed, discovered, invented or acquired by the Company or any of its Subsidiaries relating to the Project. Any intellectual property created, developed, discovered or invented by GCE or its Affiliates or employees in connection with the performance of duties by or on behalf of GCE under this Agreement shall be considered developed by the Company and shall belong to the Company. GCE or its Affiliates or employees shall execute such intellectual property rights. Each Member shall have a non-transferable, non-exclusive, perpetual, non-sublicenseable (other than to the Member's Affiliates) royalty-free license to use any intellectual property rights owned by the Company, which license shall survive the termination of a Member's Membership Interest or a dissolution of the Company.

(b) <u>Member Intellectual Property</u>. The Members acknowledge that, absent a separate written agreement between the Company and any of its Members, the Company shall have no rights to any intellectual property created, developed, discovered, invented or acquired by any of the Members or any intellectual property of another Person (other than the Company) which is used by such Member with the consent of such Person (any such intellectual property is referred to as "<u>Member IP</u>") without regard to whether any Member IP may be similar to intellectual property developed by the Company or its Subsidiaries. Member IP shall remain the property of the applicable Member and not the Company, notwithstanding that a Member may license or otherwise permit the Company to use Member IP. Any Member IP provided by a Member to the Company for its use may be used by the Company on a royalty free basis.

5.9 Payments to the Manager and Others. The Company is authorized to pay any Person remuneration or reimbursement for goods and services provided to the Company; provided, however, that the Manager and its Affiliates shall be entitled to receive only the following remuneration or reimbursement:

(a) No Management Fee. The Manager shall not be entitled to any management fee or other compensation for its services as the Manager.

(b) <u>Services Performed by Members or Affiliates</u>. The Company shall pay the Members and their Affiliates for services rendered or goods provided by them to the Company to the extent that those Members or Affiliates are not required to render such services or goods themselves without charge to the Company, and to the extent that the fees paid to those Members or Affiliates do not exceed the fees that would be payable to independent responsible third parties that are willing to perform those services or provide those goods; provided, however, that in the case of the Manager or its Affiliates, the payment of such fees also is authorized under Section 5.4.

(c) Expenses. Subject to Section 5.10, the Company shall reimburse the Manager, Members and their respective Affiliates for all reasonable out-of-pocket costs and expenses incurred by them in connection with the business and affairs of the Company, as well as organizational expenses incurred by them to form the Company and to prepare the Certificate of Formation; provided, however, that (i) such costs and expenses are consistent with the Budget approved by the Board, (ii) the Company shall pay, or reimburse the Resnick Trust and the Zilkha Trust for the payment of, any legal and accounting fees incurred by them in connection with the preparation and negotiation of this Agreement, but GCE shall be solely responsible for the payment of its legal and accounting fees, (iii) GCE shall be solely responsible for the payment of any broker's, finders or comparable fees and costs relating to the formation, structuring and initial funding of the Company or the acquisition of any land by the Company or any Subsidiary, and (iv) unless expressly approved by the Board or in the Budget approved by the Board, the Manager and its Affiliates shall not be reimbursed by the Company for (A) any salaries, compensation or fringe benefits of directors, officers or employees of the Manager or their Affiliates or (B) any overhead expenses of the Manager or its Affiliates including, without limitation, rent and general office expenses.

5.10 GCE Obligations for Public Company Compliance. GCE represents and warrants to the other Members that it is a publicly held company and that GCE presently intends to consolidate the financial results of the Company with its financial results and that of its Subsidiaries. GCE acknowledges that (a) as a publicly held company, GCE may need the financial results of the Company to be audited and/or to institute accounting or operating procedures or controls to satisfy applicable laws, rules, regulations or listing requirements, and (b) if GCE were not a publicly held company, the Company would not incur costs with respect to an audit or such procedures or controls. GCE agrees that GCE shall be solely responsible for, and shall pay directly, any audit costs, costs of establishing any such accounting procedures or controls are not capable of being paid directly by GCE, GCE shall promptly reimburse the Company for such costs. By way of example only, if an employee of the Company spends 50% of his time with respect to company for 50% of the cost thereof (including all the cost of all related benefits and taxes).

ARTICLE VI

ALLOCATIONS OF NET PROFITS, NET LOSSES AND DISTRIBUTIONS

6.1 Minimum Gain Chargeback. In the event that there is a net decrease in the Company Minimum Gain during any taxable year, the minimum gain chargeback described in Sections 1.704-2(f) and (g) of the Treasury Regulations shall apply.

6.2 Member Minimum Gain Chargeback. If during any taxable year there is a net decrease in Member Minimum Gain, the partner minimum gain chargeback described in Section 1.704-2(i)(4) of the Treasury Regulations shall apply.

6.3 <u>Qualified Income Offset</u> Any Member who unexpectedly receives an adjustment, allocation or Distribution described in subparagraphs ($\underline{4}$), ($\underline{5}$) or ($\underline{6}$) of Section 1.704-1(b)(2)(ii)(\underline{d}) of the Treasury Regulations, which adjustment, allocation or distribution creates or increases a deficit balance in that Member's Capital Account, shall be allocated items of "book" income and gain in accordance with the provisions of the "qualified income offset" as described in Section 1.704-1(b)(2)(ii)(\underline{d}) of the Treasury Regulations.

6.4 Nonrecourse Deductions. Nonrecourse Deductions shall be allocated to the Members in proportion to their Percentage Interests.

6.5 <u>Member Nonrecourse Deductions</u>. Member Nonrecourse Deductions shall be allocated to the Members as required in Section 1.704-2(i)(1) of the Treasury Regulations in accordance with the manner in which the Members bear the burden of an Economic Risk of Loss corresponding to the Member Nonrecourse Deductions.

6.6 Income from Exculpatory Liabilities. Any allocations of items of "book" income or gain attributable to an Exculpatory Liability as contemplated by Part V.B of Treasury Decision 8385, 56 Federal Register 66978-66995 (December 27, 1991), shall be allocated to the Members who have received prior allocations of "book" deductions and losses allocable to the Exculpatory Liability, in accordance with the ratio of such prior allocations of "book" deductions and losses that have not been previously charged back by this Section 6.6.

6.7 Losses from Exculpatory Liabilities. Any allocations of items of "book" loss or deduction attributable to an Exculpatory Liability as contemplated by Part V.B of Treasury Decision 8385, 56 Federal Register 66978-66995 (December 27, 1991), shall be allocated to the Members in accordance with their Percentage Interests.

6.8 <u>Allocation of Net Profits</u>. The Net Profits for each fiscal period of the Company shall be allocated to the Members in accordance with the following order of priority:

(a) first, to those Members with negative Adjusted Capital Accounts, among them in proportion to the ratio of the negative balances in their Adjusted Capital Accounts, until no Member has a negative Adjusted Capital Account;

(b) second, to those Members whose Adjusted Capital Contributions are in excess of their Adjusted Capital Accounts, among them in accordance with the ratio of these excesses, until all of these excesses have been eliminated;

(c) third, to those Members for whom the sum of their Adjusted Capital Contributions and their Unpaid Preferred Returns are in excess of their Adjusted Capital Accounts, among them in accordance with the ratio of these excesses, until all of these excesses have been eliminated; and

(d) finally, to the Members in proportion to their Percentage Interests.

6.9 Allocation of Net Losses. Net Losses for each fiscal period of the Company shall be allocated to the Members:

(a) first, to those Members whose Adjusted Capital Accounts are in excess of the sum of their Adjusted Capital Contributions and their Unpaid Preferred Returns, among them in accordance with the ratio of these excesses, until all of these excesses have been eliminated.

(b) second, to those Members whose Adjusted Capital Accounts are in excess of their Adjusted Capital Contributions, among them in accordance with the ratio of these excesses, until all of these excesses have been eliminated; and

(c) third, to those Members whose Adjusted Capital Accounts are greater than zero, among them in accordance with the ratio of their positive Adjusted Capital Account balances, until no Member has a positive Adjusted Capital Account.

6.10 Distribution of Assets by the Company. Subject to applicable law and any limitations contained elsewhere in this Agreement, the Board (and only the Board) may elect from time to time to cause the Company to distribute Distributable Cash to the Members, which Distributions shall be in the following order of priority:

(a) first, to the Members with Unpaid Preferred Returns, in proportion to their Unpaid Preferred Return until each Member's Unpaid Preferred Return has been reduced to zero;

(b) second, to those Members with positive Adjusted Capital Contributions, in proportion to their positive Adjusted Capital Contributions, until each Member's Adjusted Capital Contribution has been reduced to zero; and

(c) third, to the Members in proportion to their Percentage Interests.

6.11 <u>Allocation of Net Profits and Losses in Respect of a Transferred Interest</u>. If any Membership Interest is Transferred or is increased or decreased by reason of the admission of a new Member or otherwise during any Fiscal Year, the varying interests rule shall be applied to allocate each item of income, gain, loss, and deduction on a pro rata basis based on the percentage that the number of days before the increase and decrease (including the day of increase or decrease) constitutes as a percentage of the number of days during the year.

6.12 Tax Allocation Matters.

(a) <u>Contributed or Revalued Property</u>. Each Member's allocable share of the taxable income or loss of the Company, depreciation, depletion, amortization and gain or loss with respect to any contributed property, or with respect to revalued property where the Company's property is revalued pursuant to Paragraph (b)(2)(iv)(f) of Section 1.704-1 of the Treasury Regulations, shall be determined in accordance with the "traditional method" as set forth in Section 1.704-3(b) of the Treasury Regulations.

(b) <u>Recapture Items</u>. In the event that the Company has taxable income that is characterized as ordinary income under the recapture provisions of the Code, then Sections 1.1245-1(e) and 1.1250-1(f) of the Treasury Regulations shall apply, and in the event that the Company has taxable income that is characterized as "unrecaptured Section 1250 gain" under Section 1(h)(6) of the Code, then the principles of such Treasury Regulations shall apply.

6.13 Order of Application. To the extent that any allocation, Distribution or adjustment specified in any of the preceding Sections of this ARTICLE VI affects the results of any other allocation, Distribution or adjustment required herein, the allocations, Distributions and adjustments specified in the following Sections shall be made in the priority listed:

(a) Section 6.10.
(b) Section 6.1.
(c) Section 6.2.
(d) Section 6.3.
(e) Section 6.4.
(f) Section 6.5.
(g) Section 6.6.
(h) Section 6.7.

(i) Section 6.8.

(j) Section 6.9.

(k) Section 10.5.

These provisions shall be applied as if all Distributions and allocations were made at the end of the Company's Fiscal Year. Where any provision depends on the Capital Account of any Member, that Capital Account shall be determined after the operation of all preceding provisions for the Fiscal Year.

6.14 <u>Allocation of Liabilities</u>. Each Member's interest in "partnership" profits for purposes of determining that Member's share of "excess nonrecourse liabilities" of the Company as used in Section 1.752-3(a)(3) of the Treasury Regulations, shall be equal to that Member's Percentage Interest.

6.15 Form of Distribution. Except as provided in Section 10.3, no Member, regardless of the nature of its Capital Contribution, has the right to demand and receive any Distribution from the Company in any form other than money unless it is approved by the Board. No Member may be compelled to accept from the Company a Distribution of any asset in kind in lieu of a proportionate Distribution of money being made to other Member(s), and except upon a dissolution and the winding up of the Company, no Member may be compelled to accept a Distribution of any asset in kind.

ARTICLE VII

TRANSFER OF INTERESTS

7.1 Transfer of Interests. Except as permitted in Section 5.3 and this ARTICLE VII, no Member or holder of an Economic Interest shall be entitled to Transfer all or any part of its Membership Interest or Economic Interest except with the prior written consent of all Members, which consent may be given or withheld, conditioned or delayed (as allowed by this Agreement or the Act) as such Members may determine in their sole and absolute discretion. Any attempted Transfer in violation of this Article VII shall be null and void ab initio, and the transferee shall not become a Member or holder of an Economic Interest. If, for any reason, a court refuses to enforce the foregoing provision then, upon any such Transfer of a Membership Interest or part thereof in violation of this ARTICLE VII, the transferee shall only be entitled to become a holder of an Economic Interest to the extent of the Membership Interest attempted or purported to be Transferred to it in violation of this Agreement. After the consumation of any permitted Transfer for all or any part of a Membership Interest, the Membership Interest so Transferred shall continue to be subject to the terms and provisions of this Agreement, and any further Transfers shall be required to comply with the terms and provisions of this Agreement.

7.2 <u>Permitted Transfers</u>. Notwithstanding anything to the contrary contained herein and subject to compliance with Section 7.3 below, a Member may transfer all, but not part, of its Membership Interest or Economic Interest (a) upon the death of such Member or the trustee(s) of such Member, to the respective heirs, executors, administrators, testamentary trustees, legatees or beneficiaries of such Member or trustee(s) of such Member or to a trust created under such trustee(s) will, or (b) to one of its Affiliates or to a trust, foundation or another entity that is created pursuant to the estate planning of a Member or the trustee(s) of a Member. In addition, the Zilkha Members and the Resnick Trust can Transfer its Membership Interest to the other at any time.

7.3 <u>Admission of Transferee</u>. Notwithstanding anything in this Agreement to the contrary, no transferee of the whole or any part of a Membership Interest shall become a substituted Member in the place of its transferor unless all of the following conditions are satisfied:

(a) The Transferring Member and the transferee execute and acknowledge such other instrument or instruments as the other Members may deem necessary or desirable to effectuate the admission, including the written acceptance and adoption by the transferee of all of the terms and conditions of this Agreement as the same may have been amended, and the spouse or registered domestic partner, if any, of the transferee executes and delivers to the Manager a Consent substantially in the form of Exhibit E; and

(b) The transferee pays to the Company a transfer fee which is sufficient, in the reasonable discretion of the other Members, to cover all expenses incurred by the Company in connection with the Transfer and substitution.

7.4 <u>Further Restrictions on Transfers</u>. In addition to any other restrictions found in this Agreement, no Member may Transfer its Membership Interest or any part thereof: (a) without compliance with the Securities Act of 1933, the California Corporate Securities Law of 1968 and any other applicable securities laws, or (b) if the Transfer could result in the termination of the Company for federal or state income tax purposes or the Company not being classified as a partnership for federal or state income tax purposes. In addition, GCE cannot elect to purchase Membership Interests pursuant to Section 7.6(a) or 7.7 unless it is able to cause AGC to pay to the Resnick Trust and the Zilkha Trust at the Closing all principal, accrued interest and other amounts payable under the Land Acquisition Loans.

7.5 Member Transfer Rights.

(a) <u>Resnick Trust and Zilkha Members Transfer Rights</u> Except as permitted by Section 7.2, if the Resnick Trust desires to Transfer all or any part of its Membership Interest, it must first offer such Membership Interests to the Zilkha Trust in accordance with the procedures set forth in Section 7.6(a). If any Zilkha Member desires to transfer all or any part of its Membership Interest (the "<u>Zilkha Transferor</u>"), it must first offer such Membership Interest to the Resnick Trust in accordance with the procedures set forth in Section 7.6(a). If the Resnick Trust or the Zilkha Trust, as applicable, does not exercise its right to purchase such offered Membership Interest, the Resnick Trust or the Zilkha Trust offer such Membership Interest to GCE in accordance with the procedures set forth in Section 7.6(a).

(b) <u>GCE Transfer Rights</u>. Except as permitted by Section 7.2, GCE may not Transfer all or any part of its Membership Interest for the three year period following the date hereof. After the third anniversary of the date hereof, if GCE desires to Transfer all or any part of its Membership Interest, GCE must first offer such Membership Interest to the other Members in accordance with the procedures set forth in Section 7.6(b).

7.6 Member Transfer Procedures.

(a) <u>Resnick Trust and Zilkha Members Transfer Procedures</u>. If the Resnick Trust or a Zilkha Transferor desires to Transfer all or any part of its Membership Interests (the "<u>Transferring Member</u>") to a third party under Section 7.5(a) above, then the Transferring Member shall give written notice to the Zilkha Trust or the Resnick Trust, as applicable (the "<u>Non-Transferring Member</u>"), which notice shall (i) if the Transfer is pursuant to a Bona Fide Offer, set forth the terms of such Bona Fide Offer and the identity of the offeror(s) (the "<u>ROFR Notice</u>") or (ii) if the Transfer is not pursuant to a Bona Fide Offer, set forth the offered Membership Interest and the cash price and other terms upon which it proposes to Transfer such offered Membership Interest (the "<u>ROFO Notice</u>"). The Non-Transferring Member shall have ten (10) calendar days from the date of receipt of the ROFR Notice or ROFO Notice, as applicable, to notify the Transferring Member in writing whether the Non-Transferring Member agrees to purchase all of such offered Membership Interests upon the terms specified in the ROFR Notice or ROFO Notice, as applicable. In the event that the Non-Transferring Member does not elect to purchase the offered Membership Interests, then the Transferring Member shall deliver the ROFR Notice or ROFO Notice, as applicable, to offere Membership Interest upon the terms specified in the ROFR Membership Interest upon the terms specified in the ROFR Membership Interest, then the Transferring Member shall deliver the ROFR Notice or ROFO Notice, as applicable, to GCE and GCE shall have (10) calendar days following its receipt of such notice to elect to purchase all, but not part, of the offered Membership Interest upon the terms specified in the ROFR Membership Interest, then the Transferring Member may, if otherwise permitted by this Agreement, sell all, but not part, of the offered Membership Interest to a third party upon the terms set forth in the ROFR Notice or ROFO Notice, as applicable, within ninety (90) calenda

(b) <u>GCE Transfer Procedures</u>. If GCE desires to Transfer all or any part of its Membership Interest to a third party under Section 7.5(b) above (i) pursuant to a Bona Fide Offer, GCE shall give written notice to the other Members (the "<u>Resnick/Zilkha Members</u>"), setting forth in full the terms of such Bona Fide Offer and the identity of the offeror(s) ("<u>GCE ROFR Notice</u>") or (ii) not pursuant to a Bona Fide Offer, GCE shall give written notice to the Resnick/Zilkha Members, setting forth the offered Membership Interest and the cash price and other terms upon which GCE proposes to Transfer such offered Membership Interest (the "<u>GCE ROFO Notice</u>"). The Resnick/Zilkha Members shall then have the right and option, for a period ending ten (10) calendar days following its receipt of the GCE ROFR Notice or GCE ROFO Notice, as applicable, to elect to purchase all, but not part, of the offered Membership Interest, pro rata in accordance with the ratio of their Percentage Interests, at the purchase price and upon the terms specified in the GCE ROFR Notice or GCE ROFO Notice, as applicable. If all the Resnick/Zilkha Members do not elect to purchase the entire balance of the offered Membership Interest, then the Resnick/Zilkha Members do not elect to purchase the entire balance of the offered Membership Interest, the Resnick/Zilkha Members do not elect to purchase all of the offered Membership Interest, the Resnick/Zilkha Members do not elect to purchase. If the Resnick/Zilkha Members do not elect to purchase all of the offered Membership Interest, the Resnick/Zilkha Members shall not have a right to purchase the offered Membership Interest and GCE may, if otherwise permitted under this Agreement, Transfer all, but not part of, the offered Membership Interest to a third party upon the terms set forth in the GCE ROFR Notice or GCE ROFO Notice, as applicable, within ninety (90) calendar days after the date of the termination of the Resnick/Zilkha Members' rights under this Section 7.6(b); provided, however, that GCE may

7.7 Reciprocal Right to Purchase Membership Interests.

(a) <u>Initiation of Purchase Offer</u>. At any time after the third anniversary of the date hereof, either all of the Zilkha Members and the Resnick Trust collectively or GCE (the "<u>Offering Member(s)</u>") may notify the other (the "<u>Other Member(s)</u>") by written notice (the "<u>Notice</u>") that the Offering Member(s) elects to purchase all, but not less than all, of the Membership Interest owned by the Other Member(s) specifying in the Notice the purchase price at which the Offering Member(s) elects to purchase such Membership Interest for cash (the "<u>Purchase Price</u>").

(b) <u>Option in Favor of Other Member(s)</u>. Immediately upon its receipt of the Notice and for a period of thirty (30) calendar days thereafter, the Other Member(s) shall have an option to purchase all, but not less than all, of the Membership Interest owned by the Offering Member(s) at the Purchase Price. If, within such thirty (30) calendar day period, the Other Member(s) notifies the Offering Member(s) in writing that it elects to exercise its option to purchase all of the Membership Interest owned by the Offering Member(s), as aforesaid, the Offering Member(s) shall be obligated to sell all of said Membership Interest to the Other Member(s), and the Other Member(s) shall be obligated to purchase all of said Membership Interest, at the Purchase Price.

(c) <u>Non-Exercise of Reciprocal Option</u>. If, within such thirty (30) calendar day period, the Other Member(s) does not notify the Offering Member(s) in writing that it elects to exercise its option to purchase all of the Membership Interest owned by the Offering Member(s), as aforesaid, the Offering Member(s) shall purchase all of the Membership Interest owned by the Offering Member(s), and the Other Member(s) shall sell all of said Membership Interest to the Offering Member(s), for the Purchase Price.

(d) <u>No Challenge</u>. Each of the Members agrees to be bound by and to sell its Membership Interest in accordance with this Section 7.7 and specifically waives any rights to challenge or otherwise contest the sufficiency or adequacy of the consideration to be paid for such Membership Interest pursuant to this Section 7.7.

7.8 <u>Consummation of Sale</u>. Unless the parties involved mutually agree otherwise, delivery to the selling Member and the purchasing Members of the Membership Interest to be sold to a Member under this ARTICLE VII or Section 5.3 and payment of the purchase price therefor shall take place at a closing (the "<u>Closing</u>") to be held at the principal office of the Company at 10:00 a.m. within thirty (30) calendar days following the election to purchase or sell pursuant to Section 5.3, 7.6, or 7.7, or, if later, the determination of the applicable purchase price. At the Closing, (a) the transferring Member shall deliver to the purchasing Members a bill of sale and assignment effecting the transfer of the Membership Interest to be sold, in form and substance satisfactory to the purchasing Members, and shall deliver, in addition, any other documents reasonably requested by the purchasing Members are the Resnick Trust or the Zilkha Members, AGC shall pay in full to the Resnick Trust and the Zilkha Trust all principal, accrued interest and other amounts payable under the Land Acquisition Loans. Subject to the foregoing, title to the Membership Interest shall pass to the purchasing Members as of the date of the repurchase event free and clear of any liens or encumbrances.

7.9 Enforcement. The Transfer restrictions contained in this Agreement are of the essence of the ownership of a Membership Interest or an Economic Interest. Upon application to any court of competent jurisdiction, either the Company or any of its Members shall be entitled to a decree against any Person violating or about to violate such restrictions, requiring their specific performance, including those requiring a Member to sell all or part of its Membership Interest to the other Members, requiring a Member to purchase the Membership Interest of other Members or prohibiting a Transfer of all or part of a Membership Interest. No election by a Member to purchase, or not to purchase, all or any part of an Membership Interest shall affect in any manner the rights or remedies of the Company or the Members, whether pursuant to this Agreement, at law or in equity, relating to a breach of this Agreement by the Transferring Member.

ARTICLE VIII

CONSEQUENCES OF MEMBERSHIP TERMINATION EVENTS

8.1 No Dissolution of Company. The occurrence of a Membership Termination Event as to any Member other than the last and only remaining Member shall not dissolve the Company. Upon the occurrence of a Membership Termination Event as to the last and only remaining Member or as otherwise provided by law, the Company shall dissolve unless the personal representative or other successor-in-interest of the last and only remaining Member consents in writing within ninety (90) days of that Membership Termination Event to the continuation of the Company and to the admission of such personal representative or other successor-in-interest, or its designee or nominee, as a Member.

8.2 <u>Admission, Conversion or Purchase</u>. Upon the occurrence of a Membership Termination Event with respect to a Member under circumstances where the Company does not dissolve, the remaining Members shall determine which one of the following shall occur and give written notice thereof to the Member who suffered the Membership Termination Event (the "<u>Former Member</u>"):

(a) the Former Member's personal representative or other successor-in-interest shall be admitted as a Member of the Company in the place and stead of the Former Member to the extent of the Former Member's Membership Interest (the "Former Member's Interest");

(b) the Former Member's Interest shall be converted to a bare Economic Interest, and the Former Member's representative or other successor-in-interest shall become the owner of that Economic Interest; or

(c) Subject to the rights set forth in Article VII, the remaining Members shall purchase the Former Member's Interest for a purchase price determined in accordance with Section 5.3(d).

8.3 <u>Closing of Purchase of Former Member's Interest</u>. The closing of the sale of a Former Member's Interest shall be held no later than thirty (30) days after the determination of the purchase price. At such closing, the Former Member or the Former Member's legal representative shall deliver to the purchasers a bill of sale and assignment effecting the transfer of the Membership Interest to be sold, in form and substance satisfactory to the purchasing Members, and shall deliver, in addition, any other documents reasonably requested by the purchasing Members to effectuate the purposes of this Agreement, and the purchasers shall execute and deliver to the Former Member or the Former Member's legal representative, a promissory note in the amount of the purchase price secured by a pledge of the Membership Interest being purchased. The promissory note shall provide for thirty-six (36) equal monthly payments of principal and interest, with interest computed on a 360 day year and at the then mid-term applicable federal rate provided in the Colosing occurs, but the purchasers shall have the right to prepay in full or in part at any time without penalty. The Former Member or the Former Member's legal representative and the purchasers shall do all things and execute and deliver all papers necessary to consummate the transaction in accordance with the provisions of this Agreement. Title to the Former Member's Interest shall pass to the purchasers as of the date of the Membership Termination Event.

ARTICLE IX

ACCOUNTING, RECORDS, REPORTING BY MEMBERS

9.1 Books and Records. The books and records of the Company shall be kept, and the financial position and the results of its operations recorded, in accordance with generally accepted accounting principles. The books and records of the Company shall reflect all the Company transactions and shall be appropriate and adequate for the Company's business. Board Members, and, as long as a Member's Percentage Interest is at least ten percent (10%), then such Member and its duly authorized representative, shall have complete access to all such books and records at any time.

9.2 Bank Accounts; Invested Funds. All funds of the Company shall be deposited in such account or accounts of the Company as may be determined by the Board and withdrawals may be made upon checks signed by such persons and in such manner as the Board may determine. Temporary surplus funds of the Company may be invested in commercial paper, time deposits, short-term government obligations or other investments determined by the Board.

9.3 Tax Matters for the Company Handled by Board and Tax Matters Partner.

(a) <u>Tax Elections</u>. The Board shall from time to time cause the Company to make such tax elections as it deems to be in the best interests of the Company and the Members.

(b) <u>Designation of Tax Matters Partner</u>. GCE shall be the Tax Matters Partner. If no person shall be serving as Tax Matters Partner, the Person meeting the requirements for a tax matters partner under Code Section 6231(a)(7) and designated by vote of Management Committee shall become the Tax Matters Partner. The Tax Matters Partner may resign upon thirty (30) days' prior written notice to the other Members.

(c) <u>Powers</u>. The Tax Matters Partner has all of the powers and authority of a tax matters partner under the Code. The Tax Matters Partner shall represent the Company in connection with all administrative and/or judicial proceedings instituted by the Internal Revenue Service or any taxing authority involving any tax return of the Company. This representation shall be at the Company's expense. The Tax Matters Partner may expend the Company's funds for professional services and costs associated with any administrative and/or judicial proceedings instituted by the Internal Revenue Service or any taxing authority involving any tax return of the Company. The Tax Matters Partner shall provide to the Members prompt notice of any communication to or from or agreements with a federal, state or local taxing authority regarding any tax return of the Company (including a summary of the communication).

9.4 <u>Accounting Matters</u>. All decisions as to accounting matters shall be made by the Board; provided, however, that the Board may at any time request the Manager to provide its recommendation as to any accounting matter. At any time or upon request of the Manager, the Board shall review and approve the accounting procedures that will be implemented by the Manager.

9.5 <u>Confidentiality</u>. All books, records, financial statements, tax returns, budgets, business plans and projections of the Company, all other information concerning the business, affairs and properties of the Company and all of the terms and provisions of this Agreement shall be held in confidence by the Manager, Board Members and the Members and their respective Affiliates, subject to any obligation to comply with (a) any applicable law, (b) any rule or regulation of any legal authority or securities exchange or (c) any subpoena or other legal process to make information available to the Persons entitled thereto. Such confidentiality shall be maintained to the same degree as each Manager. Board Member and Member maintains its own confidential information and shall be maintained until such time, if any, as any such confidential information either is, or becomes, published or a matter of public knowledge.

ARTICLE X

DISSOLUTION AND WINDING UP

10.1 Dissolution. The Company shall be dissolved, its assets disposed of and its affairs wound up upon (and only upon) the first to occur of the following:

(a) the expiration of the term of the Company specified in the Certificate of Formation or any other event of dissolution specified in the Certificate of

Formation;

(b) the unanimous vote of the Members;

(c) the vote of the Preferred Members following their receipt of a written notice from GCE in accordance with Section 5.3(e) stating GCE's desire to dissolve the Company;

(d) the occurrence of a Membership Termination Event as to the last and only remaining Member if the Board and that Member's personal representative or other successor-in-interest fail to consent to the continuation of the Company in accordance with Section 8.1 within ninety (90) days after the occurrence of that event;

(e) the Company's Bankruptcy;

(f) the occurrence of an event which makes it unlawful for the business of the Company to be continued; or

(g) as otherwise required by law.

10.2 <u>Date of Dissolution</u>. Dissolution of the Company shall be effective on the day on which the event occurs giving rise to the dissolution, but the Company shall not terminate until the assets of the Company have been liquidated and distributed as provided herein. Notwithstanding the dissolution of the Company, prior to the termination of the Company the business of the Company and the rights and obligations of the Members, as such, shall continue to be governed by this Agreement.

10.3 Winding Up. Upon the occurrence of any event specified in Section 10.1, the Company shall continue solely for the purpose of winding up its affairs in an orderly manner, liquidating its assets and satisfying the claims of its creditors. The Board shall be responsible for overseeing the winding up and liquidation of the Company, shall take full account of the liabilities and assets of the Company, shall cause its assets either to be sold or distributed, as they may determine, and shall cause the proceeds therefrom, to the extent sufficient, to be applied and distributed as provided in Section 10.5; provided, however, that at the request of the Resnick Trust or the Zilkha Members, any real property owned directly or indirectly by the Company shall be distributed in kind to such Members subject to Section 10.4; provided, however, that if such distribution in kind would result in the Resnick Trust or the Zilkha Members receiving greater distributions than they would otherwise be entitled to under Section 10.5, such Members shall refund such excess distributions in cash to the Company. The Persons winding up the affairs of the Company shall give written notice of the commencement of winding up by mail to all known creditors and claimants whose addresses appear on the records of the Company.

10.4 <u>Distributions in Kind</u>. Any non-cash asset distributed to one or more Members shall first be valued at its Fair Market Value to determine the Net Profit or Net Loss that would have resulted if that asset had been sold for that value, the Net Profit or Net Loss shall then be allocated pursuant to ARTICLE VI, and the Members' Capital Accounts shall be adjusted to reflect those allocations. The amount distributed and charged to the Capital Account of each Member receiving an interest in the distributed asset shall be the Fair Market Value of the interest (net of any liability secured by the asset that the Member assumes or takes subject to). The Fair Market Value of that asset shall be determined by the Board.

10.5 Order of Payment of Proceeds Upon Dissolution.

(a) <u>Liquidating Distributions</u>. After determining that all known debts and liabilities of the Company, including, without limitation, the Land Acquisition Loans and other debts and liabilities to Members who are creditors of the Company, have been paid or adequately provided for, the remaining assets shall promptly be distributed to the Members in accordance with their positive Capital Account balances, after taking into account income and loss allocations for the Company's taxable year during which the liquidation occurs.

(b) <u>No Liability</u>. No Member shall have any liability to the Company, any Member or any creditor of the Company on account of any deficit balance in its Capital Account.

10.6 Limitations on Payments Made in Dissolution. Except as otherwise specifically provided in this Agreement, each Member shall be entitled to look only to the assets of the Company for the return of that Member's positive Capital Account balance and shall have no recourse for its Capital Contributions and/or share of Net Profits (upon dissolution or otherwise) against the Manager or any other Member.

10.7 Certificate of Cancellation. Upon completion of the winding up of the Company's affairs, the Board shall cause a Certificate of Cancellation to be filed with the Delaware Secretary of State.

10.8 Compensation for Services. The Persons winding up the affairs of the Company shall be entitled to reasonable compensation from the Company for their services.

ARTICLE XI

INDEMNIFICATION

11.1 Indemnification. The Company shall indemnify and hold harmless each of the Members and the Manager, and each of their respective officers, directors, shareholders, partners, members, trustees, beneficiaries, employees, agents, heirs, assigns, successors-in-interest and Affiliates, (collectively, "Indemnified Persons") from and against any and all losses, damages, liabilities and expenses, (including costs and reasonable attorneys' fees), judgments, fines, settlements and other amounts (collectively "Liabilities") reasonably incurred by any such Indemnified Person in connection with the defense or disposition of any civil, administrative or investigative action, suit or other proceeding, whether and whether threatened, pending or completed (collectively a "Proceeding"), in which any such Indemnified Person may be involved or with which any such Indemnified Person may be threatened, with respect to or arising out of any act performed by the Indemnified Person or any omission or failure to act if the performance of the act or the omission or failure was done in good faith and within the scope of the authority conferred upon the Indemnified Person by this Agreement or by the Act, except for acts of fraud, deceit, reckless or intentional misconduct, gross negligence, embezzlement, breach of a fiduciary duty, knowing violations of law, acts which constituted breaches of this Agreement (whether committed knowingly, negligently or otherwise) or acts from which such Indemnified Person derived an improper personal benefit. The Company's indemnification obligations hereunder shall apply not only with respect to any Proceeding brought by the Company or a Member but also with respect to any Proceeding brought by a third party. As a condition to the indemnification and other rights granted to an Indemnified Person pursuant to this Article, however, that Indemnified Person may not settle any action, suit or proceeding without the written consent of the Board.

11.2 <u>Contract Right; Expenses</u>. The right to indemnification conferred in this ARTICLE XI shall be a contract right and shall include the right to require the Company to advance the expenses incurred by the Indemnified Person in defending any such Proceeding in advance of its final disposition; provided, however, that, if the Act so requires, the payment of such expenses in advance of the final disposition of a Proceeding shall be made only upon receipt by the Company of an undertaking, by or on behalf of the indemnified Person, to repay all amounts so advanced if it shall ultimately be determined that such Person is not entitled to be indemnified under this ARTICLE XI or otherwise.

11.3 Indemnification of Officers and Employees. The Company may, to the extent authorized from time to time by the Board, grant rights to indemnification and to advancement of expenses to any officer, employee or agent of the Company to the fullest extent of the provisions of this ARTICLE XI with respect to the indemnification and advancement of expenses of Members and the Manager.

11.4 Insurance. The Company may purchase and maintain insurance on behalf of any Person who is or was an agent of the Company against any liability asserted against that Person and incurred by that Person in any such capacity or arising out of that Person's status as an agent, whether or not the Company would have the power to indemnify that Person against liability under the provisions of Section 11.1 or under applicable law.

ARTICLE XII

INVESTMENT REPRESENTATIONS

Each Member represents and warrants to the other Members and the Company as follows:

12.1 <u>Preexisting Relationship or Experience</u>. (a) The Member has a preexisting personal or business relationship with the Company or its Manager, officers or control persons or (b) by reason of the Member's business or financial experience, or by reason of the business or financial experience of the Member's financial advisor who is unaffiliated with and who is not compensated, directly or indirectly, by the Company or any Affiliate or selling agent of the Company, the Member is capable of evaluating the risks and merits of an investment in its Membership Interest and of protecting the Member's own interests in connection with the investment.

12.2 Access to Information. The Member has had an opportunity to review all documents, records and books pertaining to this investment and has been given the opportunity to consult with counsel of his or her choice with respect to all aspects of this investment and the Company's proposed business activities. Such Member has personally met with the Manager and has been provided with such information as may have been requested and has at all times been given the opportunity to obtain additional information necessary to verify the accuracy of the information received and the opportunity to ask questions of and receive answers from the Manager concerning the terms and conditions of the investment and the nature and prospects of the Company's business.

12.3 Economic Risk. The Member is financially able to bear the economic risk of an investment in its Membership Interest, including the total loss thereof.

12.4 Investment Intent. The Member is acquiring its Membership Interest for investment purposes and for the Member's own account only and not with a view to, or for sale in connection with, any distribution of all or any part of its Membership Interest. Except for the partners or members of the Member, no other Person will have any direct or indirect beneficial interest in, or right to, its Membership Interest.

12.5 <u>Consultation with Attorney</u>. The Member has been advised to consult with its own attorney regarding all legal and tax matters concerning an investment in its Membership Interest and has done so to the extent it considers necessary.

12.6 <u>Purpose of Entity</u>. If the Member is a corporation, partnership, limited liability company, trust or other entity, it was not organized for the specific purpose of acquiring its Membership Interest.

12.7 No Advertising. The Member has not seen, received or been solicited by any leaflet, public promotional meeting, newspaper or magazine article or advertisement, radio or television advertisement or any other form of advertising or general solicitation with respect to the sale of its Membership Interest.

12.8 <u>Membership Interest is Restricted Security</u>. The Member understands that its Membership Interest is a "restricted security" under the Securities Act of 1933 in that the Membership Interest will be acquired from the Company in a transaction not involving a public offering, that its Membership Interest may be resold without registration under the Securities Act of 1933 only in certain limited circumstances and that otherwise its Membership Interest must be held indefinitely.

12.9 No Registration of Membership Interest. The Member acknowledges that its Membership Interest has not been registered under the Securities Act of 1933 or qualified under any state securities law in reliance, in part, upon its representations, warranties and agreements herein.

ARTICLE XIII

MISCELLANEOUS

13.1 Amendments. No amendment to this Agreement may be made without the consent of all Members. All amendments to this Agreement must be in writing.

13.2 Offset Privilege. Any monetary obligation owing from the Company to any Member or Manager may be offset by the Company against any monetary obligation then owing from that Member or Manager to the Company.

13.3 Arbitration.

(a) <u>General</u>. In the event of any dispute, claim or controversy among the parties (other than a claim for equitable relief) arising out of or relating to this Agreement or the Certificate of Formation, whether in contract, tort, equity or otherwise, and whether relating to the meaning, interpretation, effect, validity, performance or enforcement of this Agreement or the Certificate of Formation, such dispute, claim or controversy shall be resolved by and through an arbitration proceeding to be conducted under the auspices and the commercial arbitration rules of the American Arbitration Association (or any like organization successor thereto) at Los Angeles, California. The arbitrability of the dispute, claim or controversy shall likewise be determined in the arbitration Association. Both the foregoing agreement of the parties to arbitrate any and all such disputes, claims and controversies, and the results, determinations, findings, judgments and/or awards rendered through any such arbitration shall be final and binding on the parties and may be specifically enforced by legal proceedings in any court of competent jurisdiction.

(b) Governing Law. The arbitrator(s) shall follow any applicable federal law and Delaware state law (with respect to all matters of substantive law) in rendering an award.

(c) <u>Costs of Arbitration</u>. The cost of the arbitration proceeding and any proceeding in court to confirm or to vacate any arbitration award, as applicable (including, without limitation, each party's attorneys' fees and costs), shall be borne by the unsuccessful party or, at the discretion of the arbitrator(s), may be prorated between the parties in such proportion as the arbitrator(s) determines to be equitable and shall be awarded as part of the arbitrator's award.

13.4 <u>Remedies Cumulative</u>. Except as otherwise provided herein, the remedies under this Agreement are cumulative and shall not exclude any other remedies to which any Person may be lawfully entitled.

13.5 <u>Notices</u>. Any notice to be given to the Company or any Member in connection with this Agreement must be in writing, signed by the sender, and will be deemed to have been given and received when delivered to the address specified by the party to receive the notice by courier or other means of personal service, when received if sent by facsimile, portable document format or other form of electronic transmission (as defined in the Act) or three (3) days after deposit of the notice by first class mail, postage prepaid, or certified mail, return receipt requested. Any such notice must be given to the Company at its principal place of business, and to any Member at the address specified in <u>Exhibit A</u>. Any party may, at any time by giving five (5) days' prior written notice to the other parties, designate any other address as the new address to which notice must be given. In the case of notice by facsimile, portable document format or other form of electronic transmission, a copy thereof shall be personally delivered or sent by registered or certified mail, in the manner specified above, within three (3) Business Days thereafter.

13.6 <u>Attorney's Fees</u>. In the event that any dispute between the Company, the Manager and/or the Members should result in litigation or arbitration, the prevailing party in that dispute shall be entitled to recover from the other party all reasonable fees, costs and expenses of enforcing any right of the prevailing party, including without limitation, reasonable attorneys' fees and expenses, subject, however to the provisions of Section 13.3(c).

13.7 Jurisdiction. Each Member and the Manager consents to the exclusive jurisdiction of the state and federal courts sitting in Los Angeles, California in any action on a claim arising out of, under or in connection with this Agreement or the transactions contemplated by this Agreement, provided such claim is not required to be arbitrated pursuant to Section 13.3. Each Member and the Manager further agrees that personal jurisdiction over it may be effected by service of process by registered or certified mail addressed as provided in Section 13.5 and that when so made shall be as if served upon it personally.

13.8 <u>Complete Agreement</u>. This Agreement and the Certificate of Formation constitute the complete and exclusive statement of agreement among the Members with respect to their respective subject matters and supersede all prior written and oral agreements or statements by and among the Members. No representation, statement, condition or warranty not contained in this Agreement or the Certificate of Formation shall be binding on the Members or have any force or effect whatsoever. To the extent that any provision of the Certificate of Formation conflicts with any provision of this Agreement, the Certificate of Formation shall control.

13.9 Binding Effect. Subject to the provisions of this Agreement relating to Transferability, this Agreement shall be binding upon and inure to the benefit of the Members and their respective successors and assigns.

13.10 Section Headings. All Section headings are inserted only for convenience of reference and are not to be considered in the interpretation or construction of any provision of this Agreement.

13.11 Interpretation. In the event any claim is made by any Member relating to any conflict, omission or ambiguity in this Agreement, no presumption or burden of proof or persuasion shall be implied by virtue of the fact that this Agreement was prepared by or at the request of a particular Member or that Member's counsel.

13.12 Severability. If any provision of this Agreement or the application of that provision to any person or circumstance shall be held invalid, the remainder of this Agreement or the application of that provision to persons or circumstances other than those to which it is held invalid shall not be affected.

13.13 <u>Multiple Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument.

IN WITNESS WHEREOF, all of the Members and the Manager of GCE MEXICO I, LLC, a Delaware limited liability company, have executed this Agreement, effective as of the date first written above.

"Members"

GLOBAL CLEAN ENERGY HOLDINGS, INC.

By:

Richard Palmer, President and CEO

STEWART A. RESNICK, AS TRUSTEE OF THE STEWART AND LYNDA RESNICK REVOCABLE TRUST, DATED DECEMBER 27, 1988, AS AMENDED

LYNDA RAE RESNICK, AS TRUSTEE OF THE STEWART AND LYNDA RESNICK REVOCABLE TRUST, DATED DECEMBER 27, 1988, AS AMENDED

SELIM ZILKHA, AS TRUSTEE OF THE SELIM K. ZILKHA TRUST

MICHAEL ZILKHA, AS TRUSTEE OF THE DMZ 2000 TRUST

MICHAEL ZILKHA, AS TRUSTEE OF THE LLZ 2000 TRUST

NADIA Z. WELLISZ, AS TRUSTEE OF THE JW 2000 TRUST

NADIA Z. WELLISZ, AS TRUSTEE OF THE DW 2000 TRUST

"Manager"

GLOBAL CLEAN ENERGY HOLDINGS, INC.

By:

Richard Palmer, President and CEO

EXHIBIT A

CAPITAL CONTRIBUTIONS, ADDRESSES AND PERCENTAGE INTERESTS

OF MEMBERS AS OF

April 23, 2008

0

Preferred Member's Name	Preferred Member's Address	Preferred Member's Capital Contribution Obligation	Preferred Units	Preferred Percentage
Stewart Resnick and Lynda Resnick as trustees of the Stewart and Lynda Resnick Revocable Trust dated December 27, 1988 as amended	11444 West Olympic Boulevard, 10 th Floor Los Angeles, CA 90064	\$1,116,312	500	50%
Selim Zilkha, as trustee of the Selim K. Zilkha Trust	750 Lausanne Road, Los Angeles, CA 90077	\$1,116,312	500	50%
Common Members:				
Common Member's Name	Common Member's Address	Common Member's Capital Contribution Obligation	Common Units	Percentage Interest
Global Clean Energy Holdings, Inc. 11444 W. Olympic Blvd. 10 th Floor Los Angeles, CA 90064	6033 W. Century Blvd. Suite 1090 Los Angeles, CA 90045	\$ 0	500	50%
Stewart Resnick and Lynda Resnick as trustees of the Stewart and Lynda Resnick Revocable Trust dated December 27, 1988 as amended	11444 West Olympic Boulevard, 10 th Floor Los Angeles, CA 90064	\$ 0	250	25%
Michael Zilkha, as trustee of the LLZ 2000 Trust,	750 Lausanne Road, Los Angeles, CA 90077	\$ 0	62.5	6.25%
Nadia Z. Wellisz, as trustee of the JW 2000 Trust	750 Lausanne Road, Los Angeles, CA 90077	\$ 0	62.5	6.25%
Nadia Z. Wellisz, as trustee of the DW 2000 Trust	750 Lausanne Road, Los Angeles, CA 90077	\$ 0	62.5	6.25%
Michael Zilkha, as trustee of the DMZ 2000 Trust	750 Lausanne Road, Los Angeles, CA 90077	\$ 0	62.5	6.25%
	44			

EXHIBIT B BUDGET

EXHIBIT C

FORM OF PROMISSORY NOTE

EXHIBIT D

FORM OF MORTGAGE

EXHIBIT E

FORM OF CONSENT

The undersigned acknowledges as follows:

a. The undersigned has read the foregoing Limited Liability Company Agreement (the "Agreement"), understands the contents of the Agreement, and is aware that by the provisions of the Agreement, the undersigned's spouse or registered domestic partner agrees to certain restrictions and requirements relating to the sale or other transfer of his/her Membership Interest in the Company, including the undersigned's community property or other ownership interest therein (if any) and any interest of the undersigned pursuant to the non-marital laws of contract or palimony. THE UNDERSIGNED HAS HAD THE RIGHT TO CONSULT WITH COUNSEL OF HIS OR HER CHOOSING IN CONNECTION WITH THIS CONSENT AND HE OR SHE HAS HAD AMPLE OPPORTUNITY TO DO SO. IF THE UNDERSIGNED HAS NOT CONSULTED WITH COUNSEL IN CONNECTION HEREWITH, THE UNDERSIGNED HAS KNOWINGLY AND WILLINGLY ELECTED NOT TO DO SO.

b. The undersigned (i) consents to any such restrictions and requirements, (ii) agrees to be bound by the Agreement and join therein to the extent (if any) that his or her agreement and/or joinder may be necessary, (iii) agrees that the undersigned's spouse or registered domestic partner shall have the sole and exclusive management power with respect to the Membership Interest subject to the Agreement, (iii) agrees that the undersigned will not effect or attempt to effect any sale or other transfer of such Membership Interest, or of any interest therein, (iv) agrees that the undersigned will take no action at any time to hinder the operation of the Agreement on such Membership Interest, including the undersigned's community property or other ownership interest therein (if any) and any interest of the undersigned pursuant to the non-marital laws of contract or palimony, and (v) agrees that the undersigned's spouse or registered domestic partner may join in any future amendment or modification without any further signature, acknowledgement, agreement, or consent on the part of the undersigned.

c. Should the spouse or the registered domestic partner of the undersigned die and bequeath to the undersigned any interest in the Membership Interest covered by the Agreement in such a manner that no probate is required with respect thereto, or should the applicable probate laws relating to the community property or other ownership interest (if any) of the undersigned or any interest of the undersigned pursuant to the non-marital laws of contract or palimony in such Membership Interest provide, upon the death of the undersigned's spouse or registered domestic partner, as the case may be, that the undersigned is entitled to a portion of the Membership Interest without such portion being subject to probate, or should the undersigned acquire any interest in the Membership Interest during the undersigned's spouse's or registered domestic partner's life by reason of any agreement, court order, judgment or decree, or for any other reason whatsoever, then the undersigned further agrees that the undersigned shall perform all of the obligations of the undersigned's spouse or registered domestic partner, as the case may be, imposed thereunder. d. The undersigned shall perform any further acts and execute and deliver any further documents or procure any court orders which may be reasonably necessary to carry out the provisions of this Consent.

Name:	,
Spouse or Registered Domestic	
Partner of	

SERVICE AGREEMENT

This Service Agreement (hereinafter the "Agreement") is entered into this 15th day of October 2007 between Lodemo y Asociados S.C.P., a Mexican Corporation with its primary place of business located at Calle 18, # 201-B x 23 y 25, Colonias Garcia Gineres, C.P. 97070, Merida, Yucatan, Mexico (hereinafter "Lodemo"); and Medical Discoveries, Inc., a Utah Corporation dba Global Clean Energy Holdings, with its primary place of business located at 6033 W. Century Blvd, Suite 1090, Los Angeles, CA 90045, USA (hereinafter "Global"). The terms of this Agreement shall be binding upon the parties.

WITNESSETH:

WHEREAS, Global intends to secure land and establish and operate farming operations within the Republic of Mexico (hereinafter "Mexico") for the purpose of growing Jatropha Curcas (hereinafter "Jatropha"), a non-edible agricultural product; and

WHEREAS, Global intends to harvest Jatropha seeds from its farming operations in Mexico and extract oil from the seeds for the purpose of selling the oil inside and outside Mexico as an energy source and biodiesel feedstock; and

WHEREAS, Global intends to construct and operate seed oil extraction facilities, and, in connection therewith, desires to set up logistics and transportation systems to transport production feedstocks, supplies, seed oil, biodiesel, and other end products; and

WHEREAS, Global intends to acquire professional services from Lodemo to support its activities and objectives as described in this Agreement; and

WHEREAS, the parties desire to set forth their specific understanding of their respective responsibilities and obligations associated with delivery of the services to be provided by Lodemo hereunder.

NOW, THEREFORE, in consideration of the mutual promises and covenants contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Global and Lodemo agree as follows:

1. DEFINITIONS

For purposes of this Agreement, the following terms and phrases have the following definitions:

"Facilities" means all physical assets purchased or intended for purchase by Global necessary to deliver the products and services defined in this Agreement; including structures, seeds, plants, fertilizer, planting materials and supplies, tools, machinery, equipment, and vehicles. For purposes of this Agreement, the Facilities shall not include any land.

Page 1 of 23

"Project Operations & Capital Expenditure Budget" means a detailed and itemized estimate of projected yearly expenditures necessary for construction and operation of the Project based on anticipated progress and activity by the parties, as revised and amended from time to time in accordance with the terms hereof.

"Facility Manager" means Lodemo's management representative with day-to-day operating responsibility for the Project.

"Farm" means the land owned or leased by Global in Mexico and all equipment necessary for the propagation, cultivation, and harvesting of Jatropha on such land.

"Logistics" means those assets and services that are necessary for the transportation of labor, materials, products, by-product or waste to or from the operations of the Project, Farms and Facilities.

"Oil Extraction Facility" means the assets and services required for the removal of oil from the Jatropha seeds produced from the Farms and for the storage of the seeds or oil. Global shall acquire or construct, and thereafter own, the Oil Extraction Facility. The construction or acquisition of the Oil Extraction Facility is not covered by this Agreement. The operation of the Oil Extraction Facility, however, is included in this Agreement as part of the Project, and the operation of the Oil Extraction Facility shall be included as part of the Services.

"Product" means the amount of Jatropha seed oil and the amount of biomass byproduct produced as part of the Project.

"Project" means establishment, development, and operation of Global's business to grow Jatropha in Mexico, to extract the oil from the Jatropha seeds, and the deliver of such oil to a buyer, including the purchase or lease of land in the name of Global, the establishment and operation of one or more Jatropha nurseries, the clearing, planting and cultivation of the Jatropha fields, the harvesting of the Jatropha seeds, the operation of Global's oil extraction facilities, and the logistics associated with the foregoing. The Project does not include the Joint Venture or the Biodiesel Refinery, which activities will be subject to one or more separate agreements between the parties hereto or their respective affiliates.

"Prudent Operating Practice" means any of the practices, methods and acts which, in the exercise of reasonable judgment in the light of the facts known at the time that a decision was made, could reasonably have been expected to accomplish the desired result at the lower reasonable cost, consistent with licensing and regulatory considerations, environmental considerations, reliability, safety and expedition.

Page 2 of 23

"Construction & Operating Plan" means a forecast established by the parties for each year describing on a calendar month basis the requested level of production of Product from the Project for the following calendar year.

"Scheduled Output" means the designed capacity of the Project on a month to month basis as set forth in the Construction & Operating Plan.

"Services" means the services to be provided by Lodemo under this Agreement required to accomplish the goals of the Project.

"Service Standards" has the meaning set forth in Section 3.2(a).

"Utilities" means, collectively, those utilities which are utilized or required in connection with the Project, including electricity, fuel, water, wastewater, and temporary portable lavatories, applied or consumed in the operation of the Project and the supply or performance of Services.

"Joint Venture" means business arrangement the parties intend to establish for the purpose of designing, constructing, and operating a biodiesel manufacturing facility to be located in Yucatan, Mexico (hereinafter the "Biodiesel Refinery"). The parties currently anticipate that any Joint Venture established by them shall be equally owned by them and will be financed, constructed, and operated on terms to be negotiated by them.

2. <u>SCOPE OF SERVICES</u>

2.1. General Scope of Services

Subject to Global's payment obligations as more fully set forth in Section 6, Lodemo's Services will include the following:

- 2.1.1. <u>Project Management</u>. Lodemo will provide Project Management services, which services, subject to the rights of the Global Project Manager to the extent granted herein, shall consist of implementing the Construction & Operating Plan, including the overall management and supervision of all aspects of the Project and the responsibility for ensuring the proper and timely performance of all Services to Global. Project Management provided by Lodemo will be provided by senior Lodemo staff and include participation in meetings with Global, planning, reporting, scheduling, budgeting, and day-to-day direction of all staff and activities.
- 2.1.2. <u>Acquisition of Land Parcels</u>. In cooperation with and on Global's behalf, Lodemo will identify potential land parcels suitable for nursery and Jatropha farming operations. Lodemo will negotiate with land owners the terms for the long-term use of the land, either by sale or by lease, for the purpose of Jatropha farming. Lodemo will assist Global by all necessary means to negotiate and consummate land sales and/or lease agreements. Notwithstanding the foregoing, Lodemo shall not have the right to enter into any land agreements (lease or purchase) on behalf of Global, and Lodemo shall not have the right to bind or obligate Global in any such land agreements. Nothing set forth in this Agreement obligates Global to enter into any land sales or lease Agreement, and Global shall have the sole right to determine whether to acquire land and the terms of such acquisition. The land contracts shall be entered into by an affiliate of Global, organized in Mexico.

Page 3 of 23

- 2.1.3. <u>Clearing of Land</u>. Lodemo will prepare all nursery and farming land parcels as required to allow for efficient planting, irrigation, drainage, and propagation of Jatropha. Lodemo will clear trees, shrubs, structures, large rocks, and other foreign objects from the land as required.
- 2.1.4. <u>Nursery Construction and Operation</u>. Lodemo will set up and construct a Jatropha plant nursery at a site location approved by Global. The nursery will be for the staging operations and storage and care of Jatropha seeds, seedlings, stem cuttings, and other materials and supplies necessary for nursery operations. Global will be responsible for providing all Jatropha seeds, seedlings, stem cuttings, and other materials and supplies necessary for nursery operations. Lodemo will provide staff for ongoing daily operations of the nursery.
- 2.1.5. <u>Planting and Cultivation of Jatropha</u>. Lodemo will plant and cultivate Jatropha on the land owned or leased by Global. Planting will include transfer of seeds, seedlings, and/or established stem cuttings supplied by and through the nursery. Lodemo will provide high quality long-term care of the Jatropha to include irrigation, removal of weeds, and seasonal pruning.
- 2.1.6. <u>Jatropha Seed Harvesting and Transportation</u>. Upon proper plant maturity, Lodemo will harvest and clean all mature Jatropha seeds from all of Global's planted land parcels described above in this Agreement on a periodic basis as required. All Global Jatropha seeds harvested by Lodemo will be cleaned (removal of seed casings, stems, and leaves) and transferred by Lodemo to the Oil Extraction Facility or to another Yucatan oil extraction facility designated by Global.
- 2.1.7. <u>Operation of Oil Extraction Facilities</u>. Lodemo will assist Global in identifying suitable locations for one or more Oil Extraction Facilities. Lodemo will provide all operations and logistics services for the management and operations of a Global Oil Extraction Facility. The Oil Extraction Facility will be designed, constructed, and owned by Global. Lodemo will be responsible for all operations and site logistics, including transfer and disposal of seed meal and transfer of seed oil to buyers and local Global biodiesel manufacturing facilities.

2.2. Other Services

The parties understand that in the process of executing its Services, Global and Lodemo may from time to time hire or cause to be hired contractors and subcontractors to complete certain tasks. If during the process of reviewing contractor and subcontractor proposals, Lodemo determines it can and desires to provide the same services for an equal or lower price and at an equal or better quality; Global will provide Lodemo the opportunity to enter into agreement and perform said services for its proposed price and under similar terms.

2.3. Construction of Facilities

Page 4 of 23

2.3.1. <u>Global Responsibilities</u>. Global will fund the Project's costs and the fees and expenses set forth in this Agreement in the manner as specified in the Project Operations & Capital Expenditure Budget. Global shall also provide the necessary technology, including plant and soil science expertise, for the setup and construction of one or more nurseries and for Farm operations as necessary for the bulk production of Jatropha seeds. Global will provide all necessary funds, technology, engineering, procurement, and construction for the Oil Extraction Facilities.

2.4. Operation of Facilities

2.4.1. <u>Services on Behalf of Global</u>. Subject to (a) Global's right to exercise managerial control in accordance with default provisions as outlined in this Agreement, (b) Global's other rights hereunder, and (c) the other provisions of this Agreement, Lodemo will provide its Services on behalf of and for the account of Global, will have full authority and responsibility for executing its Services, and will perform its Services according to expectations and at such times as directed by Global.

2.5. Service Standards

- 2.5.1. <u>Standard of Quality</u>. Lodemo shall adhere to commercially reasonable standards in performing the Services in accordance with Global's direction and expectations as communicated in writing and through Project meetings with Global from time to time and shall perform or cause to be performed the Services in all material respects in accordance with (i) Prudent Operating Practice, (ii) Lodemo's standard practices for such services consistent with the scope and quality of similar services provided by Lodemo for itself in its other business activities, and (iii) all reasonable instructions of Global (collectively, the "Service Standard").
- 2.5.2. <u>Compliance/No Violations</u>. Lodemo shall perform its Services in accordance with applicable laws, rules, regulations, permits and licenses and orders of governmental authorities as in effect from time to time (collectively referred to as "Applicable Law"). Lodemo, in rendering its Services, shall have no obligation to take any action or otherwise to perform to the extent it reasonably believes such action or performance is or may be in violation of any Applicable Law or may involve any material risk to the Project, Facilities or Farm, or any part thereof, or any persons or property.

2.6. Personnel

2.6.1. Formation of Labor Management Company. The parties understand and agree that Lodemo may create a new Mexican corporation (herein referred to as "NEW Co.") to perform the functions of hiring in its name and managing personnel required for executing its Services under this agreement. In its function, NEW Co. will be responsible for all labor-related management functions, including all payroll operations and payment of social security and health and welfare contribution obligations. Lodemo agrees that, notwithstanding the formation of New Co, Lodemo will be responsible for the supervising and managing NEW Co. and that Lodemo shall be responsible to Global under the terms of this Agreement for the actions or omissions of NEW Co. Even though personnel required for execution of the Services may be employees of NEW Co., such personnel will remain under the sole and exclusive supervision and control of Lodemo. All Lodemo and NEW Co. employees will be properly classified, qualified, trained and supervised by Lodemo for the execution of the Services.

Page 5 of 23

- 2.6.2. Use of Employment Agencies. In order to hire appropriate personnel and fulfill the staffing obligations under this Agreement, Lodemo is authorized to hire personnel through an employment agency or labor outsourcing company. Additional fees required to hire personnel through an employment agency are considered Allowable Costs under the terms of this Agreement. All employees acquired through any employment agency will remain under the sole and exclusive responsibility, supervision, and control of Lodemo.
- 2.6.3. <u>Right to Remove Personnel</u>. If Global determines in good faith that the continued assignment of any person, subcontractor, or employment agency performing or providing Services is not in accordance with the requirements and standards set forth in this Agreement, Global may deliver a notice to Lodemo stating its complaint and requesting removal and/or replacement of such person(s), subcontractor, or employment agency. Promptly after its receipt of such complaint and request by Global, Lodemo will investigate the matters stated in the complaint and request and discuss its own findings with Global. If Global still, in good faith, requests removal and/or replacement of such person(s), subcontractor, or employment agency, Lodemo will promptly remove and/or replace that person(s), subcontractor, or employment agency with one of suitable ability and qualifications reasonably acceptable to Global. In such cases, all costs incurred for the removal and replacement of such person(s), subcontractor, or employment agency as well as any damages or repairs caused, will be considered as Non-Allowable Costs.

2.7. Utilities

Lodemo will arrange for all necessary temporary and permanent Utilities to be provided to the nursery, any Farm, and any Oil Extraction Facility. In providing such Utilities, Lodemo will use such Utilities prudently and provide such services as economically as possible for such operation in connection with the Services.

2.8. Raw Materials, Supplies and Packaging Material.

Page 6 of 23

Lodemo may acquire on behalf of Global all minor raw materials and supplies reasonably necessary for the Project, including maintenance, repair and operating supplies, spare parts, packaging materials, and utilities services and facilities required by Global and for execution of the Services. Such reasonably necessary minor raw materials and supplies are subject to periodic review by Global, but shall be considered as Allowable Costs.

2.9. Maintenance and Repairs

In the course of conducting operations and performing its Services, Lodemo may provide maintenance and repairs which may or may not involve capital expenditures, in accordance with Global's quality standards and standard Lodemo practice. If, in the reasonable opinion of Lodemo, capital expenditures are required with respect to proper operation and maintenance of the Project, Lodemo may make such capital expenditures directly and consider them as Allowable Costs provided that such capital expenditures are planned and included in the Project Operations & Capital Expenditure Budget. If such capital expenditures are not planned and included in the Project Operations & Capital Expenditure Budget, Lodemo may make such capital expenditures directly and consider them as Allowable Costs provided that Lodemo will not proceed without Global's written consent, which shall not be unreasonably withheld, prior to approving any capital project in excess of \$5,000 for any one expenditure or collectively for all related expenditures.

2.10. Global's Presence at Site

During all of Lodemo's activities and for the purpose of reviewing Lodemo's progress and performance, Global will have total unrestricted access to all Project locations, including the nursery, farms, Oil Extraction Facility, and any other sites where Global's materials are stored or work under this Agreement is being conducted.

2.11. Permits and Licenses

Lodemo shall be responsible for obtaining and maintaining in Global's name any and all permits or licenses required by Applicable Law relating to the construction, operation, and ownership of the Project, Farm, Facilities or Logistics (but not relating to employees or other labor hired by Lodemo or New Co.). In the event that under Applicable Law Lodemo is required to be a party to such permits or licenses, Lodemo shall hold those permits and licenses on behalf of Global and for Global's account. Lodemo or New Co. shall obtain and maintain all permits and licenses required by Applicable Law to hire the employees and laborers used in the Project, including persons working on the Farms, in the nursery, and in the Oil Extraction Facility. All direct costs associated with obtaining and maintaining permits are considered Allowable Costs. Lodemo shall be responsible for obtaining and maintaining such permits and licenses.

Page 7 of 23

3. <u>BUDGETING</u>

3.1. The Budgeting Process

Lodemo shall, for each calendar year, provide Global with Lodemo's proposed annual (i) Construction & Operating Plan and (ii) Project Operations & Capital Expenditure Budget (hereinafter the "Budget"). The draft Construction & Operating Plan and Budget is attached hereto as Exhibits D and E, which Construction & Operating Plan and Budget will be updated and submitted for approval by Global; all subsequent annual Construction & Operating Plans and Budget's will be delivered to Global at least 90 days prior to the beginning of each subsequent calendar year. Global shall provide prompt review and comments to Lodemo as soon as reasonably practicable, but no later than forty five (45) days after receipt of the proposed Construction & Operating Plan and Budget. Global's comments may include questions, comments, objections or suggested modifications which Global may have with respect to such proposed plan and budget, and the parties shall cooperate with each other in developing a mutually acceptable Construction & Operating Plan and Budget. Global must approve the Construction & Operating Plan and Budget in writing before they may be considered valid and before costs associated with the Construction & Operating Plan and Budget may be considered as Allowable Costs. If Global does not approve the annual Construction & Operating Plan and Budget, mult such time as a new Construction & Operating Plan and Budget may be established through negotiation or according to dispute resolution methodology described in Section 15. Lodemo will provide Global with a monthly reconciliation of actual expenditures compared to the Budget, and Global and Lodemo will review annual Construction & Operating Plan and Budget may be revised with the written consent of both parties during monthly reviews to reflect revisions necessitated by changed circumstances, including changes in law, scope, emergencies, and force majeure events.

3.2. Budget Detail and Format

Information included in the annual Construction & Operating Plan and Budget, as well as all supporting data, formulas, calculations, and back-up information will be available for review at any time by either party and will be shared upon request by the parties. Annual Construction & Operating Plans and Budgets will be provided by Lodemo in significant detail consistent with common professional practice and in a form that Global may request in order for it to adequately provide review and make prudent business decisions.

4. <u>MANAGEMENT AND CONTROL</u>

4.1. <u>Responsibility of Personnel</u>

Except as otherwise provided in this Agreement, Lodemo shall have exclusive managerial control over and responsibility for its personnel (and any personnel hired by New Co.), any subcontractors, and suppliers in the execution of its Services. Such control shall include decision making authority over coordination and scheduling of labor, logistics, and operations; <u>provided</u>, however, that such control shall be effected in a manner which does not adversely affect the quality of service, result in delays, or violate or contradict any labor laws, rules, or moral obligations applicable to employees under the responsibility, supervision and control of either Lodemo, its subcontractors or suppliers, NEW Co., or any employment agency performing services hereunder, and such employees shall not be required to report to any person who is not ultimately reporting up through Lodemo.

Page 8 of 23

4.2. Global Right to Review

Global reserves the right to review the management practices of Lodemo, its subcontractors and suppliers, NEW Co., and any employment agency performing services hereunder to determine and ensure that the Services are managed using methods that are not detrimental to safety and health of any employed person, or are not in violation of applicable standards or laws.

4.3. Global Project Manager

Global shall have the right at any time to appoint a Global representative as the person responsible for overall strategic planning and management of the Project and who may be located at any given time at any of the Facilities (the "Global Project Manager"). The Global Project Manager may also be assisted in his work by other Global staff. The Global Project Manager's function shall be to provide overall direction regarding the Project, including (i) the purchasing/leasing land, (ii) approving changes to the Budget, (iii) establishing overall practices and procedures for planting and harvesting Jatropha plants and seeds, and (iv) monitoring Lodemo's progress in carrying out the Construction & Operating Plan. The Global Project Manager and his staff, if any, will not provide supervision of Lodemo construction, operations, or logistics personnel and will not otherwise be responsible for, or involved in, the day-to-day operations of the Project. Global shall bear all costs and expense associate with its Global Project Manager and staff will be provided access to all Facilities and financial and operating records of the Project, and at the Global Project Manager's request, will have the right to participate in Lodemo's planning, scheduling, and budgeting activities and meetings convened concerning operations or the execution of its Services ("Management Meetings").

4.4. Project Management Meetings

Lodemo will plan and arrange for monthly project management meetings to occur on or about the $2\theta^{h}$ day of every calendar month during the term of the Agreement (Management Meetings"). Through the Management Meetings, the parties shall perform progress and budget reviews and coordinate any shutdowns, curtailments, service outages, and changes to operations.

Page 9 of 23

5. <u>ENVIRONMENTAL POLICIES</u>

5.1. Environmental Plan

As a part of its Services, Lodemo shall provide Global with a project environmental plan, which will include a description of its waste collection and disposal strategy, land clearing and biomass disposal strategy, site drainage plans, and its plan for providing any other services in accordance with local and national environmental laws and Global's corporate polices and practices.

5.2. Environmental Permits

On behalf of Global, Lodemo will apply for and bear responsibility for acquiring any required environmental permits and/or licenses, and will maintain and administer said permits and/or licenses for the Project and Facilities.

6. <u>COMPENSATION FOR SERVICES</u>

As compensation for Lodemo providing the Services herein, Global shall make payments to Lodemo as follows:

6.1. <u>Allowable and Non-Allowable Costs</u>

Allowable and Non-Allowable Costs for the Project shall be determined as follows.

6.1.1. Allowable Costs

"Allowable Costs" shall mean the actual costs described in this Subsection that are paid or payable by Lodemo and necessarily incurred during performance of the Services. Allowable Costs are subject to additions or deductions which may be made in accordance with this Agreement. Allowable Costs shall only include the following items:

6.1.1.1. Direct costs of salaries and wages actually paid to Lodemo's full-time employees listed in Exhibit "B". Other personnel who are specifically named in Exhibit "B" may also be charged to the Project to the extent that use of such personnel is included in the Budget or otherwise approved by Global. Any changes to such chargeable personnel during the course of the Project must be approved in writing by Global.

6.1.1.2. A factor of fifteen percent (15%) (Lodemo's "Labor Burden Rate") of the actual salaries and wages described in Subsection 6.1.1 above to compensate Lodemo for the cost of all statutory payroll taxes levied or assessed by any governmental body during the performance of the Services, including but not limited to retirement, unemployment taxes and unemployment insurance, and worker's compensation costs; and for the cost of any and all company paid employee benefits, including but not limited to holiday pay, vacation pay, sick leave, retirement plans, group medical and life insurance benefits. The Labor Burden Rate shall be applied to all hours at the salaries or wages listed in Exhibit "B".

Page 10 of 23

6.1.1.3. Actual direct cost to Lodemo of non-employee agency personnel in accordance with the amounts paid pursuant to the applicable agency hire agreements plus an allowance of five percent (5%) to compensate Lodemo for its overhead costs pertaining to such personnel.

6.1.1.4. Actual cost of amounts paid or payable by Lodemo to its subcontractors and vendors for services performed pursuant to subcontracts and purchase orders which have been reviewed and approved by Global.

6.1.1.5. Actual direct cost to Lodemo of all materials and equipment incorporated into the Services by Lodemo including the direct costs of transportation thereof. Said costs shall be invoiced at actual trade and quantity discount prices, when applicable. Any salvage value actually realized by Lodemo at the end of the Project for any excess items paid for by Global shall be a credit for Global's account.

6.1.1.6. Actual direct costs to Lodemo of rental charges for all necessary construction machinery and equipment utilized in the Services, exclusive of small tools, including the direct costs of installation, dismantling, removal, maintenance, oil and grease; insurance; transportation and delivery. Such rental charges shall not exceed the prevailing rates in the area of the Project.

6.1.1.7. Actual and reasonable travel and subsistence expenses of Lodemo's Site Personnel listed in Exhibit "B" hereto while traveling in the discharge of duties in connection with the Services. Lodemo shall use its best efforts to obtain the lowest cost for such travel and expenses.

6.1.1.8. Actual direct costs of sales and use taxes directly relating to the Services that are imposed by governmental authorities and paid by Lodemo on behalf of Global.

6.1.1.9. Costs of clean-up and removal of debris.

6.1.1.10. Costs incurred due to an emergency affecting the safety of persons and property.

6.1.1.11. Costs of site security services for protection of the Services and Project unless provided by Global.

6.1.1.12. Project-related license fees required by statute.

6.1.1.13. Construction, building, or environmental permit fees paid by Lodemo when approved in advance and in writing by Global.

Page 11 of 23

6.1.1.14. Other actual direct costs incurred in the performance of the Services if and to the extent such costs are incurred in the course of Lodemo's execution of the Plan and are contemplated in the Budget as approved in accordance with this Agreement.

6.1.1.15. Costs of claims, remedial actions, fines and damages resulting from Global's refusal or failure to implement plans and courses of actions specified in the annual Plan and Budget, including any labor termination and subcontractor termination cost.

6.1.2. Non-Allowable Costs

"Non-Allowable Costs" shall mean the direct and/or indirect costs described in this Subsection 6.1.2. All such Non-Allowable Costs are included in Lodemo's Management Fee set forth in Section 6.4 and/or as provided by mutual written agreements, including Change Orders, as provided for in this Agreement. Lodemo shall not be entitled to receive any additional reimbursement for any of the items described as follows:

6.1.2.1. All direct and indirect operating, maintenance and overhead costs of any kind relating to Lodemo's principal and branch offices which is not dedicated to or reserved for use on the Project, including but not limited to office space, furniture and equipment; rent; maintenance; local telephone; utilities; depreciation; security; furniture; office equipment; office supplies; property taxes; the development of construction manuals, standards or computer programs; personnel training other than for safety training; and janitorial services. The parties agree that Global shall only pay or reimburse Lodemo for any expenses related to its offices that are dedicated to or reserved for the use of the Project is Global has approved such offices in writing.

6.1.2.2. Any expenses relating to Lodemo's operating capital, including interest on Lodemo's capital employed in support of the Services.

6.1.2.3. All costs arising out of grossly negligent acts or omissions by Lodemo, any subcontractor, vendor or anyone directly or indirectly employed by any of them, or for whose acts any of them may be liable, for: (a) all costs including defense costs, losses and damages arising out of Lodemo's indemnity obligations to the extent defined in Article 8 hereof; and (b) the cost of all deductibles and losses not covered by any of the insurance policies required to be provided pursuant to Article 8 hereof.

6.1.2.4. All costs incurred, at any time during the course of executing the Services by Lodemo, any subcontractor, vendor or anyone directly or indirectly employed by any of them, or for whose acts any of them may be liable, for correction, removal, replacement and disposal of any non-conforming work, materials or equipment to the extent defined in this Agreement.

Page 12 of 23

6.1.2.5. All costs incurred by Lodemo for bonuses, stocks options, profits sharing arrangements and similar incentive programs.

6.2. Reimbursement for Allowable Direct Costs

Lodemo will maintain complete and detailed records of all of its direct and actual out-of-pocket costs and expenses related to the Project or otherwise subject to this Agreement. On or about the 20th day of each month during the term of this Agreement, Lodemo will submit its monthly application for payment ("Monthly Invoice"). Lodemo's Monthly Invoice will include a detailed record of the previous month's direct and actual costs including employee man-hours and costs, agency personnel man-hours and costs, subcontractor costs, material costs, and equipment costs. All direct costs must be supported as Allowable Costs by verifiable documentation including timesheets and receipts. Undocumented or Non-Allowable Costs costs will not be reimbursed.

6.3. Operating Account

6.3.1. Establishment of Operating Account

Lodemo and Global shall establish a local depository account at a bank or other institution approved in writing by Global (the "Operating Account"). If requested by Global, all funds in the Operating Account shall be invested in a cash management program approved by Global ("Cash Management Account"). Lodemo will be authorized to make deposits and withdrawals from the Operating Account and Cash Management Account in accordance with this Agreement and within the guidelines of the then current Budget. Project funds shall not be commingled with Lodemo's other funds and only Project funds shall be deposited in the Operating Account and in no other account. Lodemo shall pay all of its direct costs authorized to be paid under this Agreement out of the Operating Account or Cash Management Account.

6.3.2. Funding of Operating Account

On or before the 1st day of each month, Global agrees to deposit into the Operating Account or Cash Management Account funds in an amount sufficient to maintain the Minimum Balance. The term "Minimum Balance" when used herein shall mean an amount equal to the projected operating and capital expenses of the Project projected for the following two months, as forecasted in the Budget, less any existing balances in the Operating Account. If, upon review of Lodemo's Monthly Invoice it is determined that Lodemo has expended funds from the Operating Account or Cash Management Account determined by Global to be Non-Allowable Costs, either Lodemo will replace these funds directly or credit Global by the same amount in the following month's Monthly Invoice. If the Non-Allowable costs are not replaced or credited, the Minimum Balance will be reduced by the same amount invoiced as a Non-Allowable Cost.

Page 13 of 23

6.4. Management Fees

6.4.1. Basic Fee

On a monthly basis, Global shall pay to Lodemo for the duration of this Agreement, a Management Fee equal to the greater of (i) \$60 per hectare per year (\$5 per hectare per month) for every full hectare planted and under cultivation in Jatropha, and (ii) the monthly fee shown by schedule in Exhibit A (hereinafter the "Minimum Fee"). Notwithstanding the foregoing, if at the time the Management Fee is determined the number of hectares planted and under cultivation is less than the amount set forth on Exhibit A solely as a result of Lodemo's failure to comply with its obligations under this Agreement, including its obligations to provide land aggregation, nursery and planting Services in a timely manner, the Minimum Fee shall be readjusted to reflect the number of hectares not planted as a result of Lodemo's failure to comply.

6.4.2. Performance Payment

Global will pay to Lodemo an incentive to reduce costs of operation of the Project, Farm and Facilities. Lodemo will be paid 20% of the reduction in actual direct costs (Allowable Costs) below the approved annual Budget for each year it reduces said costs. This cost reduction applies to all costs including land leases and other direct costs. This reduction may not in any way adversely impact quality of Product, worker treatment, safety or long term viability of the Farm or Project or violate any law or environmental rule or policy.

6.4.3. Provision for Transfer of Oil to Mexico Joint Venture

As additional compensation for Lodemo's Services under this Agreement and its agreement to enter into the Joint Venture, Global agrees to sell 10% of the net Jatropha seed oil produced by the Project to the Joint Venture for exclusive use as feedstock for the Biodiesel Refinery. The price at which such Jatropha seed oil will be sold to the Joint Venture shall equal Global's actual cost of production, including the allocated portion of the payments made by Global to Lodemo under this Agreement (excluding this Subsection 6.4.3).

6.4.4. Improved Production Bonus

Global will provide to Lodemo an incentive for increased oil production from the Project. Twenty percent (20%) of all oil produced by Global as part of the Project in excess of the projected baseline production shown in Exhibit C (as amended with the mutual consent of the parties from time to time) will be sold to the Joint Venture at the same average cost as other oil sold by Global during any given month for exclusive use as feedstock to the Joint Venture.

Page 14 of 23

6.4.5. Biomass Sales Incentive

Global will pay to Lodemo an incentive for all sales of the biomass byproduct produced from the seed oil extracted at the Oil Extraction Facility (hereinafter the "Biomass Incentive"). The Biomass Incentive will be 50% of the net price of the sale, which will be calculated by subtracting from the actual price received the amount of the actual handling costs of the biomass, including shipping, from the gross sales price. The Biomass Incentive will be paid within 30 days of receipt by Global of funds from the buyer.

All potential sales of biomass are subject to review by Global, and Global reserves the right to either accept or reject any potential sale. However, if Global chooses to sell the biomass to a buyer for a lower price than an alternate buyer, Global will pay to Lodemo its Biomass Incentive based on the documented price of the alternate buyer.

6.4.6. Payment of Redundancy Costs in the Event of Suspension or Termination

At any time during the term of this Agreement, upon suspension or termination of this Agreement or the supply or performance of Services hereunder, in whole or in part, or upon cessation of production at the Project, or upon any major reduction by Global in the quantity of Products to be produced at the Project hereunder, in each case for any reason other than a breach by Lodemo of this Agreement, Global shall reimburse Lodemo for all direct costs incurred in separation and outplacement associated with any of Lodemo's or it's subcontractor's personnel or employees to the extent made redundant as a result of any such suspension, termination, cessation, or permanent reduction or shutdown, or major reduction. Any direct costs incurred and approved according to terms described in this Section will be debited from any remaining balance of funds previously advanced by Global to Lodemo through the Operating Account or Cash Management Account.

7. INSURANCE

Each party shall purchase and maintain its own insurance in such amounts and covering such risks as are usually carried by companies engaged in the same or similar business and similarly situated, including insurance against public liability and property damage. Both parties will work together to minimize the existence of non-required or redundant insurance coverages. Notwithstanding, the parties agree to purchase appropriate insurance coverage for the risks related to this Project. These risks and insurance coverage may include heavy rainfall, hurricanes, floods, fire and explosion damage. Cost of insurance is generally an Allowable Cost, but must be approved in advance in writing by Global.

8. THIRD PARTY INDEMNIFICATION

Page 15 of 23

To the extent not covered by insurance, Lodemo agrees, to the fullest extent permitted by law, to indemnify, defend, and hold Global and its officers, agents, and employees of Global harmless from and against any and all third party claims, demands, causes of action, damages, losses, and expenses of whatsoever nature, character, or description, regardless of merit thereof, which are or may be asserted against Global by any person or entity, and which arise out of or result from, in whole or in part, (i) the negligent acts or omissions of Lodemo in the performance of the Services under this Agreement, or (ii) the breach by Lodemo of any terms of this Agreement. The acceptance of the Services by Global shall not operate as a waiver of such right of indemnification.

To the extent not covered by insurance, Global agrees, to the fullest extent permitted by law, to indemnify, defend, and hold Lodemo, its officers, directors, employees, and subcontractors harmless from and against all third party claims, demands, causes of action, damages, losses, and expenses of whatsoever nature, character, or description, regardless of the merit thereof, which are or may be asserted against Lodemo by any person or entity, and which arise out of or result from, in whole or in part, (i) the negligent acts or omissions of Global in the performance of Lodemo's obligations under this Agreement, or (ii) the breach by Global of any terms of this Agreement.

9. LIMITATION OF LIABILITY AND DISCLAIMER

Each party to this Agreement shall only be liable to the other party and its agents and employees for losses sustained by the other party and its agents and employees directly as a result of the first party's lack of performance under this Agreement and losses resulting from intentional misconduct or gross negligence by the first party, its employees, it's subcontractors (including the employees of NEW Co.). Neither party shall be liable for special, incidental, indirect, punitive or consequential damages under this Agreement; provided that this limitation shall not apply in the case of a party's willful acts. Anything contained herein to the contrary notwithstanding, Global acknowledges that Lodemo makes no representation or warranty of any kind with respect to having past experience in the provision of the Services, the financial viability of the Project, or the fitness of Jatropha oil for its intended use as a feedstock.

10. <u>TAXES</u>

Global shall reimburse Lodemo on a current basis for all taxes, excises or other charges which Lodemo may be required to pay to any government (federal, state, or local) relating to the Project, whether current in nature or as the result of any tax audit, by reason of any of its activities hereunder and not otherwise provided for herein. Notwithstanding the foregoing, Global shall not be required to reimburse Lodemo for taxes on or measured by Lodemo's fees, net income or profit,

11. OWNERSHIP

Under the terms of this Agreement, Lodemo shall not receive or acquire legal or equitable ownership or possession rights in any part of the Project or to any of Global's assets. Any ownership or equity interest Lodemo may ultimately receive will be covered under a separate agreement.

Page 16 of 23

Except for the rights of access and operation granted in this Agreement, Lodemo shall not receive or acquire any legal or equitable ownership or possession rights to any part of the Project or any of Global's personal or real property contained or stored at the Project or the Facility. This exclusion to access and ownership includes seed oils and/or chemicals stored in tanks, trucks, railcars, tanks or barges.

The following documents shall be collectively referred to in this Agreement as the "Project Documents": all information, intellectual property, work product, and all documents (each in whatever medium or format, including computerized reports and information on disk) related to arising out of the Project, and the performance of the Services, including, without limitation, all land plans, maps, engineering studies, soil studies, geological studies, and other engineering information, all documentation prepared or obtained by Lodemo or Global in preparation for filings with or filed with a governmental or quasi-governmental agency plans and specifications and environmental reports, and all other studies, tests, work product, analyses, development plans, studies, drawings, designs, and sketches, memoranda, construction documents, marketing plans, financial analyses, books, records, data, and reports prepared or obtained by Lodemo or Global, their its employees, agents, consultants or subcontractors relating in any manner to the Project. The Project Documents shall be and remain the sole and exclusive property of Global, and Global shall have the right to use such for any purpose without any additional compensation to Lodemo. Lodemo shall acquire no ownership rights in the Project Documents, but shall have the right to use such Project Documents in performance of the Services hereunder. Lodemo acknowledges that it is acting as an agent of Global, and all of the work product and Project Documents, and other intellectual property created, produced, or procured by Lodemo or any consultant or subcontractor, regardless of form of medium, are the sole property of Global, and the patent, if issued, shall be in Global's name and be the sole property of Global. All trademarks, trade names, logos, and other copyrightable materials shall be owned by Globable and may be registered by Global.

12. CONFIDENTIALITY

Neither party shall, directly or indirectly, disclose, communicate, divulge, furnish or make accessible or available, in whole or in part, to any person, firm, company, corporation or other entity, other than to its employees and other representatives to the extent necessary to discharge and perform its obligations under this Agreement or as required by any law or regulation, any data, know—how, drawings, plans, written instructions, or other writings, processes, techniques, methods, designs, inventions, materials, formula, equipment, machinery, devices and the like (whether or not patentable) of a secret and confidential nature, and any other confidential information, material or matter, or trade secrets, relating or pertaining to the business of the other party that it learns of pursuant to this Agreement including, any such information relating to the processes and equipment for the production of products by the process used in the Project or by a similar process.

Page 17 of 23

13. <u>TERM</u>

Unless terminated pursuant to Section 14, the term of this Agreement shall commence on the date hereof and continue for 20 years from the date hereof. At the expiration of the term, or upon termination pursuant to Section 14, Global shall continue to own all rights to the contracts, land, agricultural output, Facilities, Products, intellectual rights, and all other Project assets. Sections 8, 9, 11 and 12 shall survive the termination of this Agreement.

14. TERMINATION FOR DEFAULT; FORCE MAJEURE

14.1 In the event Lodemo is in default in the performance of any obligation under this Agreement and shall fail to diligently proceed to correct such default within thirty (30) calendar days following written notice from Global (or such longer time period to the extent the cure of the default is of a nature so as to require more time, provided that Lodemo commences the cure and continues the prosecution of the cure until completed), or if Lodemo dissolves, files a petition in bankruptcy, or makes a general assignment for the benefit of its creditors, or if a petition in bankruptcy is filed against Lodemo or a receiver is appointed for reasons of insolvency; Global may, without prejudice to any other rights or remedies Global may have, terminate this Agreement by written notice. In the event of termination for default, Global may take possession of and finish the Services by whatever method Global deems expedient. In such event, Lodemo shall be liable for all damages sustained by Global by reason of such termination, whether at law or in equity. Global's right to require strict performance of any and all obligations in this Agreement shall not be affected by any previous waiver, forbearance or course of dealing prior to such termination. Global shall be entitled to recover its reasonable attorneys' fees, costs and disbursements in any action successfully brought to enforce its rights under this Agreement.

14.2 Global shall have the option, exercisable in good faith at its reasonable judgment, to terminate this Agreement, upon 90 days' prior written notice to Lodemo, if both (i) an event of Infeasibility has occurred, and (ii) Global thereafter ceases to independently pursue the Project. For purposes of this Agreement, "Infeasibility" shall mean the inability of the parties to proceed with the Project or impracticality of the Project for reasons beyond their reasonable control, including, without limitation:

1. Political actions of a governmental agency that makes the Project more costly or time consuming or financially burdensome to undertake, including, without limitation, restrictions on land ownership or use, restrictions on the processes involved in the Project, nationalization of the Project, or enactment of laws that significantly and adversely affect the Project.

2. Determination that the oil extraction process does not result in commercially significant quantities of usable oil, or that the price for such oil makes the Project commercially unsustainable.

3. Natural disaster that affects a material portion of the Project (including the Jatropha plants, the distribution system, or the Oil Extraction Facility), and is not able to be corrected within twelve months.

Page 18 of 23

4. Unexpectedly high costs of operations, provided that high costs shall not be grounds for claiming Infeasibility unless the overall operating costs exceed 200% of the Budget.

15. CLAIMS AND DISPUTE RESOLUTION

All claims and disputes and other matters in question arising out of or relating to this Agreement or the breach thereof, shall be submitted first to voluntary Mediation, and if Mediation is not successful, then to binding Arbitration, in accordance with the then-current Model Procedure for Mediation of Business Disputes of the Center for Public Resources. Arbitrators and Mediators shall be jointly selected by the parties, and have experience in international dispute resolution involving business between Mexico and the United States. Judgment on any arbitration award may be entered in a court of any competent jurisdiction.

The prevailing party shall be entitled to recover its reasonable attorneys' fees, costs and disbursements in any action brought to enforce or its rights under this Agreement or to interpret the provisions of this Agreement. The existence of any claim, dispute or legal proceeding shall not relieve Lodemo from its obligation to properly perform its Services during such Mediation or Arbitration proceedings as set forth herein.

Notwithstanding the above, before either party pursues Mediation or Arbitration, the parties agree to attempt to resolve any dispute amicably at a meeting to be attended by persons with decision-making authority. If, within thirty (30) calendar days after such meeting, the parties have not succeeded in negotiating a resolution of the dispute, they agree to resolve the dispute according to the Mediation and Arbitration dispute resolution process described in this Section.

16. MISCELLANEOUS

16.1. Successors and Assigns

Except as expressly provided below, neither this Agreement nor any right, interest or obligation hereunder may be assigned by either party without the prior written consent of the other, except that (i) either party may assign this Agreement to an Affiliate of such party; <u>provided</u> that the party assigning this Agreement shall remain liable notwithstanding such assignment, (ii) Lodemo may assign this Agreement in whole to any purchaser of its business and assets, but as a condition of any such assignment, shall procure the written assumption of this Agreement, and (iii) Global may assign this Agreement to any purchaser of all or substantially all of the assets of the Project, <u>provided</u> that Global shall, as a condition of any such assignment, procure the written assumption of this Agreement by such assignee. The terms, covenants and conditions contained in this Agreement, including the obligations of indemnity contained herein, are binding upon and inure to the benefit of Lodemo and Global and their respective successors and permitted assigns and shall survive any transfer of the ownership or control of Lodemo or Global.

(The remainder of this page is intentionally left blank)

Page 19 of 23

17. ENTIRETY OF AGREEMENT

This Agreement contains the full and complete understanding of the parties pertaining to the Project and the Services and supersedes any and all prior representations, negotiations, agreements or understandings between the parties, whether written or oral. This Agreement may not be modified except by a subsequent writing executed by both parties.

The parties hereby execute this Agreement by their respective duly authorized representatives as of the Effective Date stated in the preamble of this Agreement.

Medical Discoveries, Inc. dba Global Clean Energy Holdings (Global) Lodemo y Asociados S.C.P. (Lodemo)

Page 20 of 23

EXHIBIT A

MINIMUM FEE SCHEDULE

Page 21 of 23

EXHIBIT B

LODEMO PERSONNEL

Page 22 of 23

EXHIBIT C

BASELINE PRODUCTION

Page 23 of 23

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors

Global Clean Energy Holdings, Inc.

We consent to the incorporation by reference in Registration Statement on Form S-8 (Nos. 333-92446) of Global Clean Energy Holdings, Inc. of our report dated April 14, 2009, appearing in this Amendment No. 1 to Annual Report on Form 10-K/A of Global Clean Energy Holdings, Inc. for the year ended December 31, 2008.

/s/ HANSEN, BARNETT & MAXWELL. P.C. Hansen, Barnett & Maxwell. P.C. REGISTERED PUBLIC ACCOUNTANTS December 1, 2009

Certification of the Principal Executive Officer Under Section 302 of the Sarbanes-Oxley Act

I, Richard Palmer, certify that:

- 1. I have reviewed this report on Form 10-K/A of Global Clean Energy Holdings, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d- 15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: December 1, 2009

By: /s/ RICHARD PALMER

Name: Richard Palmer Title: President and Chief Executive Officer

Certification of the Principal Financial Officer Under Section 302 of the Sarbanes-Oxley Act

I, Bruce Nelson, certify that:

- 1. I have reviewed this report on Form 10-K/A of Global Clean Energy Holdings, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: December 1, 2009

By: /s/ BRUCE NELSON

Name: Bruce Nelson Title: Chief Financial Officer

CERTIFICATION OF THE PRINCIPAL EXECUTIVE OFFICER

Pursuant to 18 U.S.C. § 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Global Clean Energy Holdings, Inc. (the "Company") hereby certifies that, to his knowledge:

(i) The Annual Report on Form 10-K/A of the Company for the fiscal year ended December 31, 2008 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and

(ii) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: December 1, 2009

By: /s/ RICHARD PALMER

Name: Richard Palmer Title: President and Chief Executive Officer

CERTIFICATION OF THE PRINCIPAL FINANCIAL OFFICER

Pursuant to 18 U.S.C. § 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Global Clean Energy Holdings, Inc. (the "Company") hereby certifies that, to his knowledge:

(i) The Annual Report on Form 10-K/A of the Company for the fiscal year ended December 31, 2008 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and

(ii) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: December 1, 2009

By: /s/ BRUCE NELSON

Name: Bruce Nelson Title: Chief Financial Officer